1

# **Council Meeting Agenda - 14 April 2021**

Meeting will be held in the Council Chamber, Level 2, Philip Laing House, 144 Rattray Street, Dunedin

Members:

Cr Andrew Noone, Chairperson Cr Michael Laws, Deputy Chairperson Cr Hilary Calvert Cr Michael Deaker Cr Alexa Forbes Hon Cr Marian Hobbs

Cr Gary Kelliher Cr Kevin Malcolm Cr Gretchen Robertson Cr Bryan Scott Cr Kate Wilson

Cr Carmen Hope

Senior Officer: Sarah Gardner, Chief Executive

Meeting Support: Dianne Railton, Governance Support Officer

14 April 2021 01:00 PM

## Agenda Topic

#### 1. **APOLOGIES**

6.

7.

No apologies were noted at the time of publication of the agenda.

#### PUBLIC FORUM 2.

Requests to speak should be made to the Governance Support team on 0800 474 082 or to governance@orc.govt.nz at least 24 hours prior to the meeting; however, this requirement may be waived by the Chairperson at the time of the meeting.

2.1 Richard Robson requested to speak at Public Forum about water quality in local streams.

#### CONFIRMATION OF AGENDA 3.

Note: Any additions must be approved by resolution with an explanation as to why they cannot be delayed until a future meeting.

#### 4. CONFLICT OF INTEREST

Members are reminded of the need to stand aside from decision-making when a conflict arises between their role as an elected representative and any private or other external interest they might have.

#### CONFIRMATION OF MINUTES 5.

The Council will consider minutes of previous Council Meetings as a true and accurate record, with or without changes.

|        | 5.1        | Minutes of the 24 March 2021 Council Meeting              | 3  |
|--------|------------|---|----|
| 6.     | ACTI       | ONS (Status of Council Resolutions)                       | 14 |
| The Co | ouncil wil | review outstanding resolutions.                           |    |
| 7.     | MAT        | TERS FOR COUNCIL CONSIDERATION                            | 15 |
|        | 7.1        | DRAFT TERMS OF REFERENCE FOR REGIONAL COORDINATION GROUPS | 15 |



Page

|                    |   | 7.1.1   | Attachment 1: Terms of Reference Regional Coordination Group for Wilding Conifer Management   | 17  |
|--------------------|---|---|---|-----|
|                    |   | 7.1.2   | Attachment 2: Terms of Reference Regional Coordination Group for Wallaby Management   | 21  |
|                    |   | 7.1.3   | Attachment 3: MPI ORC Wilding Conifer Control Funding Agreement signed 2020.11.06   | 25  |
|                    |   | 7.1.4   | Attachment 4: MPI ORC Wallaby Management Programme Funding<br>Agreement signed 2020.11.06   | 60  |
|                    | 7.2   | SoE W   | ATER QUALITY REPORT   | 103 |
|                    |   | 7.2.1   | Attachment 1:State and Trends of Lake and River Water Quality in the Otago Region 2000-2020   | 112 |
| 8.                 | RESO  | LUTION  | N TO EXCLUDE THE PUBLIC   | 267 |
| That the<br>•<br>• | public be<br>Confirma<br>LiDAR D<br>Port Ota<br>Head Of | e exclude<br>ation of n<br>Data Capi<br>Igo Direc<br>ifice Acco | ed under LGOIMA Sec 48(1)(a) from discussions on the following items:<br>ninutes of the public-excluded Council Meeting of 28 February 2021<br>ture Contract Approval<br>tor Appointment<br>ommodation Update |     |
|                    | 8.1   | Public  | Exclusion Table   | 267 |
| 9.                 | CLOS  | JRE   |   |     |



Minutes of an ordinary meeting of Council held in the Waitaki District Council Chamber, Oamaru on Wednesday 24 March 2021 at 1:00 PM

#### Membership

Cr Andrew Noone Cr Michael Laws Cr Hilary Calvert Cr Alexa Forbes Cr Michael Deaker Hon Cr Marian Hobbs Cr Carmen Hope Cr Gary Kelliher Cr Kevin Malcolm Cr Gretchen Robertson Cr Bryan Scott Cr Kate Wilson (Chairperson) (Deputy Chairperson)

#### Welcome

Chairperson Noone welcomed Councillors, members of the public and staff to the meeting at 1.00 p.m.

Staff present included: Sarah Gardner (Chief Executive), Gwyneth Elsum (GM Strategy, Policy and Science), Gavin Palmer (GM Operations), Amanda Vercoe (Executive Advisor), Dianne Railton and Liz Spector (Governance Support), Eleanor Ross, Ryan Tippett, Mike Roesler, and via Zoom: Nick Donnelly (GM Corporate Services), Sean Geary, Anita Dawe, Joanna Gilroy, Anne Duncan and Warren Hanley.

#### For our future

70 Stafford St, Private Bag 1954, Dunedin 9054 | ph (03) 474 0827 or 0800 474 082 | www.orc.govt.nz

## 1. APOLOGIES

Resolution

That the apology for Cr Robertson be accepted.

Moved: Cr Noone Seconded: Cr Hope CARRIED

### 2. PUBLIC FORUM

No public forum was held.

### 3. CONFIRMATION OF AGENDA

A late paper was received on Delegations COVID-19 Recovery (Fast-Tracking Consenting) Act 2020.

Chair Noone moved:

#### Resolution

That the report Delegations COVID-19 Recovery (Fast-Tracking Consenting) Act 2020 be added to the agenda.

Moved: Cr Noone Seconded: Cr Wilson CARRIED

#### 4. CONFLICT OF INTEREST

No conflicts of interest were advised.

#### 5. CONFIRMATION OF MINUTES Resolution

That the minutes of the (public portion of the) Council meeting held on 24 February 2021 be received and confirmed as a true and accurate record.

Moved: Cr Hobbs Seconded: Cr Wilson CARRIED

### 6. ACTIONS (STATUS OF COUNCIL RESOLUTIONS)

The outstanding Actions were reviewed.

#### 7. MATTERS FOR CONSIDERATION

#### 7.1. Long-term Plan 2021-31 Consultation Document and Supporting Information

The report was provided to seek Council's approval of the consultation document and supplementary information supporting the draft Long-term Plan 2021-31. Mike Roesler (Manager Corporate Planning and Nick Donnelly (GM Corporate Services) were present to speak to the report and respond to questions.

Minutes Council Meeting 2021.03.24

Cr Malcolm thanked the Chief Executive and GM Corporate Services and staff for the work undertaken on the Long-term Plan. After further discussion, Cr Malcolm moved:

#### Resolution

That the Council:

1) Approves the signing of the Audit representation letter by the Chair.

| Moved:     | Cr Malcolm |
|------------|------------|
| Seconded:  | Cr Hope    |
| CARRIED    |            |
| Resolution |            |

That the Council:

1) **Notes** the independent Auditor's Report on the "Have Your Say on Our Future – 2021-2031 Long-term Plan Consultation Document" will be provided following their receipt of the Audit Representation Letter and Council's approval of the Consultation Document.

| Moved:     | Cr Malcolm |
|------------|------------|
| Seconded:  | Cr Deaker  |
| CARRIED    |            |
| Resolution |            |

That the Council:

1) **Resolves,** on the basis of the independent auditor's report, that the 'Have Your Say On Our Future – 2021-2031 Long-term Plan Consultation Document' will achieve the purpose of providing an effective basis for public participation in deciding a Long-term Plan and in accordance with sections 93B and 93C of the Local Government Act 2002.

Moved: Cr Malcolm Seconded: Cr Deaker CARRIED

#### Resolution

That the Council:

1) Adopts the Draft Long-term Plan 2021-31 supplementary information that is relied on for the content of the 'Have Your Say On Our Future – 2021-2031 Long-term Plan Consultation Document' in accordance with section 93G of the Local Government Act 2002.

Moved: Cr Malcolm Seconded: Cr Deaker CARRIED

Minutes Council Meeting 2021.03.24

#### Resolution

That the Council:

1) **Notes** that revised Council 'Fees and charges' are attached to this report and included as supplementary information to the 'Have Your Say On Our Future – 2021-2031 Long-term Plan Consultation Document'

Moved: Cr Malcolm Seconded: Cr Deaker CARRIED

#### Resolution

That the Council:

1) Adopts for public consultation the Draft Revenue and Financing Policy and associated information summarising changes to this policy as provided in the Have your say on Our Future – 2021-2031 Long-term Plan Consultation Document'

Moved: Cr Malcolm Seconded: Cr Deaker CARRIED

#### Resolution

That the Council:

1) **Approves** the inclusion of any final changes to the 'Have Your Say On Our Future – 2021-2031 Long-term Plan Consultation Document' as directed by the external audit process and as tabled separately to this report at the 24 March 2021 Council meeting.

Moved: Cr Malcolm Seconded: Cr Deaker CARRIED

#### Resolution

That the Council:

1) **Approves** the inclusion of any final changes to the 'Have Your Say on Our Future – 2021-2031 Long-term Plan Consultation Document' as directed by the external audit process and as tabled separately to this report at the 24 March 2021 Council meeting, noting that Option 3 for Lake Hayes is to be included in the table.

Moved: Cr Malcolm Seconded: Cr Deaker CARRIED

Minutes Council Meeting 2021.03.24

#### 7.2. Port Otago Constitution

The report was provided for Council to approve the Constitution of Port Otago Limited and revised Guidelines for the Appointment of Directors to Port Otago Limited. Nick Donnelly was present to speak to the report and respond to questions.

Following discussion on the Port Otago Guidelines and the Constitution, Cr Calvert moved:

#### Resolution

That the Council:

- 1) **Receives** this report.
- 2) Approves the proposed Constitution of Port Otago Limited.
- 3) **Approves** the proposed Guidelines for Appointment of Directors to Port Otago Limited.

| Moved:    | Cr Calvert |
|-----------|------------|
| Seconded: | Cr Malcolm |
| CARRIED   |            |

#### 7.3. ORC Submission on Climate Change Commission's Report

The report was provided to request that Council approve a submission from Otago Regional Council (ORC) on He Pou a Rangi/the Climate Change Commission's (the Commission) '2021 Draft Advice for Consultation' document (the draft advice). Gwyneth Elsum (GM Strategy, Policy & Science), Anne Duncan (Manager Strategy) and Warren Hanley (Senior Resource Planner Liaison) were present to speak to the report and respond to questions.

#### Cr Hobbs sat back due to a possible conflict of interest.

There was discussion on the draft submission that was prepared following a Council workshop 9 March 2021. Cr Forbes expressed her disappointment with the submission content. Following further discussion, the submission will be revised to follow the direct format and a copy of the final submission will be circulated to Councillors.

#### Resolution

That the Council:

1) **Directs** staff to follow the direct format with the knowledge that we have now within our strategic directions and existing plans.

| Moved:    | Cr Calvert |
|-----------|------------|
| Seconded: | Cr Forbes  |
| CARRIED   |            |

#### Resolution

That the Council:

1) **Approves** that a finalised version of the submission on the draft advice is, subject to any changes required by Council, signed under delegation by the ORC Chairman and lodged.

Minutes Council Meeting 2021.03.24

Moved: Cr Noone Seconded: Cr Deaker CARRIED

Cr Malcolm, Cr Kelliher, Cr Laws requested their votes be recorded as against the motion.

Cr Hobbs rejoined the meeting.

#### 7.4. RMA s27 Six Monthly Report to the Minister

The report was provided to present, for adoption by Council, the third progress report to the Minister for the Environment, in accordance with section 27 of the Resource Management Act 1991 in relation to the recommendations made under section 24A of the Resource Management Act 1991.

Following discussion, Cr Hope moved:

#### Resolution

That the Council:

- 1) **Receives** this report.
- 2) **Approves** the third report to the Minister for the Environment, that reports on progress against the recommendations contained in his letter of 18 November 2019; and
- 3) **Notes** that the next report will be required to be provided by 30 September 2021.

Moved: Cr Hope Seconded: Cr Hobbs CARRIED

#### 7.5. Proposed Deed for Building Consent Authority Transfer

The report was provided for Council to approve the Agreement transferring the Building Consent Authority functions in relation to 'large dams' to Environment Canterbury. Joanna Gilroy (Manager Consents) and Charles Horrell (Consents) were present via Zoom to speak and respond to questions.

Following discussion, Cr Calvert moved:

#### Resolution

That the Council:

- 1) **Receives** this report.
- 2) **Approves** the affixing of the common seal to the transfer agreement subject to Environment Canterbury executing the Transfer Agreement.
- 3) **Approves** the variations to the existing transfer agreements with Southland and West Coast Regional Councils to ensure consistency with the transfer of functions to Environment Canterbury.

Minutes Council Meeting 2021.03.24

4) **Approves** the affixing of the common seal to the variation to the transfer agreements subject to Environment Southland and West Coast Regional Council executing the Transfer Agreements.

Moved: Cr Calvert Seconded: Cr Hope CARRIED

Cr Kelliher requested his vote be recorded as against the motion.

#### 7.6. Local Government New Zealand Annual Conference 2021 Attendance

The purpose of the report was to determine ORC representation at the Local Government New Zealand 2021 conference. Chair Noone said that Crs Malcolm and Wilson were to attend last year but the conference was cancelled due to COVID-19. Cr Laws questioned the cost and benefit of the LGNZ membership and attending the conference. Ms Gardner responded that both membership and attendance at the conference it is a valuable connection for ORC.

Cr Scott moved and Cr Noone seconded:

That the Council:

- 1) **Receives** this report.
- 2) **Nominates** Cr Malcolm, Cr Noone and Cr Wilson as ORC Councillor representatives to attend the 2021 LGNZ conference.

Following discussion, Cr Calvert moved an amendment seconded by Cr Hope:

That the Council:

1) **Requests** that the attendees report back on the conference to the next Governance, Comms and Engagement Committee meeting.

Moved: Cr Calvert Seconded: Cr Hope CARRIED

Then amendment carried and the substantive motion became:

That the Council:

- 1) **Receives** this report.
- 2) **Nominates** Cr Malcolm, Cr Noone and Cr Wilson as ORC Councillor representatives to attend the 2021 LGNZ conference.
- 3) **Requests** that the attendees report back on the conference to the next Governance, Comms and Engagement Committee meeting.

Moved: Cr Scott Seconded: Cr Noone CARRIED

Cr Laws requested his vote to be recorded as against the motion.

Minutes Council Meeting 2021.03.24

#### 7.7. Delegations COVID-19 Recovery (Fast Track Consenting) Act 2020

This report was provided to amend Council's Delegation Manual to provide delegation to make nominations on membership of Expert Consenting Panels for projects being considered under the COVID-19 Recovery (Fast-Track Consenting) Act 2020. Joanna Gilroy (Consents Manager) was present via Zoom to speak and respond to questions.

Cr Scott returned to the meeting at 02:59 pm.

After questions and discussion of the report, Cr Wilson moved:

#### Resolution

That the Council:

- 1) **Receives** this report.
- 2) **Approves** the delegation to make recommendations for panel members under the COVID-19 Recovery (Fast-Track Consenting) Act 2020 to the Chair of the Council.
- 3) **Approves** the Chief Executive to amend the Council's Delegations Manual to include the recommended addition.

Moved: Cr Wilson Seconded: Cr Malcolm CARRIED

#### 8. RECOMMENDATIONS ADOPTED AT COMMITTEE MEETINGS

# 8.1. Recommendations of the Finance Committee Resolution

That the Council adopts the resolutions of the 24 February 2021 Finance Committee.

Moved: Cr Hope Seconded: Cr Wilson CARRIED Resolution

# 8.2. Recommendations of the Data and Information Committee Resolution

That the Council adopts the resolutions of the 10 March 2021 Data and Information Committee.

Moved: Cr Wilson Seconded: Cr Calvert CARRIED

A further recommendation was made by Cr Wilson:

Minutes Council Meeting 2021.03.24

#### Resolution

That Council refer the report on Groundwater State of Environment to Strategy and Planning Committee for advice on where there are issues highlighted in the Discussions and Recommendations section of the report what action if any staff doing to rectify the situation.

Moved: Cr Wilson Seconded: Cr Calvert CARRIED

The meeting was adjourned at 3.10 p.m. for a short break. The meeting reconvened at 3.30 p.m.

**8.3.** Recommendations of the Implementation Committee *Cr* Forbes left the meeting at 03:29 pm.

#### Resolution

That the Council adopts the resolutions of the 10 March 2021 Implementation Committee.

Cr Forbes returned to the meeting at 03:30 pm.

Moved: Cr Hope Seconded: Cr Scott CARRIED

#### Resolution

That the Council adopts the resolutions of the 10 March 2021 Implementation Committee.

### 8.4. Recommendations of the Regulatory Committee

Resolution

That the Council adopts the resolutions of the 11 March 2021 Regulatory Committee.

Moved: Cr Kelliher Seconded: Cr Noone CARRIED

**8.5. Recommendations of the Otago and Southland Transport Committee** Dr Palmer provided a brief transport update to Council.

#### Resolution

That the Council adopts the resolutions of the 22 February 2021 Otago and Southland Transport Committee.

Moved: Cr Wilson Seconded: Cr Forbes CARRIED

Minutes Council Meeting 2021.03.24

#### 9. CHAIRPERSON'S AND CHIEF EXECUTIVE'S REPORTS

#### 9.1. Chairperson's Report

There was discussion on Three Waters and the meetings for the future of South Dunedin. Ms Gardner provided a brief update on Three Waters. Chair Noone will have further discussion with Ms Gardner and Dr Palmer on ORC's involvement with the Future of South Dunedin meetings.

#### Resolution

That the Chairperson's report be received.

Moved: Cr Wilson Seconded: Cr Hope CARRIED

# 9.2. Chief Executive's Report Resolution

That the Chief Executive's report be received.

Moved: Cr Calvert Seconded: Cr Hope CARRIED

Cr Laws returned to the meeting at 04:16 pm.

#### **10. NOTICES OF MOTION**

10.1. Notice of Motion - Investigate resource needs and timeline for possible collaborative development of an Otago Harbour Plan with ORC, DCC, Port Otago and Community Input

Cr Hope left the meeting at 04:18 pm.

In accordance with Standing Order 27.1, Cr Bryan Scott submitted the following Notice of Motion for inclusion on the agenda of the Council Meeting held on 24 March 2021.

Following discussion, Cr Scott moved:

#### Recommendation

That the Council:

1) **Request** staff to report back before completion of the LTP process on timeline and resourcing needs for collaborative development of an Otago Harbour Plan with ORC, DCC, Port Otago, iwi and community input.

Moved: Cr Scott Seconded: Cr Wilson CARRIED

Cr Hope returned to the meeting at 04:20 pm.

Minutes Council Meeting 2021.03.24

### **11. CLOSURE**

There was no further business and Chairperson Noone declared the meeting closed at 04:44 pm.

| Chairperson | Date |  |
|-------------|------|--|
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |
|             |      |  |

Minutes Council Meeting 2021.03.24

## ACTION REGISTER – OUTSTANDING RESOLUTIONS OF COUNCIL (PUBLIC) AT 14 APRIL 2021

| Meeting<br>Date | ltem  | Status      | Action Required  | Assignee/s   | Action Taken  | Due Date   | Completed<br>(Overdue) |
|-----------------|---|-------------|--|--|---|------------|------------------------|
| 26/08/2020      | GOV1937 Electoral<br>System for 2022 and 2025<br>Local Body Elections   | In Progress | Work with Electoral Officer to include a poll asking for voter preference for STV/FPP alongside voting papers for the 2022 local elections.  | Committee Secretary,<br>Executive Advisor                              | Contacted Electoral Officer Anthony Morton of<br>Electionz for information. He will update our file,<br>noting the request to conduct the poll with the 2022<br>election. He indicated additional cost of approx<br>\$75,000, not including additional comms that will be<br>necessary.<br>Public Notice in ODT on 12/9/20 to meet legislative<br>requirements and to advise ORC intends to conduct<br>a poll on voting systems alongside the 2022 local<br>body elections. | 01/01/2022 |                        |
| 24/03/2021      | Notice of Motion -<br>Investigate resource needs<br>and timeline for possible<br>collaborative development<br>of an Otago Harbour Plan<br>with ORC, DCC, Port<br>Otago and Community<br>Input | Assigned    | Report back before completion of the LTP process on timeline and resourcing needs for collaborative development of an Otago Harbour Plan with ORC, DCC, Port Otago, iwi and community input.   | General Manager<br>Strategy, Policy and<br>Science                     |   | 27/05/2021 |                        |
| 24/03/2021      | SPS2114 ORC Submission<br>on Climate Change<br>Commission's Report  | Assigned    | Circulate to Councillors, a copy of the final submission on the Climate Change Commission's Report.  | General Manager<br>Strategy, Policy and<br>Science                     |   | 14/04/2021 |                        |
| 24/03/2021      | GOV2113 Local<br>Government New Zealand<br>Annual Conference 2021<br>Attendance   | Assigned    | Provide a report to on the 2021 LGNZ Annual Conference to the Governance,<br>Communications and Engagement Committee, August 2021.   | Chairperson  |   | 11/08/2021 |                        |
| 24/03/2021      | GOV2115 Delegations<br>COVID-19 Recovery (Fast<br>Track Consenting) Act<br>2020   | Assigned    | Amend the Council's Delegations Manual to include the delegation to make recommendations for panel members under the COVID-19 Recovery (Fast-Track Consenting) Act 2020 to the Chair of the Council.   | Chief Executive,<br>General Manager<br>Regulatory                      |   | 14/04/2021 |                        |
| 24/03/2021      | Recommendations of the<br>Data and Information<br>Committee   | Assigned    | Refer the report on Groundwater State of Environment to the Strategy and<br>Planning Committee for advice on - where there are issues highlighted (in red) in<br>the Discussions and Recommendations section of the report, what action if any,<br>are staff doing to rectify the situation. | General Manager<br>Strategy, Policy and<br>Science, Manager<br>Science |   | 12/05/2021 |                        |

#### 7.1. Draft ToR for Regional Co-Ordination Groups

| Prepared for: | Council                                  |
|---------------|--|
| Report No.    | OPS2102                                  |
| Activity:     | Environmental: Land                      |
| Author:       | Gavin Palmer, General Manager Operations |
| Endorsed by:  | Gavin Palmer, General Manager Operations |
| Date:         | 31 March 2021                            |

#### PURPOSE

[1] To approve the terms of reference for the Regional Coordination Group for Wilding Conifer Management in Otago and the Regional Coordination Group for Wallaby Management in Otago.

#### **EXECUTIVE SUMMARY**

- [2] Council is the "recipient" of central government funding for the National Wilding Conifer Programme in Otago and the National Wallaby Programme in Otago.
- [3] The funding agreements between government (Ministry for Primary Industries) and ORC require that ORC set up a Regional Coordination Group (RCG) for each programme.
- [4] This paper seeks approval of the terms of reference and proposed membership of each RCG.

#### RECOMMENDATION

That the Council:

- 1) **Receives** this report.
- 2) **Approves** the terms of reference and membership proposed for the Regional Coordination Group for Wilding Conifer Management in Otago and the Regional Coordination Group for Wallaby Management in Otago.

#### BACKGROUND

- [5] In late 2020 Council approved ORC entering into funding agreements for the National Wilding Conifer Programme in Otago<sup>1</sup> and the National Wallaby Programme in Otago<sup>2</sup>.
- [6] ORC's responsibilities under each agreement include the establishment of a Regional Coordination Group (RCG) for each programme. The purpose of each RCG is described in the agreements (attached).

#### DISCUSSION

<sup>&</sup>lt;sup>1</sup> National Wilding Pine Funding Round 2020-24, Otago Regional Council Report No. OPS1018, 23 September 2020.

<sup>&</sup>lt;sup>2</sup> National Wallaby Funding Round 2020-2024, Otago Regional Council Report No. OPS1023, 20 October 2020

- [7] The RCGs provide an opportunity for ORC to take a leadership role in the coordination of operational activity and to connect and communicate with key stakeholders on a more structured basis.
- [8] Draft terms of reference for each RCG have been prepared by staff having regard to the purpose described in each agreement and the additional outcomes ORC is seeking (attached).
- [9] Members of each RCG have been proposed based on organisations with large landholdings or representatives of such organisations. They have not yet been approached to check their interest and availability.

#### CONSIDERATIONS

#### Strategic Framework and Policy Considerations

[10] Not applicable.

#### **Financial Considerations**

[11] The cost to ORC of establishing and running the two RCGs is not funded by Central Government. Provision has been made in the Draft 2021/31 Long Term Plan to fund ORC's costs.

#### Significance and Engagement

[12] This matter does not trigger ORC's Policy on Significance and Engagement.

#### **Legislative and Risk Considerations**

[13] Establishing the RCGs is a requirement of the funding agreements.

#### **Climate Change Considerations**

[14] Not applicable.

#### **Communications Considerations**

[15] Establishment of the RCGs will assist communication between ORC, stakeholders and the community.

#### **NEXT STEPS**

[16] Approach the proposed members of each RCG to check their interest and availability in participating in the RCGs.

#### ATTACHMENTS

- 1. Terms of Reference Regional Coordination Group for Wilding Conifer Management [7.1.1 - 4 pages]
- 2. Terms of Reference Regional Coordination Group for Wallaby Management [**7.1.2** 4 pages]
- 3. MPI ORC Wilding Conifer Control Funding Agreement signed 20201106 [7.1.3 35 pages]
- 4. MPI ORC Wallaby Management Programme Funding Agreement signed 20201106 [7.1.4 - 43 pages]

Council Meeting 2021.04.14



### DRAFT TERMS OF REFERENCE REGIONAL CO-ORDINATION GROUP FOR WILDING CONIFER MANAGEMENT

#### 1. Purpose

The purpose of the Regional Co-ordination Group (RCG) is to ensure that:

- a) obligations under Ministry of Primary Industries (MPI) funding contract are met, namely, to ensure:
  - that operational activities (to be performed under the Management Unit Operational Plan (MUOP)) are planned in a cohesive and coordinated way
  - alignment between the activities carried out MPI funding and any other wilding conifer control work funded through other mechanisms
  - effective communication between the organisations responsible for wilding conifer management in Otago
  - that progress on the operational activities (to be performed under the MUOP) are communicated with local communities including relevant land occupiers; and
  - issues and risks are identified and dealt with or escalated to the Ministry of Primary Industries.

b) Otago Regional Council and its stakeholders' strategic priorities and operational delivery with respect to wilding conifer management are:

- Defined, prioritised, and communicated
- Allocated according to need
- Aligned with MPI directed work
- Open and transparent, taking onboard feedback from key stakeholders
- Evaluated for effectiveness.

#### 2. Reporting

Via the Chair, the Regional Co-ordination Group will provide quarterly updates to the Otago Regional Council's Implementation Committee and any other reporting as required by MPI.

The RCG is not a committee or sub-committee of Council and is not a decision-making body of Council.

Page 1 of 4



#### 3. Membership

| Group                        | Representation  | Role  |
|------------------------------|---|---|
| Otago Regional Council       | Cr Alexa Forbes   | Chair   |
| Kāi Tahu                     | via Aukaha (TBC with runaka)  |   |
| Landowner<br>Representatives | Wakatipu Wilding Conifer Control Group<br>Central Otago Wilding Conifer Control<br>Group<br>Land Information New Zealand<br>Department of Conservation<br>NZ Forest Owners Association  |   |
| Council Staff                | Andrea Howard<br>Richard Lord<br>Jess Thomson<br>Project Manager  | Operational Expertise (in<br>attendance)<br>Operational Expertise (in<br>attendance)<br>Meeting Support |
| Partner Agencies             | NZ Wilding Conifer Group<br>Te Uru Rākau (Forestry New Zealand)<br>Farm Forestry New Zealand<br>Environment Canterbury (TBC)<br>Dunedin City Council (TBC)<br>Waitaki District Council (TBC)<br>Central Otago District Council (TBC)<br>Queenstown District Council (TBC) |   |

#### 4. Role of the Chair

The role of the Chair is to:

- Ensure that the Terms of Reference are applied.
- Provide strategic guidance.
- Set the agenda and purpose for each meeting.
- Clarify and summarise what is happening throughout each meeting.
- Encourage broad participation from members in discussion.
- End each meeting with a summary of decisions and actions and assign responsibility.

Page 2 of 4



The Chair may temporarily delegate this role, as appropriate.

#### 5. Role of Members

Individual RCG members (including the Chair) have the following responsibilities:

- Understand the goals, objectives, and desired outcomes of the programme.
- Understand and represent the interests of programme stakeholders.
- Take a genuine interest in the programme's outcomes and overall success.
- Act on opportunities to communicate positively about the programme.
- Check that the programme is making sensible decisions especially in responding to issues, risks and proposed programme changes.
- Actively participate in meetings through attendance, discussion, and review of minutes, papers and other documents.
- Support open discussion and debate and encourage fellow members to voice their insights.

#### 6. Quorum and Decision-making

A minimum number of four RCG members are required for decision-making purposes.

#### 7. Conflict of Interest

Members must perform their functions in good faith, honestly and impartially, and avoid situations that might compromise their integrity or otherwise lead to conflicts of interest.

The Chair will ask members to declare any actual or perceived interests at the start of each meeting.

There is a recognition that landowner representatives will have a direct and unavoidable conflict by virtue of owning land.

Members are encouraged to raise real or perceived conflicts during any discussions related to land owned or managed by them. The Chair will then determine whether or not the interest represents a conflict, and if so, what action will be taken.

#### 8. Meeting Location

Meetings will alternate between in-person (Alexandra) and via technology (Teams). A technology option will be provided wherever possible.

#### 9. Meeting Frequency

The Regional Co-ordination Group will meet quarterly, or more frequent as required.

#### 10. RCG Timeframe

Page 3 of 4



The Group will meet for the duration of Ministry of Primary Industries national funding but may continue if funding ceases by mutual agreement of members.

#### 11. Agenda

Agenda items should include the following plus items the Chair deems necessary:

- Apologies
- Actions from Previous Minutes
- Update from each party on the status of the wilding conifer programme for their particular area
- Health and Safety
- Risks and opportunities
- Papers Requested by the RCG for next meeting
- Key messages for advocacy.

#### 12. Minutes of Meeting

Actions will be recorded by the Project Manager (provided by ORC) and circulated to all members and attendees. Action points/minutes must be submitted to the Chair within five working days of each meeting.

ORC will maintain a register of all Minutes.

Minutes from each meeting to be forwarded to attendees within seven working days of each meeting.

Page 4 of 4



#### DRAFT TERMS OF REFERENCE REGIONAL CO-ORDINATION GROUP FOR WALLABY MANAGEMENT

#### 1. Purpose

The purpose of the Regional Co-ordination Group (RCG) is to ensure that:

- a) obligations under Ministry of Primary Industries (MPI) funding contract are met, namely, to ensure:
  - that operational activities (to be performed under the Management Unit Operational Plan (MUOP)) are planned in a cohesive and coordinated way
  - that operational activities (to be performed under the MUOP) achieve the objectives in New Zealand Wallaby Management Strategy (once approved by the Governance Group)
  - alignment between the activities carried out in relation to the National Programme and any other wallaby management work funded through other mechanisms
  - effective communication between the organisations responsible for wallaby management in Otago
  - that progress on the operational activities (to be performed under the MUOP) are communicated with local communities including relevant land occupiers; and
  - issues and risks are identified and dealt with or escalated to the Ministry of Primary Industries.
- b) Otago Regional Council and its stakeholders' strategic priorities and operational delivery with respect to wallaby management are:
  - Defined, prioritised, and communicated
  - Allocated according to need
  - Aligned with MPI directed work
  - Open and transparent, taking onboard feedback from key stakeholders
  - Evaluated for effectiveness.

#### 2. Reporting

Via the Chair, the Regional Co-ordination Group will provide quarterly updates to the Otago Regional Council's Implementation Committee and any other reporting as required by MPI.

The RCG is not a committee or sub-committee of Council and is not a decision-making body of Council.

Page 1 of 4



#### 3. Membership

| Group                  | Representation                             | Role                                |
|------------------------|--|-------------------------------------|
| Otago Regional Council | Cr Kevin Malcolm                           | Chair                               |
| Kāi Tahu               | via Aukaha (TBC with runaka)               |                                     |
|                        | Maniototo Pest Company (TBC)               |                                     |
|                        | Land Information New Zealand (TBC)         |                                     |
| Landowner              | Meridian Energy (TBC)                      |                                     |
| Representatives        | Department of Conservation (TBC)           |                                     |
|                        | Federated Farmers (TBC)                    |                                     |
|                        | NZ Forest Owners Association (Phil Taylor) |                                     |
|                        | Andrea Howard                              |                                     |
| Council Staff          | Richard Lord                               | Technical Expertise (in attendance) |
|                        | Simon Stevenson                            | Technical Expertise (in attendance) |
|                        | Wallaby Project Manager                    | Meeting Support                     |
|                        | Environment Canterbury (TBC)               |                                     |
| Partner Agencies       | Waitaki District Council (TBC)             |                                     |
|                        | Central Otago District Council (TBC)       |                                     |

#### 4. Role of the Chair

The role of the Chair is to:

- Ensure that the Terms of Reference are applied.
- Provide strategic guidance.
- Set the agenda and purpose for each meeting.
- Clarify and summarise what is happening throughout each meeting.
- Encourage broad participation from members in discussion.
- End each meeting with a summary of decisions and actions and assign responsibility.

The Chair may temporarily delegate this role, as appropriate.

#### 5. Role of Members

Individual RCG members (including the Chair) have the following responsibilities:

Page 2 of 4



- Understand the goals, objectives, and desired outcomes of the programme.
- Understand and represent the interests of programme stakeholders.
- Take a genuine interest in the programme's outcomes and overall success.
- Act on opportunities to communicate positively about the programme.
- Check that the programme is making sensible decisions especially in responding to issues, risks and proposed programme changes.
- Actively participate in meetings through attendance, discussion, and review of minutes, papers and other documents.
- Support open discussion and debate and encourage fellow members to voice their insights.

#### 6. Quorum and Decision-making

A minimum number of four RCG members are required for decision-making purposes.

#### 7. Conflict of Interest

Members must perform their functions in good faith, honestly and impartially, and avoid situations that might compromise their integrity or otherwise lead to conflicts of interest.

The Chair will ask members to declare any actual or perceived interests at the start of each meeting.

There is a recognition that landowner representatives will have a direct and unavoidable conflict by virtue of owning land.

Members are encouraged to raise real or perceived conflicts during any discussions related to land owned or managed by them. The Chair will then determine whether or not the interest represents a conflict, and if so, what action will be taken.

#### 8. Meeting Location

Meetings will alternate between in-person (Oamaru/Ranfurly) and via technology (Teams). A technology option will be provided wherever possible.

#### 9. Meeting Frequency

The Regional Co-ordination Group will meet bi-monthly, or more frequent as required.

#### 10. RCG Timeframe

The Group will meet for the duration of Ministry of Primary Industries national funding but may continue if funding ceases by mutual agreement of members.

#### 11. Agenda

Agenda items should include the following plus items the Chair deems necessary:

Page 3 of 4



- Apologies
- Actions from Previous Minutes
- Update from each party on the status of the wallaby programme for their particular area
- Risks and opportunities
- Papers Requested by the RCG for next meeting
- Key messages for advocacy.

#### 12. Minutes of Meeting

Actions will be recorded by the Project Manager (provided by ORC) and circulated to all members and attendees. Action points/minutes must be submitted to the Chair within five working days of each meeting.

ORC will maintain a register of all Minutes.

Minutes from each meeting to be forwarded to attendees within seven working days of each meeting.

Page 4 of 4

Ministry for Primary Industries Manatū Ahu Matua



# **Funding Agreement**

in relation to Wilding Conifer Control

Between Ministry of Primary Industries (**MPI**) and Otago Regional Council (**Recipient**)

# **Funding Agreement**

## Wilding Conifer Control

| Backgro<br>Agreed<br>PART A<br>1. C<br>2. A<br>3. A<br>3.1 In<br>3.2 F<br>4. F<br>4.1 P<br>4.2 M<br>4.4 C<br>5. G<br>5.1 F<br>6. T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  | d terms   | 5  |
|--|---|----|
| Agree<br>PART A<br>1. C<br>2. A<br>3. A<br>3.1 Ir<br>3.2 R<br>4. F<br>4.1 P<br>4.2 N<br>4.3 N<br>4.4 C<br>5. G<br>5.1 R<br>6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  |   | 6  |
| PART A<br>1. C<br>2. A<br>3. A<br>3.1 In<br>3.2 F<br>4. F<br>4.1 P<br>4.2 M<br>4.3 M<br>4.4 C<br>5. G<br>5.1 F<br>6. T<br>6. T<br>6. T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | OPERATIONAL AND COMMERCIAL DETAILS                          |    |
| 1.       C         2.       A         3.1       Ir         3.2       F         4.       F         4.1       P         4.2       M         4.3       M         4.4       C         5.1       F         6.       T         6.1       T         6.2       T         7.       C         PART B       1.         2.       C         2.1       M   | OFERATIONAL AND COMMERCIAL DETAILS                          | 6  |
| 2. A<br>3. A<br>3.1 Ir<br>3.2 R<br>4. F<br>4.1 P<br>4.2 M<br>4.3 M<br>4.4 C<br>5. G<br>5.1 R<br>6. T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  | Dutcomes  | 6  |
| 3.       A         3.1       Ir         3.2       R         3.2       R         4.       F         4.1       P         4.2       M         4.3       M         4.4       C         5.       G         5.1       R         6.1       T         6.2       T         6.3       E         7.       C         PART B       1.         4.       C         2.       C         2.1       M | cknowledgement of appropriation                             | 6  |
| 3.1       Ir         3.2       R         4.       F         4.1       P         4.2       M         4.3       M         4.4       C         5.1       R         6.1       T         6.2       T         6.3       E         7.       C         PART B       1.         2.       C         2.1       M  | scivities   | 6  |
| 3.2       R         4.       F         4.1       P         4.2       M         4.3       M         4.4       C         5.       G         5.1       R         6.1       T         6.3       E         7.       C         PART B       1.         4.       C         2.       C         2.1       M   | scope   | 6  |
| 4. F<br>4.1 P<br>4.2 M<br>4.3 M<br>4.4 C<br>5. G<br>5.1 R<br>6.1 T<br>6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | estricted and/or Prohibited Activities                      | 12 |
| 4.1 P<br>4.2 M<br>4.3 M<br>4.4 C<br>5. G<br>5.1 R<br>6.1 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | unding  | 12 |
| 4.2 M<br>4.3 M<br>4.4 C<br>5. G<br>5.1 R<br>6.1 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  | urpose  | 12 |
| 4.3 M<br>4.4 C<br>5. G<br>5.1 R<br>6. T<br>6.1 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | faximum amount  | 12 |
| 4.4 C<br>5. G<br>5.1 F<br>6. T<br>6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | lethod of payment   | 12 |
| 5. G<br>5.1 R<br>6. T<br>6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  | ther provisions   | 12 |
| 5.1 R<br>6. T<br>6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  | Sovernance  | 13 |
| 6. T<br>6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | ole and responsibilities                                    | 13 |
| 6.1 T<br>6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M   | erm and termination   | 13 |
| 6.2 T<br>6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 M  | erm   | 13 |
| 6.3 E<br>7. C<br>PART B<br>1. A<br>2. C<br>2.1 N   | ermination of this Agreement                                | 13 |
| 7. C<br>PART B<br>1. A<br>2. C<br>2.1 N  | xpiry of existing Funding Agreement                         | 13 |
| PART B<br>1. A<br>2. C<br>2.1 N  | contact details   | 14 |
| 1. A<br>2. C<br>2.1 M  | GENERAL TERMS AND CONDITIONS                                | 15 |
| 2. C   | ctivities   | 15 |
| 2.1 N  | bligations  | 15 |
|  | lutual obligations of the parties                           | 15 |
| 2.2 R  | ecipient's general obligations                              | 15 |
| 2.3 N  | IPI's obligations   | 15 |
| 3. H   | lealth and safety   | 16 |
| 3.1 C  | ompliance with health and safety legislation and directions | 16 |
| 3.2 H  | ealth and safety policies, practices and plans              | 16 |
| 4. S   | ubcontractors and Personnel                                 | 17 |
| 4.1 S  | ubcontractors   | 17 |
| 4.2 P  | ersonnel  | 18 |
| 5. S   | ignificant Events process                                   | 18 |
| 5.1 N  | otification of Significant Event                            | 18 |
| 5.2 R  | ectification process  | 19 |
| 5.3 C  | ost of remedying  | 19 |
| 5.4 C  | ther remedies   | 20 |
| 6. P   | ayment  | 20 |
| 6.1 N  | IPI's obligation to pay the Funding                         | 20 |
| 6.2 T  | otal amount payable   | 20 |
| 0,3 Ir   | ivoicing  | 20 |
| 6.4 L  | ligible Expanditure   | 20 |
| 66 E   | ingnois Experioritore                                       | 20 |
| 0.0 1  | inerroler menagement  | 21 |

National Wilding Conifer Control Programme

|      | A CONTRACTOR OF | 24 |
|------|---|----|
| 6.7  | Funding Freeze  | 21 |
| 6,8  | Repayment   | 21 |
| 6,9  | Insumcient Funding  | 22 |
| 6.10 | Tax .   | 22 |
| 7.   | Change control procedure  | 22 |
| 7.1  | Change request  | 22 |
| 7.2  | Understanding consequence of change request   | 22 |
| 8.   | Information management  | 23 |
| 8.1  | Provision of Information  | 23 |
| 8.2  | Privacy   | 23 |
| 8.3  | Recordkeeping   | 23 |
| 9.   | Confidentiality   | 24 |
| 9.1  | Security and non-disclosure   | 24 |
| 9,2  | Permitted disclosure  | 24 |
| 9.3  | Information sharing   | 24 |
| 9.4  | Mandatory disclosures   | 24 |
| 9.5  | Publicity and announcements   | 24 |
| 10.  | Warranties  | 25 |
| 10.1 | Warranties  | 25 |
| 11.  | Intellectual Property and data  | 25 |
| 11.1 | Background Intellectual Property  | 25 |
| 11.2 | New Intellectual Property   | 25 |
| 11.3 | Licence   | 25 |
| 11.4 | WCIS  | 26 |
| 11.5 | Know How  | 26 |
| 12   | Third party rights  | 26 |
| 12.1 | Third Party Claim   | 26 |
| 13   | Conflicts of interest   | 26 |
| 12.1 | Avoiding conflicts  | 26 |
| 13.1 | Notifying conflicts   | 26 |
| 10.2 | Audia   | 26 |
| 14.  | Audit   | 20 |
| 14.1 | Audit   | 20 |
| 15.  | Dispute resolution  | 2/ |
| 15.1 | Negotiation   | 27 |
| 15.2 | Escalation  | 27 |
| 15.3 | Mediation   | 27 |
| 15.4 | Right to seek relief  | 27 |
| 16.  | Liability   | 27 |
| 16.1 | Maximum liability   | 27 |
| 16.2 | Exclusions  | 28 |
| 16.3 | Uncapped liability  | 28 |
| 16.4 | Mitigation  | 28 |
| 17.  | Force Majeure   | 28 |
| 18.  | Insurance   | 28 |
| 19.  | Term and termination  | 28 |
| 19.1 | Term  | 28 |
| 19.2 | Termination for default and insolvency  | 29 |
| 19.3 | Effects of expiry or termination  | 29 |
| 19.4 | Survival clauses  | 29 |
| 20.  | Notices   | 29 |
| 21.  | General   | 29 |
|      | I CE ALA CAR  |    |
|      | National Wilding Contral Programme  |    |

National Wilding Conifer Centrol Programme

| Execution                                   |  |
|---|--|
|   |  |
| Schedule 1 – Definitions and Interpretation |  |

National Wilding Conifer Control Programme

.

## Details

#### Date Parties The Sovereign in right of New Zealand acting by and through The Ministry of Name Primary Industries with delegated financial authorisation to enter into the agreement, by the Chief Operational Officer & Head of Biosecurity, Ministry for **Primary Industries** Short form name MPI Notice details Address: PO Box 2526, Weilington 6140 Email: Wilding.Conifers@mpi.govt.nz Attention: National Wilding Conifer Control Programme Otago Regional Council, a body corporate under the Local Government Act 2002 Name having its head office at 70 Stafford Street, Dunedin Short form name Recipient Notice details Address: 70 Stafford Street, Dunedin Email: Sarah.Gardner@orc.govt.nz Attention: Sarah Gardner

#### Background

A. Wilding conifers are a serious and increasing pest issue for New Zealand. The issue is one that can only be addressed through a collaborative national control programme. The programme described in the New Zealand Wilding Conifer Management Strategy 2015-2030 (the National Programme) was created to support the effective collaboration between land occupiers, researchers, regulators, iwi, hapu and communities to address the critical overarching issues facing wilding conifer management.

- B. The New Zealand Government has allocated funding for the National Programme, to be distributed through MPI.
- C. MPI wishes to distribute funding on a regional basis to selected territorial authorities for the purpose of that territorial authority carrying out activities that help eradicate and/or control the spread of wilding conifers, as part of the National Programme. Each Region will comprise specific Management Units within which specific activities will be carried out.
- D. The Recipient is a territorial authority who wishes to eradicate and/or control the spread of wilding conifers in its Region.
- E. As MPI and the Recipient are part of the public sector, each wishes to ensure that there is appropriate oversight across the Recipient's activities and use of public funding, focus on health and safety, and sharing of useful information.
- F. The Recipient and MPI were parties to a funding agreement in relation to the National Programme, which has now expired. The purpose of this Agreement is to record the terms on which MPI will distribute the Funding to the Recipient under this Agreement and the terms on which the Recipient will apply the Funding and ensure that the Activities are carried out (by either itself or through others), in each case, for the purpose of the National Programme.

National Wilding Conifer Control Programme

# Agreed terms

#### PART A: OPERATIONAL AND COMMERCIAL DETAILS

#### 1. Outcomes

- (a) The intended outcomes for this Agreement (the Outcomes) are to:
  - (i) help eradicate and/or control the spread of wilding conifers in New Zealand;
  - support the implementation of the New Zealand Wilding Conifer Management Strategy 2015-2030, in particular to:
    - (A) co-ordinate operations by faciliating co-ordinated control amongst land owners and other stakeholders; and
    - (B) promote information sharing of best practice and technological gains in control methods;
  - (iii) support the Recipient to tailor Outcomes to reflect:
    - the regional strategies for the Region for which the Recipient is responsible; and
    - (B) feedback from the Regional Coordination Group and the Recipient's Fund Managers for the Region for which the Recipient is responsible;
  - build and maintain a collaborative and constructive relationship between the parties and other stakeholders to address and manage the critical overarching issues caused by wilding conifers;
  - support the Recipient to ensure the health and safety of Workers and others involved in and/or affected by the Activities, as required by law; and
  - (vi) for the Region in which the Recipient operates, enable oversight of the operational programme and use of the Funding.
- (b) In performing the Activities, the Recipient will use all reasonable endeavours to contribute to the achievement of the Outcomes.

#### Acknowledgement of appropriation

Each party acknowledges that the Funding under this Agreement is provided under the 'Border and Domestic Biosecurity Risk Management MCA appropriation of Vote Agriculture, Biosecurity, Fisheries and Food Safety'. The scope of the appropriation is 'Biosecurity Incursion Response and Long-Term Pest Management' (Scope of Appropriation). Regardless of any other clause in this Agreement, the Recipient agrees that MPI does not provide the Funding, and the Recipient will not use the Funding, for any purpose outside the Scope of Appropriation.

#### 3. Activities

#### 3.1 In scope

The Recipient will perform the Activities and provide the corresponding Outputs, as each is set out in the table below. Where indicated in that table, the Recipient:

- may use the Funding to perform the Activity, where indicated as "Funded" in the corresponding column (a Funded Activity); and
- (b) will perform each Activity at its cost, where indicated as "Recipient's cost" in the corresponding column.

National Wilding Conifer Control Programme

|     | Activity   | Output  | Funded or Recipient's<br>cost |
|-----|--|---|-------------------------------|
|     | Operation  | al Planning   |                               |
| (a) | For each Financial Year during the term of this Agreement, and for each<br>Management Unit in the Region that the Recipient is responsible for, prepare<br>(or procure the preparation of) a draft Management Unit Operational Plan<br>using the template provided by MPI.   | Each draft Management Unit Operational Plan is reviewed and<br>endorsed in writing by the:<br>1. RCG; and<br>2. MPI Programme Manager.  | Recipient's cost              |
| (b) | <ul> <li>If one does not already exist, set up a Regional Coordination Group for each Region and:</li> <li>(i) ensure that the RCG members adequately represent existing groups that have been established to manage wilding conifers in that Region; and</li> <li>(ii) ensure the RCG meets periodically for the purpose of oversaeing the planning and work completed (or to be completed) under the National Programme in that Region.</li> </ul> | The Regional Coordination Group operates effectively to<br>represent the key stakeholders of that particular Region<br>and the Management Unit(s) within that Region.<br>The RCG carries out its responsibilities in accordance<br>with any agreed "Terms of Reference" for that RCG. | Recipient's cost              |
| (c) | Implement (or procure the Implementation of) the Management Unit<br>Operational Plan in accordance with its terms, including by engaging<br>Personnel and Subcontractors to implement that plan and help deliver the<br>Outcomes.  |   | Funded                        |
| (d) | Actively manage its Personnel angaged for the purpose of implementing the<br>relevant Management Unit Operational Plan and verify that their work and<br>engagement is carried out in accordance with the terms of that plan and this.<br>Agreement.   |   | Recipient's cost              |

National Wilding Casilar Control Programma

 $\tau$ 

|     | Activity   | Output   | Funded or Recipient's<br>cost         |
|-----|--|--|---------------------------------------|
|     | Financial and Op   | erational Reporting  |                                       |
| (e) | Comply with the WCIS user guide provided by MPI to the Recipients, as may<br>be amended from time to time (WCIS User Guide).   | Keep accurate, up to date records showing the delivery of the<br>Activities against the relevant plan, including by taking minutes | Recipient's cost                      |
| (1) | Ensure that all data is uploaded to WCIS in an accurate and timely manner,<br>in each case, as described and in accordance with the WCIS User Guide.   | Provide adequate information and reports to MPI at meetings.   |                                       |
| (g) | Input (or procuring the inputting of) the financial data into WCIS, in accordance with the WCIS User Guide, and do so accurately to reflect (amongst other things) the costs of completed operations, and:<br>(i) Funding (i) Funding (ii) Funding (iii) Funding (iiii) Funding (iii) Fundin |  |                                       |
|     | (iii) In Kind Contributions.   |  |                                       |
| (h) | Comply with any audit under clause 14.1 of Part B.   |  | Funded (subject to<br>clause 14.1(e)) |
|     | Health and Saf   | ety Management   |                                       |
| (i) | Work with MPI to ensure a Health and Safety Risk Management Plan<br>(HSRMP) is in place and agreed prior to the commencement of any Funded<br>Activity.  | HSRMP is documented and agreed with MPI prior to the<br>commencement of any Funded Activity.                                       | Recipient's cost                      |
| (1) | For each PCBU, ensure a HSRMP is in place prior to the commencement of<br>any Funded Activity, which:  |  | Recipient's cost                      |
|     | <ul> <li>(i) meets the relevant requirements of the HSRMP agreed with MPI in (i);</li> <li>(ii) identifies and manages specific health and safety risks and hazards for<br/>the work being (or to be) carried out;</li> <li>(iii) captures the health and safety risks and hazards arising under and in<br/>relation to the work being (or to be) carried out under and in relation to<br/>this Agreement.</li> </ul>  | 19   |                                       |
|     | Make the HSRMP for each PCBU available to MPI when requested.  |  |                                       |

National Wilding Conifer Control Programme

|     | Activity  | Output   | Funded or Recipient's<br>cost |
|-----|---|--|-------------------------------|
| (k) | Ensure each Health and Safety incident is reported to MPI within the timeframes below:  |  | Recipient's cost              |
|     | <ul> <li>Notifiable Event within 24 hours of becoming aware of the Notifiable<br/>Event and following confirmation of event being notified to WorkSafe NZ;</li> <li>Serious incident within 24 hours of becoming aware of the serious<br/>incident</li> <li>Incident within 5 Business Days of becoming aware of the incident;</li> <li>Near Miss within 5 Business Days of becoming aware of the near miss.</li> </ul> |  |                               |
|     | Notifiable Events and serious incidents will be notified to the MPI Operations<br>Lead (or delegate) by telephone in order to confirm investigation roles,<br>responsibilities and actions.   |  |                               |
|     | Incidents or near misses should include the results of the health and safety<br>investigation that followed the incident or near miss and be submitted by<br>inputting the relevant information into WCIS in accordance with the WCIS<br>User Guide.  |  |                               |
| (I) | Ensure health and safety audits are completed and any required<br>improvements that are identified during an audit are implemented.<br>Make the results of audits available to MPI when requested.  |  | Recipient's cost              |
| (m) | Review and update the HSRMP agreed under this Agreement in periodic<br>meetings with MPI.   | Meet at frequency agreed with MPI in order to cooperatively<br>ensure that health and safety hisks and hazards related to<br>operations are being managed appropriately. | Recipient's cost              |
|     | Reporting t   | l<br>and meetings  |                               |
| (n) | Attend any meeting with MPI that MPI reasonably request.  |  | Recipient's cost              |

Nevonal Wilding Conifer Control Programmer

-4

|     | Activity   | Output | Funded or Recipient's<br>cost |
|-----|--|--------|-------------------------------|
| (a) | Report to MPI in accordance with the "Operational Meeting & Reporting<br>Cycle" table below (or as otherwise agreed in writing between the parties). At<br>any "face to face" meeting, each party will endeavour to meet physically in<br>person or, where a physical meeting is not possible or desirable, by video-<br>conference instead. | 1      | Recipient's cost              |

|         | 1.              | New Ottesting  | Automations         |
|---------|-----------------|--|---------------------|
| Month   | Type            | Meeting Objectives   | Attendees           |
| July    | Conference call | Confirm whether all previous Financial Year data is in WCIS.   | Fund Manager        |
|         |                 | Confirm complete operations and spend for Financial Year (includes journals for expected invoices).                      | MPI Operations Lead |
|         |                 | Confirm financial contributions from each co-funding partner and land-occupier, as identified in the MUOP<br>(Partners). | MPI Coordinator     |
|         |                 | Confirm In Kind Contributions from Partners.   |                     |
|         |                 | Agree actions and next meeting agenda.   | 1                   |
| August. | Face to Face    | MPI present and discuss key National Programme processes, responsibilities and expectations.                             | Fund Manager        |
|         |                 | Confirm forecast financial contributions from Partners.  | MU Manager/s        |
|         |                 | Agree the MU Manager for each MU.  | MPI Operations Lead |
|         |                 | Confirm Subcontractor/s for each operation (or agree dates when this will be confirmed).                                 | MPI Coordinator     |
|         |                 | Confirm that a HSRMP is in place for each MU (or agree dates when this can be completed).                                |                     |
|         |                 | Agree actions and next meeting agenda.   |                     |
| October | Conference Call | Current operations and spend - % complete  | Fund Manager        |
|         | 1               | Identify and agree resolution actions for key operational issues.  | MPI Operations Lead |
|         |                 | Review HSRMPs and health and safety risks/issues.  | MPI Coordinator     |
|         |                 | Agree actions and next meeting agenda.   |                     |

National Wilding Conifer Control Programme

|             |                 | Operational Meeting & Reporting Cycle – Recipients   |  |
|-------------|-----------------|--|--|
| December    | Conference call | Current operations and spend - % complete.<br>Review risks/issues (includes health and safety).<br>Confirm financial contributions received from Partners (YTD).<br>Confirm 'in kind' contributions made by Partners (YTD).<br>Agree actions and next meeting agenda.  | Fund Manager<br>MPI Operations Lead<br>MPI Coordinator |
| February    | Conference call | Current operations and spend - % complete.<br>Review risks/issues (includes health and safety).<br>Agree actions and next meeting agenda.  | Fund Manager<br>MPI Operations Lead<br>MPI Coordinator |
| April - May | Face to Face    | Agree Management Unit operational planning process for upcoming Financial Year.<br>Confirm date for operational plans to be submitted.<br>Current operations and spend - % complete.<br>Review risks/issues (includes health and safety).<br>Forecast budget for upcoming Financial Year.<br>Agree actions and next meeting agenda.                    | Fund Manager<br>MPI Operations Lead<br>MPI Coordinator |
| June        | Conference call | Confirm approved operations and budget for upcoming Financial Year (or actions required to complete).<br>Confirm actions to complete sign off for any variations to this Agreement and the supporting HSRMP.<br>Agree actions and next meeting agenda.<br>Ágree upcoming Financial Year meetings schedule.<br>Agree annual debrief date and attendoes. | Fund Manager<br>MPI Operations Lead<br>MPI Coordinator |

Nellanel Wilding Caniler Coninci Programme

#### 3.2 Restricted and/or Prohibited Activities

The Recipient must not use the Funding for the following prohibited activities (**Prohibited Activities**). MPI may add, remove and/or amend any of the Prohibited Activities from time to time by giving written notice to the Recipient. Provided that MPI has consulted with the Governance Group in relation to a potential Prohibited Activity, any notification of that Prohibited Activity by MPI will take effect as from the date of notification.

| Prohibited Activity   | Comments                     |  |
|---|------------------------------|--|
| The use of mineral based diesel (including<br>blends of mineral diesel and vegetable oil),<br>fuels and oils as carriers in Aerial Basal Bark<br>Application ("ABBA") operations under the<br>National Programme. | Effective as at 1 July 2020. |  |

#### 4. Funding

#### 4.1 Purpose

The Recipient will only use the Funding for the purpose of carrying out the Funded Activities.

#### 4.2 Maximum amount

Regardless of anything else in this Agreement, MPI will not be obliged to provide Funding under this Agreement which, in total, exceeds the following amount for the corresponding Management Unit for the corresponding Financial Year. The amounts in the table may change from Financial Year to Financial Year by written notice from MPI, which may be given only after consultation with the Governance Group:

| Management Unit | Financial Year | Maximum amount of<br>Funding (plus GST if any) |
|-----------------|----------------|--|
| Alexandra       | 2020/21        | \$62,175                                       |
| Dunstan         | 2020/21        | \$9,205  |
| Glenorchy       | 2020/21        | \$300,000                                      |
| Kawarau         | 2020/21        | \$682,075                                      |
| Lammermoor      | 2020/21        | \$174,798                                      |
| Northern Eyre   | 2020/21        | \$392,275                                      |
| Remarkables     | 2020/21        | \$460,355                                      |
| Rough Ridge     | 2020/21        | \$73,871                                       |
| Shotover        | 2020/21        | \$869,755                                      |
| Naseby          | 2020/21        | \$336,825                                      |
| Wakatipu        | 2020/21        | \$2,540,395                                    |

#### 4.3 Method of payment

The Recipient will invoice MPI by sending a valid Tax Invoice to the MPI Operations Lead in accordance with clause 6 of Part B. Each Tax Invoice may only be issued when the invoicing criteria (as notified by MPI to the Recipient from time to time) for that Tax Invoice has been met to MPI's reasonable satisfaction.

#### 4.4 Other provisions

Further detail regarding the method of payment, eligible expenditure, financial management, funding freezes and repayment is set out in clause 6 of Part B.

National Wilding Conifer Control Programme
# 5. Governance

# 5.1 Role and responsibilities

The party described as "appointor" in the table below will engage and keep engaged suitably qualified, skilled and experienced Personnel in the roles set out in the table below. Each party will ensure its appointee performs the responsibilities for that role, as described in the table below and elsewhere in this Agreement.

| Name of Role   | Responsibilities  | Appointer  |
|--|---|--|
| Representative on<br>the Regional<br>Coordination<br>Group | <ol> <li>Have oversight and involvement in<br/>the development, endorsement-<br/>seeking process, implementation<br/>and monitoring of each relevant<br/>MUOP.</li> </ol> | The Recipient  |
|  | <ol> <li>Keeps up to date on progress<br/>towards delivering the MUOP.</li> <li>Is involved in risk mitigation (all<br/>informed and involved).</li> </ol>                |  |
|  | <ol> <li>Assist in coordination of control<br/>work (and other related work).</li> </ol>  | (  |
| Representation on<br>the Governance<br>Group               | To help govern the National<br>Programme, and oversee associated<br>Crown funding and National Programme<br>operational delivery.   | MPI and, where the Recipient<br>is part of LGNZ, through the<br>Recipient's LGNZ<br>representative |

# 6. Term and termination

# 6.1 Term

| Start Date             | 1 July 2020  |
|------------------------|--|
| End Date               | 30 June 2024   |
| Extending the End Date | The parties may extend the term of this<br>Agreement by written agreement. |

# 6.2 Termination of this Agreement

In addition to the rights contained in clause 19 of Part B, MPI or the Recipient may terminate this Agreement at any time for convenience, without cause, by giving at least four (4) months' prior written notice to the other party.

# 6.3 Expiry of existing Funding Agreement

- (a) The parties acknowledge that the previous funding agreement between the parties dated 1 December 2016 (Previous Funding Agreement) has expired.
- (b) The Recipient agrees that any policy, instruction or guidance given by MPI under the Previous Funding Agreement is deemed to be given and continue under this Agreement also, unless MPI has expressly replaced or withdrawn that policy, instruction and/or guidance. To the extent that there is any inconsistency between such policy, instruction, or guidance given under the Previous Funding Agreement (or any policy, instruction, or guidance given by MPI under this Agreement), the terms of this Agreement (or the terms of any such policy, instruction or guidance given by MPI under this Agreement, as the case may be) will prevail.
- (c) If the Recipient has not spent all of the funding allocated to the Recipient by MPI under the Previous Funding Agreement, MPI and the Recipient agree that, the unspent funding will be deemed to be provided by MPI as part of (but not in addition to) the Funding provided under this Agreement.

National Wilding Conifer Control Programme

# 7. Contact details

- (a) All matters or enquiries regarding the technical implementation or operation of this Agreement will be directed to MPI's Technical Liaison or the Recipient's Technical Liaison, as appropriate.
- (b) All general matters relating to this Agreement (including concerning the interpretation of this Agreement) will be directed to MPI's Contract Manager or the Recipient's Contract Manager, as appropriate.
- (c) If the Recipient cannot contact MPI's Contract Manager or MPI's Technical Liaison within a reasonable timeframe in the circumstances, enquiries can be directed to the person for the time being holding the office of "MPI Manager Recovery and Pest Management".
  - (d) If MPI cannot contact the Recipient's Contract Manager or the Recipient's Technical Liaison, enquiries can be directed to the Recipient's Chief Executive or Managing Director.
  - (e) The initial contact persons for each party are below. If a party's contact persons or their details change, that party must notify the other party in writing 10 Business Days in advance.

| Party     | Role                                    | Details  |
|-----------|---|--|
| MPI       | Manager Recovery and Pest<br>Management | John Sanson<br>04 894 0836<br>John Sanson@mpi.govt.nz                            |
| MPI       | Technical Liaison                       | Alan Tinnelly<br>029 943 3706<br>Alan.Tinnelly@mpi.govt.nz                       |
| MPI       | Contract Manager                        | Sherman Smith<br>04 894 0831<br>Sherman.Smith@mpi.govt.nz                        |
| Recipient | Chief Executive                         | Sarah Gardner<br>Sarah.Gardner@orc.govt.nz                                       |
| Recipient | Managing Director                       | Gavin Palmer<br>General Manager of Operations<br><u>Gavin.Palmer@orc.govt.nz</u> |
| Recipient | Technical Liaison                       | Richard Lord<br>03 474 0827<br>Richard.Lord@orc.govt.nz                          |
| Recipient | Contract Manager                        | Andrea Howard<br>Andrea.Howard@orc.govt.nz                                       |

National Wilding Conifer Control Programme

# PART B: GENERAL TERMS AND CONDITIONS

1. Activities

MPI agrees to provide the Funding to the Recipient, and the Recipient agrees to use the Funding and carry out the Activities (or ensure the Activities are carried out), subject to and in accordance with the terms of this Agreement, and will use the Funding solely for the purpose of performing the Funded Activities under the National Programme.

Obligations

(i)

# 2.1 Mutual obligations of the parties

- (a) Each party agrees to:
  - act in good faith in all matters relating to this Agreement and, without abandoning their own interests, to demonstrate honesty, integrity, openness, reasonableness, and accountability in their dealings with each other; and
  - discuss any matters affecting this Agreement or the performance of the Activities, whenever necessary.
- (b) Each party will ensure it performs its obligations under this Agreement:
  - with due diligence, care and skill and using a standard of skill, care and diligence no lower than that expected of an experienced and well-resourced organisation performing that obligation;
  - (ii) in accordance with the specified timeframes or, if no timeframe is specified, then within a reasonable period; and
  - (iii) in accordance with all applicable laws, regulations and, where legally binding upon that party, codes and standards of practice.

#### 2.2 Recipient's general obligations

- (a) The Recipient will ensure that it performs the Activities, and will ensure that any Personnel engaged to provide the Activities does so:
  - (i) using appropriately trained, qualified, experienced and supervised persons;
  - (ii) in accordance with all Government or MPI internal policies and procedures relevant to this Agreement, as notified in writing to the Recipient at or before the Start Date;
  - (iii) in accordance with the Standards of Integrity and Conduct issued by the State Services Commission (see <u>www.ssc.govt.nz</u>); and
  - in a manner that:
  - (iv) is culturally appropriate for Maon, Pacific and other ethnic or indigenous groups;
  - (v) respects the personal privacy and dignity of all participants and stakeholders; and
  - (vi) meets MPI's satisfaction, as reasonably described by MPI in writing from time to time.
- (b) Unless otherwise agreed in writing, the Recipient will ensure that it obtains all necessary and prudent authorisations, consents, permits, licences and the like, to ensure the Recipient can perform its obligations under this Agreement in compliance with the law and any third party's rights, including in relation to performance of any Activities carried out on MPI's or a third party's premises.

#### 2.3 MPI's obligations

If reasonably requested by the Recipient, and without limiting the Recipient's obligations under this Agreement, MPI will:

 (a) cooperate and collaborate with the Recipient in the same way that MPI generally cooperates and collaborates with other recipients similar to the Recipient in the relevant circumstances;

National Wilding Conifer Control Programme

- (b) provide or give access to MPI's resources and information, including templates, which MPI generally makes available to recipients similar to the Recipient, for the purpose of the National Programme; and
- (c) endeavour to find opportunities to share best practices and lessons learned across the National Programme's delivery.
- 3. Health and safety
- 3.1 Compliance with health and safety legislation and directions
  - (a) During the term of this Agreement, the Recipient will:
    - consult, cooperate and coordinate with MPI to ensure that the parties comply with their respective obligations under HSWA Legislation as they relate to this Agreement;
    - perform its obligations under this Agreement, and ensure that its Personnel performs their respective obligations in relation to this Agreement, in compliance with the HSWA Legislation;
    - comply with all reasonable written directions given by MPI from time to time relating to health and safety on any matters not addressed in the Recipient's Health and Safety Risk Management Plan;
    - (IV) perform its, and ensure that each of its Personnel perform their respective, obligations under or in relation to this Agreement in compliance with the HSWA Legislation, including the:
      - (A) duties of a PCBU under sections 36-43 of the HSWA;
      - (B) duties relating to the identification of hazards and implementation of Control Measures under the HSWA (GR&WM) Regulations 2016; and
      - (C) duties relating to the keeping of records under section 57 of the HSWA; and
    - duly notify WorkSafe NZ of any Notifiable Event, as required under section 56 of the HSWA.
  - (b) MPI acknowledges that any Worker engaged by the Recipient in the performance of the Activities has a statutory right to cease or refuse to carry out unsafe work, as that term is defined under section 82 of the HSWA.

# 3.2 Health and safety policies, practices and plans

- (a) During the term of this Agreement, the Recipient will:
  - maintain general health and safety policy and practices that are appropriate to the nature of the Activities; and
  - comply with its health and safety policy and practices, and ensure its Workers and other Personnel also comply.
- (b) Within 10 Business Days after MPI has supplied the HSRMP template to the Recipient (or any longer period that parties agree in writing), the Recipient will submit a draft written HSRMP to MPI for review. The Recipient will ensure the draft is a duly completed template provided by MPI or complies with clause 3.2(c) below.
- (c) If the Recipient does not use a template provided by MPI to document the Recipient's HSRMP, then the Recipient will ensure that its draft HSRMP includes (at a minimum) the following matters:
  - the "Health & Safety Risk Register", specific to performance of the work contemplated under this Agreement, including identified risks and mitigations, and assigned responsibilities for managing those risks;
  - (ii) confirmation of how those risks will be communicated to Workers;
  - clearly assigned health and safety roles and responsibilities (the name of the person responsible, rather than position titles);
  - (iv) details of regular meetings to discuss health and safety;

National Wilding Conifer Control Programme

1B

- (v) details of regular reporting (at least monthly) on any health and safety Notifiable Event (including near misses) and corrective actions;
- (vi) required personal protective equipment (PPE), and assigned responsibility for checking PPE;
- any applicable standard operating procedures (step-by-step instructions on key tasks) in place and how Workers access these;
- (viii) the name of any and all Workers engaged to assist with the work (and how they will be managed by or on behalf of the Recipient);
- (ix) the health and safety induction process for Workers involved in work under or in relation to this Agreement;
- identification of required training and/or certification to carry out the work (and confirmation of who checks that training and/or certification has been completed);
- (xī) emergency procedures, specific to the Activities; and
- (xii) details on how the Recipient's health and safety performance will be monitored (key performance indicators; site inspections).
- (d) After the Recipient submits the draft HSRMP to MPI, MPI will review and suggest any changes to the HSRMP. Where MPI suggests a change that would cause the Recipient to incur direct new or increased costs, or that the Recipient considers is unreasonable, the Recipient will promptly advise MPI of the reason why and of any additional funding it considers it would need to implement the change. The parties will endeavour to agree and finalise the draft HSRMP within a further 10 Business Days (or any longer period agreed in writing by the parties). The draft HSRMP will be finalised and become binding only once each party agrees in writing to the HSRMP.
- (e) If the parties do not agree the HSRMP within 25 Business Days of MPI supplying the HSRMP template to the Recipient, then MPI in its sole discretion may suspend or terminate all or part of the Activities and/or Funding under this Agreement, until the HSRMP is agreed in writing between the parties.
- (f) The Recipient must comply, and will ensure its Workers and other Personnel comply, with the HSRMP. The Recipient must also review and if necessary update the HSRMP at least every six (6) months, and must update the HSRMP whenever MPI (acting reasonably) advises the Recipient that a new or increased risk or hazard has arisen in performing the Actitivies.

# 4. Subcontractors and Personnel

- 4.1 Subcontractors
  - (a) The Recipient must ensure that:
    - each Subcontractor is fully aware of the Recipient's obligations under this Agreement to the extent necessary for the Subcontractor to properly perform its obligations;
    - each subcontract it enters into with a Subcontractor is consistent with this Agreement, to the extent relevant and material for the performance of the Subcontractor's obligations; and
    - (iii) each subcontract with a Subcontractor requires that further subcontracting is also consistent with this Agreement.
  - (b) During the term of this Agreement, if MPI requests in writing, the Recipient will:
    - conduct a health and safety pre-qualification process for all, or any particular, Subcontractors the Recipient (or the Recipient's Subcontractor) proposes to use to perform the Activites; and/or
    - submit a HSRMP, or amend an existing HSRMP, to address specific health and safety risks posed by the proposed use of the Subcontractor.
  - (c) The Recipient must ensure that, prior to the performance of the Activities by any Subcontractor, the Subcontractor has been advised of, and agrees to comply with;

National Wilding Conifer Control Programme

- (i) the Recipient's health and safety policy and procedures,
- the health and safety risk management plan for the relevant PCBU (as described in (j) of the Activities table in Part A);
- the plan to manage any specific site within the relevant Management Unit based on health and safety issues in performing the Activities; and
- (iv) any health and safety directions given by MPI to the Recipient under clause 3.1(a)(iii) of this Part B.
- (d) The Recipient will be responsible for the acts and omissions of each Subcontractor as if the act or omission was the Recipient's. The Recipient will not be relieved of any of its llabilities or obligations under this Agreement by entering into any subcontract.
- (e) The Recipient will ensure that:
  - its contract with each Subcontractor it engages in relation to the National Programme contains generally the same obligations; and
  - (ii) gives MPI the same rights as are set out in this Agreement and relate to the Subcontractor's performance of the Activities. MPI can directly enforce those obligations and exercise its rights against the Subcontractor pursuant to the Contract and Commercial Law Act 2017, provided that MPI will not:
    - (A) enforce its rights under 4.1(e)(ii) without seeking to resolve any issues in the first instance with the Recipient; or
    - (B) engage with Subcontractors without the prior written consent of the Recipient, such consent not to be unreasonably withheld.

# 4.2 Personnel

If MPI is at any time dissatisfied on reasonable grounds with the performance of the Recipient's Personnel in its performance of the Recipient's obligations under this Agreement:

- (a) MPI may notify the Recipient in writing of that dissatisfaction; and
- (b) the parties, in good faith, will discuss an appropriate course of action in relation to that Personnel.

If the Personnel performs a role that involves the accounting or allocation of Funding and has been arrested for or committed a dishonesty offence, the Recipient will remove that Personnel from that role immediately, if requested by MPI.

# Significant Events process

# 5.1 Notification of Significant Event

Each party will notify the other party as soon as reasonably practicable upon becoming aware of an actual or likely:

- Serious failure: significant failure (or series of related failures which together are significant) of the Recipient to perform its obligations in accordance with this Agreement;
- (b) MPI Breach: significant event, breach or likely breach by MPI of its obligations under this Agreement;
- (c) Reputational Event: any event relating to the Agreement which is likely to cause MPI, the government, the Recipient, its Personnel and/or Subcontractors significant adverse media or parliamentary attention, and in any event, if there is a change in the criminal record of the Recipient, and/or its Personnel;
- (d) Security Event: security event, breach or likely breach relating to the Agreement that compromises or is likely to compromise the integrity or security of MPI, the Recipient, the Activities, Funding, MPI assets (including any technology system) or MPI Data (such event, breach or likely breach being a Security Event); and
- (e) Regulatory Event: any event, or a near miss, relating to the Agreement that would cause the Recipient to notify a regulator, including any health and safety regulator, the office of the Privacy Commissioner or a financial services regulator,

National Wilding Conifer Control Programme

1.0

(each such event, circumstance, breach or failure being a Significant Event)

# 5.2 Rectification process

If a party advises the other of a Significant Event, and MPI requests, the parties will endeavour to agree a course of action. If no course of action is agreed within the timeframes advised by MPI (which, depending on the nature of the Significant Event, could be required within an hour), the Recipient must:

- (a) Co-operate: co-operate with MPI and any other third party specified by MPI;
- (b) Follow instructions: follow any reasonable instructions given by MPI, including any instruction given:
  - (i) under clause 6.7(b) of this Part B; and/or
  - (ii) to stop following a previous instruction (or any part of it),

and, if the Significant Event is related to a Notifiable Event, near miss, or breach, MPI's instruction may include:

- (iii) to stop performing the Activities (or any part of them);
- (Iv) to suspend any work being carried out under or in relation to this Agreement by the Recipient, and/or its Personnel until MPI (in its sole discretion) considers the health and safety risk has been eliminated or adequately mitigated; and/or
- (v) to amend the HSRMP in the manner MPI instructs.
- (c) Investigate: investigate the cause of the Significant Event and give a report to MPI (in writing if MPI requests) which describes (to the extent reasonably known):
  - (i) the cause of the Significant Event;
  - (ii) where the Significant Event is a Security Event and to the extent known, the identity of the person or persons who have gained or attempted to gain unauthorised access; and
  - (iii) the actions proposed to be taken by the Recipient.

If MPI requests, allow MPI to be involved in relation to that investigation to the level MPI requests.

- (d) Update: update MPI on regular basis of any subsequent developments relating to the Significant Event.
- (e) Remedy/Mitigate: remedy or mitigate the immediate effects of the Significant Event, or the circumstances or issue giving rise to the Significant Event, as soon as reasonably practicable, in the manner reasonably requested by MPI and provide MPI with a report (in writing, if MPI requests) of the action taken. MPI will take reasonable steps to remedy or mitigate the effects of the Significant Event on MPI.
- (f) Prevent Recurrence: take all reasonable action to prevent a recurrence of the relevant Significant Event and notify MPI (in writing if MPI requests) of any such action taken.
- (g) Remedy plan: within the period reasonably requested by MPI, prepare and submit to MPI for its approval a remedy plan setting out how the Recipient considers the Significant Event could be optimally remedied, the relevant timeframes for such remediation, what the Recipient considers would be required of the Recipient, MPI and/or any third party and any other details reasonably required in order for the Significant Event to be remedied in an optimal manner. If MPI approves that submitted plan, each party will comply with the requirements of that party as set out in that plan and any other requirements notified by MPI (acting reasonably).

# 5.3 Cost of remedying

MPI will pay the Recipient for the actual reasonable costs incurred by the Recipient in complying with clause 5.2 of this Part B but only:

 (a) up to the amount of the costs that was advised by the Recipient to MPI in advance of the Recipient incurring those costs; and

National Wilding Comfet Control Programme

(b) to the extent that the Significant Event is not due to the Recipient's or its Personnel's breach or misconduct.

# 5.4 Other remedies

The issuing of a notice under clause 5.1 of this Part B does not limit either party's other rights and remedies.

6. Payment

# 6.1 MPI's obligation to pay the Funding

MPI's obligation to pay the Funding is subject to:

- (a) MPI continuing to have sufficient funding within its budget for the Funded Activities;
- (b) there being no un-remedied non-trivial breach of this Agreement by the Recipient: and
- (c) MPI being satisfied on reasonable grounds that the Funding is being appropriately expended on the Funded Activities and is not being used in breach of this Agreement.

# 6.2 Total amount payable

The Funding is the total amount payable by MPI for the provision of the Funded Activities. All other costs, disbursements and other expenses incurred by the Recipient in relation to this Agreement are at the Recipient's cost and not to be paid for by the Funding.

# 6.3 Invoicing

- (a) The Recipient will invoice the Funding in accordance with this clause 6,3 of Part B, Each Tax Invoice submitted by the Recipient (whether submitted monthly or otherwise) will include sufficient details to enable MPI to identify:
  - the Agreement number (if any);
  - (ii) the period covered by the invoice;
  - (iii) the particular Funded Activities which are the subject of the invoice; and
  - (Iv) the amount of Funding payable.
- (b) Unless provided otherwise in this Agreement, the Recipient:
  - (i) will provide MPI with aTax Invoice for the aggregate amount of Funding for each Management Unit for the relevant Financial Year, as set out in the table at clause 4.2 of Part A (as those amounts may be amended in accordance with clause 4.2 of Part A); and
  - may provide that Tax Invoice annually in advance for the immediately following Financial Year.
- (c) Without limiting MPI's other rights under this Agreement, MPI will pay the Recipient's Tax Invoices by the within 20 Business Days of receiving that invoice or before the beginning of the Financial Year to which the invoice relates, whichever is the later date. All payments by MPI will be deemed to have been made in Wellington, New Zealand. Payment by MPI is not evidence that the Activities to which the invoice relates have been provided in accordance with this Agreement.

#### 6.4 Disputed invoices

If MPI has a bona fide dispute in relation to all or any portion of any Tax Invoice, MPI may withhold payment of the amount subject to the dispute, provided that:

- (a) MPI will pay the undisputed amount when it becomes due and payable; and
- (b) the Recipient will continue to perform its obligations under this Agreement while the dispute is resolved.

# 6.5 Eligible Expenditure

The Recipient must apply the Funding only to expenditure that:

- (a) is incurred by the Recipient;
- (b) for the purpose of the Funded Activities or for purposes incidental to the Funded Activities;

120

National Wilding Conlifer Control Programme

- (c) is incurred during the term of this Agreement; and
- (d) is otherwise approved in writing by MPI (in its absolute discretion) as eligible expenditure for the purposes of this Agreement,

(Eligible Expenditure).

# 6.6 Financial Management

The Recipient must:

- ensure that any payments of Funding made to a third party in connection with this Agreement (including to its Personnel) are correctly made and properly authorised;
- (b) maintain proper and diligent control over the incurring of all liabilities in relation to the Funding;
- (c) maintain an appropriate financial management system to ensure that the Funding is separately identified and managed within its accounts; and
- (d) except with MPI's prior written approval, not use the Funding for the purposes of a guarantee or security for any loan, credit, payment or other interest, or in the context of any litigation.

# 6.7 Funding Freeze

- (a) Without limiting any other right or remedy of MPI, if the Recipient materially breaches this Agreement, then MPI may issue the Recipient with a notice directing the Recipient not to spend the Funding (Funding Freeze Notice).
- (b) If the Recipient receives a Funding Freeze Notice under this clause 6.7 of this Part B, then, until the breach is remedied to MPI's satisfaction, the Recipient must not spend any Funding:
  - (i) unless and until MPI notifies the Recipient otherwise; and
  - except to the extent required to pay for any bona fide and unavoidable pre-existing contractual commitments to spend the Funding in accordance with this Agreement.

# 6.8 Repayment

- (a) Without limiting any other right or remedy of MPI, MPI may recover Funding from the Recipient in the following circumstances (and the Recipient will pay that Funding within 5 Business Days of MPI's written notice requiring payment).
  - (i) Misspent Funding: At any time MPI may recover the amount of any Funding that has been spent or used other than in accordance with this Agreement, together with interest on all such amounts calculated at 10% per annum from the date of the misspending to the date the money is repaid.
  - (ii) Uncommitted Funding: On expiry or termination of this Agreement, MPI may recover any Funding which the Recipient:
    - has not spent or contractually committed to spend in accordance with this Agreement; and/or
    - (B) has spent or contractually committed to spend in accordance with this Agreement but which the Recipient can have refunded or released from that commitment, and the Recipient must use all reasonable endeavours to obtain such refund or release.
  - (iii) Activities Abandoned: MPI may recover an amount up to the total value of the Funding less the amount that the Recipient spent in accordance with this Agreement if the Recipient;
    - states an intention not to perform the Activities in accordance with this Agreement; and/or
    - (B) within 10 Business Days of being requested to do so by MPI, does not demonstrate (to MPI's reasonable satisfaction) that the Recipient will proceed with performing the Activities in accordance with this Agreement.

National Wilding Conifer Control Programme

- (iv) Excess Funding: Where the total Funding paid under this Agreement and any other money received by the Recipient to carry out the Funded Activities exceeds or is likely to exceed the funding required to perform the Activities, or the Recipient has excess Funding at the termination or expiry of this Agreement, the Recipient must notify MPI upon becoming aware and, if MPI requests, promptly refund to MPI the excess amount.
- (b) If the Recipient does not pay, as required by this clause, the Recipient must pay MPI interest calculated at 10% per annum from the date payment is due until the date the money is repaid.

## 6.9 Insufficient Funding

If the Recipient uses (or reasonably expects to use) all of the Funding in accordance with this Agreement without completing its performance of the Funded Activities during the term of this Agreement, then:

- (a) the Recipient will give MPI written notice as soon as reasonably practicable (but the Recipient does not need to give more than six (6) months advance notice), detailing the amount of additional funding requested by the Recipient (Additional Funding) and supporting evidence as to why the Additional Funding will be sufficient for the Recipient to complete the performance of the Funded Activities; and
- (b) within a reasonable period of time after receiving the Recipient's request for Additional Funding under clause 6.9(a)above, MPI will give the Recipient written notice as to whether it will provide:
  - the Additional Funding in full (in which case the Recipient will be obliged to complete the performance of the Funded Activities in accordance with this Agreement); or
  - (ii) part of the Additional Funding (in which case the parties will follow the change control procedure set out at clause 7 to agree upon a reduction in the scope of Funded Activities to be performed by the Recipient); or
  - (IIi) no Additional Funding (in which case the Recipient will not be obliged to continue to perform the Activities under this Agreement after all of the funding has been spent on the Funded Activities in accordance with this Agreement).

# 6.10 Tax

All amounts set out in this Agreement are stated before the addition of GST, which, if payable, will be paid at the same time as the supply to which the GST relates. Subject to the previous sentence, MPI will not be obliged to pay any of the Recipient's and/or its Personnel:

- (a) taxes, duties, excess, levies, carbon credits;
- (b) holiday pay, sick pay or other payment under the Holidays Act 2003;
- (c) redundancy or any other form of severance pay; and/or
- (d) other amounts in relation to the performance of its obligations under this Agreement that are not the performance of the Funded Activities.

MPI will not be required to gross up any payment or Funding to the Recipient where MPI deducts tax due to non-resident contractors' tax.

# Change control procedure

# 7.1 Change request

A party may request an amendment to the terms of this Agreement at any time by providing a written change request to the other. Unless the parties' respective Contract Managers agree in writing to use a truncated process, each party agrees to follow the process set out in this clause 7 when requesting a change to the terms of this Agreement.

#### 7.2 Understanding consequence of change request

(a) Where MPI submits a change request, it will provide a reasonable description and explanation of the change sought.

National Wilding Conifer Control Programme

- (b) Where the Recipient submits a change request, or within a reasonable period of time of receiving a change request from MPI, the Recipient must provide MPI with a reasonable description of:
  - how the proposed change would be implemented, including details of how the requested change would affect the Activities and any other term of this Agreement;
  - the cost of implementing the change and the Recipient's calculation of changes to the Funding (if any), which must be reasonable and calculated in good faith;
  - (iii) any information reasonably requested by MPI; and
  - (iv) any other relevant information, which may include how any change request can be most optimally implemented.
- No change request will be binding unless each party accepts the change request in writing.

# 8. Information management

# 8.1 Provision of information

Each party will:

- (a) provide the other party with all information relating to the Activities as reasonably requested from time to time, within the timeframe reasonably requested; and
- (b) as soon as reasonably practicable, provide the information requested by the other party to comply with its statutory, parliamentary or other public sector reporting obligations; and
- (c) ensure that all information provided in writing to the other party under or in connection with this Agreement is materially accurate, complete and not misleading at the time the information was provided, for the purpose for which it was provided.

# 8.2 Privacy

- (a) Each party must act in accordance with the Privacy Act 1993.
- (b) Each party will not use any Personal Information acquired in the course of performing its obligations under this Agreement, or disclose that Personal Information to any person other than MPI or the Recipient (as applicable) or the individual to whom the information relates, except:
  - (i) to the extent reasonably required to perform its obligations under this Agreement;
  - (ii) in accordance with the Privacy Laws; or
  - (iii) with the other party's or person's written consent.

# 8.3 Recordkeeping

The Recipient will (and will ensure that each of its Subcontractors under and in relation this Agreement):

- (a) keep full, accurate and up-to-date records, including:
  - financial records relating to the performance of the Activities and all monies paid and payable by MPI under or in relation to this Agreement sufficient to enable MPI to meet its obligations under the Public Finance Act 1989; and
  - (ii) any health and safety event which has been investigated internally by the Recipient, or a health and safety regulator, relating to the Activities (and the outcome of such investigations); and
  - (iii) any health and safety due diligence processes, compliance records, and any associated documents, relating to the health and safety compliance and performance of Workers and other Personnel involved in the Activities;
- (b) retain such records for 7 years from the expiry or termination of this Agreement; and
- (c) on request by MPI, and at no cost to MPI, provide to MPI copies of any such records reasonably requested by MPI and any information or materials (including financial information relating to the provision of Activities).

Nelional Wilding Confer Control Programme

# 9. Confidentiality

# 9.1 Security and non-disclosure

Subject to clauses 9.2 and 9.3 of this Part B, each party agrees that it will:

- use all reasonable endeavours to keep the Confidential Information of the other party confidential at all times;
- (b) not Use, communicate, make available or re-supply the other party's Confidential Information to any person, other than for the purposes of performing its obligations and/or exercising its rights in accordance with this Agreement;
- (c) effect and maintain all reasonable security measures to safeguard the other party's Confidential Information from access, damage or use by unauthorised persons;
- (d) store all Confidential Information in New Zealand; and
- (e) ensure that any Personnel or professional adviser to whom a party discloses the other party's Confidential Information is aware of, and complies with, the provisions of this clause 9 of this Part B.

## 9.2 Permitted disclosure

A party will not be bound by the obligations of confidentiality in clause 9.1 of this Part B to the extent that any disclosure of Confidential Information:

- (a) is required by law, convention, parliamentary rules or the rules of any stock exchange (Requirement), in which case, if permitted by the Requirement and as soon as reasonably practicable, the disclosing party will notify the other party of the Requirement, the information it will disclose and the date on which it will disclose the information, and will comply with that notice;
- (b) Is required for the purposes of performing and enforcing this Agreement;
- (c) is of information that is publicly available through no fault of that party;
- is made with the other party's prior written approval of the disclosure or (in the case of clause 9.1(d)) the transfer outside New Zealand;
- which was rightfully received from a third party without restriction or without breach of this Agreement; or
- (f) was developed independently of the Confidential Information.

# 9.3 Information sharing

In addition to clause 9.2 of this Part B, and in recognition of the mutual benefit of information sharing for the National Programme as a whole, the obligations of confidentiality in clause 9.1 of this Part B do not apply to the extent that any disclosure of Confidential Information is for the purpose of consulting, cooperating or coordinating with other then current recipients of funding from MPI under the National Programme, including in relation to:

- (a) current best practice in relation to the performance of the Activities; and/or
- (b) a party's health and safety management under this Agreement.

# 9.4 Mandatory disclosures

If any party is required to disclose any Confidential Information (as contemplated by clauses 8.1(a) or 9.2 of this Part B), that party will use its reasonable endeavours to obtain the recipient's commitment that it will treat the confidential information confidentially.

#### 9.5 Publicity and announcements

- (a) The Recipient will acknowledge MPI as a source of funding in all publications and publicity regarding the Activities, and will obtain MPI's approval of the form of the acknowledgement.
- (b) MPI may publicise and report on its distribution of Funding to the Recipient on any website, in any media releases, general announcements or annual reports. MPI may disclose the name of the Recipient, any of its Subcontractors, the amount of the Funding and a brief description of Activities.

National Wilding Conifer Control Programme

- (c) Neither party may post on websites, social networking sites or publicly display objectionable or derogatory comments about the Activities, this Agreement, each other, or any of their Personnel.
- (d) The Recipient will promptly refer any enquiries from the media or any other person about the terms or performance of this Agreement to the MPI Programme Manager. If the MPI Programme Manager cannot be contacted, the Recipient will instead contact the person holding the office of "MPI National Procurement and Contracts Manager".
- 10. Warranties

# 10.1 Warranties

The Recipient warrants that at the date of this Agreement:

- (a) it is not insolvent or bankrupt and no action has been taken to initiate any form of insolvency or administration in relation to the Recipient;
- there are no circumstances or matters that are likely to have a material adverse effect on the performance of its obligations under this Agreement;
- (c) it is not aware of any information that has not been disclosed to MPI which may, if disclosed, materially adversely affect the decision of MPI to provide the Funding;
- (d) all information relating to the Activities that was provided by the Recipient to MPI prior to MPI's execution of this Agreement, including in any proposal or presentation by the Recipient, was accurate, complete and true at the time it was provided. The Recipient acknowledges that MPI is entering into this Agreement in reliance on such information; and

on a continuing basis during the term of this Agreement (to the best of the Recipient's knowledge and belief):

- (e) the performance of the Activities under this Agreement will not constitute or give rise to a breach of:
  - the rights of any person (including any Intellectual Property Rights or physical property rights); or
  - (ii) any law with which the Recipient or Personnel is required to comply.
- 11. Intellectual Property and data

# 11.1 Background Intellectual Property

Except to the extent specified in this Agreement, nothing in this Agreement confers on a party any right, title or interest in the other party's Background Materials.

# 11.2 New Intellectual Property

Unless otherwise agreed between the parties, all Intellectual Property Rights in the New Materials arising under or in relation to this Agreement are the property of MPI from the time of creation or will vest in MPI immediately afterwards.

# 11.3 Licence

Subject to clause 11.4, each party (Licensor) grants the other (Licensee) a perpetual, irrevocable, world-wide, royalty-free, non-exclusive licence to Use any of the Licensor's:

- (a) Background Materials;
- (b) New Materials;
- (c) Third Party Materials (provided that the Licensor holds the rights to grant this licence in relation to Third Party Materials); and/or
- (d) any of the Licensor's data,

provided or made available by the Licensor to the Licensee under or in connection with this Agreement. The Licensee may sub-license its rights under this clause 11.3 to its Subcontractors, but only to the extent necessary for that Subcontractor to perform the Activities or provide the Outputs in accordance with this Agreement.

Nelional Wilding Coniter Control Programme

# 11.4 WCIS

The Recipient must not (and will ensure its Personnel does not) copy, reverse engineer, rent, decompile or attempt to circumvent any security measures on WCIS or any part of it.

#### 11.5 Know How

Notwithstanding any other provision of this Agreement, MPI and the Recipient will not be prevented or restricted by this Agreement from using any technique, idea, concept or know-how relating to this Agreement.

# 12. Third party rights

# 12.1 Third Party Claim

If a third party claims that a party (the Infringing Party) has infringed that third party's Intellectual Property Rights, other rights of that third party or the law (Third Party Claim), then:

- the other party will allow the Infringing Party to control settlement or defence of the Third Party Claim. The Infringing Party will consult with the other party on the settlement or defence of the Third Party Claim;
- (b) at the Infringing Party's cost and reasonable request, the other party will co-operate with the Infringing Party in the defence and any related settlement negotiations concerning any Third Party Claim; and
- (c) the other party will not enter into any settlement, negotiation or compromise with the third party without the Infringing Party's written consent (not to be unreasonably withheld).

# Conflicts of interest

# 13.1 Avoiding conflicts

Each party must maintain a high standard of honesty and integrity at all times in the performance of this Agreement and must use all reasonable endeavours to avoid any actual or perceived conflict of interest, being:

- (a) a situation which might create an actual or perceived conflict of interest, or compromise the Recipient's or MPI's integrity; and/or
- (b) any financial or other interest or undertaking that might interfere with or compromise the performance of the other party's obligations under this Agreement, or the integrity or standing of either party.

# 13.2 Notifying conflicts

Each party must advise the other party in writing of any actual or perceived conflict of interest that might arise in the performance of its obligations under, or in connection with, this Agreement. Each party will assist the other party to manage, avoid or mitigate that conflict and follow any reasonable instructions (which may include stopping the performance of certain work) that the parties consider necessary or appropriate to manage the conflict of interest.

#### 14. Audit

- 14.1 Audit
  - (a) As part of the public sector, each party acknowledges the need to ensure transparency and accountability in the use of public funds, and also the cost and resource involved in carrying out an audit. Accordingly, MPI will work with the Recipient to identify an appropriate audit frequency and scope but, as the funder, MPI must have default rights of audit that it can use at any stage.
  - (b) The Recipient must allow MPI and any auditor nominated by MPI to conduct an audit from time to time, for the purpose of:
    - determining the level of compliance with this Agreement (including whether there
      has been a breach of this Agreement) by the Recipient and/or its Personnel;
    - determining whether Activities invoiced for or by the Recipient have been performed and invoiced according to this Agreement; and/or
    - (III) assisting in resolving a matter in dispute between the parties.

National Wilding Conifer Control Programme

- (c) The Recipient will (and will ensure that each of its Personnel will) provide reasonable access and assistance to MPI (or an auditor appointed by MPI) in relation to an audit conducted under this clause 14.1 of this Part B.
- (d) In conducting an audit, MPI must:
  - give the Recipient reasonable notice of the audit, which in any event does not need to be more than 5 Business Days; and
  - (ii) comply with the Recipient's reasonable security requirements,
- (e) The Recipient may use the Funding for the reasonable costs of any audit unless a material non-compliance or charging error with this Agreement is discovered through the audit, in which case, the Recipient must reimburse MPI for the amount of any Funding that has been used to pay for the costs of the audit.
- (f) MPI will promptly notify the Recipient of the draft and final results of any audit conducted under this clause 14.1 of this Part B.
- 15. Dispute resolution

# 15.1 Negotiation

Both parties will endeavour to resolve any dispute that may arise under or in connection with this Agreement through negotiation.

#### 15.2 Escalation

If the parties are unable to resolve a dispute by negotiation within 10 Business Days after the dispute was first notified in writing by one party to the other (**Dispute Notice**), and either party gives written notice to the other, then:

- the dispute will be referred for resolution by their respective senior responsible officers with oversight of this Agreement; and
- (b) if the dispute cannot be settled by the senior responsible officers within 20 Business Days of delivery of the Dispute Notice, the dispute will be referred for resolution by their respective Chief Executives.

#### 15.3 Mediation

Subject to clause 15.4 of this Part B, if a dispute cannot be settled within 30 Business Days of delivery of the Dispute Notice under clause 15.2 of this Part B (or such other timeframe agreed in writing by the parties), the dispute may be referred by either party to non-binding mediation in which case:

- (a) the parties agree to the mediation, which will be held in Wellington, New Zealand;
- (b) the mediation will be mediated by a sole mediator agreed in writing by the parties; or
- (c) if the parties cannot agree on a mediator within 10 Business Days of either party referring the dispute to mediation, either party may request the President of the New Zealand Law Society to appoint a suitably qualified independent mediator.

# 15.4 Right to seek relief

This clause 15 of this Part B does not affect either party's right to seek urgent interlocutory and/or injunctive relief from any New Zealand Court of competent jurisdiction.

#### 16. Liability

#### 16.1 Maximum liability

Without limiting clauses 16.2 and 16.3 of this Part B, the maximum aggregate liability, whether in contract, tort (including for negligence), breach of statutory duty or otherwise, of one party to the other party under or in connection with this Agreement is limited to the total amount of Funding payable by MPI under this Agreement had the Funded Activities been carried out in accordance with this Agreement.

National Wilding Conlier Control Programme

# 16.2 Exclusions

- (a) Neither party is liable for any indirect or consequential loss or damage arising under or in connection with this Agreement, even if foreseeable and even if advised of the possibility of such loss or damages.
- (b) The Recipient will not be liable under this Agreement to the extent the Recipient's breach of this Agreement is a direct and unavoidable result of MPI's express written instruction (given knowingly).
- (c) Notwithstanding any other clause in this Agreement, this clause 16 applies only to the extent permitted by law.

# 16.3 Uncapped liability

Clauses 16.1 and 16.2 of this Part B will not limit either party's liability under or in connection with this Agreement:

- (a) for damage to the other's or a third party's physical property;
- (b) for personal injury or death;
- (c) for fraud or wilful default; or
- (d) an obligation to pay any amount due and owing under this Agreement.

## 16.4 Mitigation

Each party must take reasonable steps to mitigate any loss or damage, cost or expense it may suffer or incur arising out of anything done or not done by the other party under or in connection with this Agreement.

# 17. Force Majeure

- (a) Where either MPI or the Recipient (the Affected Party) is unable to carry out its obligations in accordance with this Agreement due to a Force Majeure Event, and:
  - the Affected Party gives the other party immediate written notice of the nature, the expected duration of, and the obligation affected by, the Force Majeure Event;
  - the Affected Party gives regular updates of the expected duration of and effects of the Force Majeure Event; and
  - (iii) the Affected Party uses all reasonable endeavours to:
    - (A) mitigate the effects of the Force Majeure Event on the Affected Party's obligations under this Agreement, and
    - (B) perform the Affected Party's obligations under this Agreement despite the Force Majeure Event,

that obligation is suspended for so long as, and to the extent that, the Affected Party is affected by the Force Majeure Event.

(b) If the Recipient's obligation is relieved under this clause 17 of Part B, MPI will not be obliged to provide Funding for the Activities that have not been provided in accordance with this Agreement due to the Force Majeure Event.

#### 18. Insurance

For the term of this Agreement, and for at least three years after the term of this Agreement, the Recipient must effect and maintain adequate insurance to cover standard commercial risks, including professional indemnity, public liability and property damage (including fire) with sufficient coverage to cover its obligations under this Agreement. The Recipient must, upon the request of MPI, provide MPI with evidence of its compliance with this clause 18 of Part B.

# 19. Term and termination

19.1 Term

This Agreement commences on the Start Date and continues until the End Date unless terminated earlier in accordance with its terms including under clause 6.2 of Part A.

National Wilding Conifer Control Programme

#### 19.2 Termination for default and insolvency

Either party may immediately terminate this Agreement, by written notice to the other party, if the other party breaches any material provision of this Agreement and the breach is not remedied within 20 Business Days of the receipt of written notice from the first party requiring the other to remedy the breach.

# 19.3 Effects of expiry or termination

- (a) Termination of this Agreement is without prejudice to other rights and obligations of the parties accrued up to and including the date of termination.
- (b) On expiry or termination of this Agreement:
  - MPI may require the Recipient to provide evidence of how the Funding has been spent; and
  - any Funding that has not yet been distributed by MPI will not be distributed to the Recipient

# 19.4 Survival clauses

Clauses 6.8, 8.2, 8.3, 9, 10.1(e), 11. 12, 14, 15, 16, 17, 18, 19, 20 and 21 of Part B any other clauses that expressly or by necessary implication survive this Agreement, continue after expiry or termination of this Agreement or part of it.

# 20. Notices

- (a) Any notice or other communication under this Agreement will be deemed to be validly given if in writing and delivered by hand, registered mail, national post or international post, or email (subject to the remainder of this clause 20) to the Manager Recovery and Pest Management (for MPI) or Fund Manager (for the Recipient).
- (b) Any notice will be deemed to have been given on the date when actually delivered personally, by registered mail or following international posting, on the third Business Day following posting to a national address, and on the date that receipt of an emailed notice is acknowledged by the recipient personally (that is, not by any automatically generated email). However, if delivery occurs after 5:00 pm on a Business Day, it will be deemed delivered at 9.00 am on the next Business Day.
- (c) The parties agree that no notice required or permitted to be given under clause 15 (Dispute resolution) or clause 19 (Term and termination) may be given by email alone.

# 21. General

- (a) Subject to clause 21(b) of this Part B, neither party will assign or transfer this Agreement or any rights or obligations under this Agreement without the prior written consent of the other party (such consent not to be unreasonably withheld).
- (b) MPI may assign, transfer or subcontract this Agreement or any rights or obligations under this Agreement to another public sector agency who has appropriation for the National Programme without the prior written consent of the Recipient.
- (c) No third party is entitled to the benefit of, or to enforce, this Agreement.
- (d) Each party will bear its own legal and other costs relating to the preparation and execution of this Agreement, unless agreed otherwise in writing.
- (e) This Agreement constitutes the entire agreement between the parties in connection with its subject matter and supersedes all previous agreements or understandings between the parties in connection with its subject matter.
- (f) The provisions of this Agreement are severable. If any one provision is determined to be judicially unenforceable, the remaining provisions will continue to be binding and enforceable.
- (g) This Agreement may be executed in any number of counterparts (including by executed, scanned or emailed copies) each of which will be deemed an original. Once each party has received a copy of the executed counterpart from the other party, each counterpart will be deemed to be as valid and binding on the party executing it as if it had been executed by all the parties.

National Wilding Conifet Control Programma

(b)

This Agreement will be governed by, and construed in accordance with, the laws of New Zealand and the parties submit to the non-exclusive jurisdiction of the courts of New Zealand in relation to this Agreement.

National Weding Confige Control Programme

Council Meeting 2021.04.14

# Execution

**EXECUTED** as an Agreement

Signed by The Sovereign in right of New Zealand acting by and through the Ministry of Primary Industries by:

Signature

John Walsh

Name

Acting Deputy Director General Biosecurity NZ Title

9 September 2020

Date

Signed for and on behalf of Otago Regional Council by:

Signature Name H ELIHIVE Title 2020 6 U Date

National Wilding Conifer Control Programme

# Schedule 1 – Definitions and Interpretation

# 1. Defined terms

# 1.1 Defined terms

In this document, unless the context requires otherwise, the following terms have the following meanings:

Activities means the activities performed or to be performed (and any outputs provided or to be provided) by or on behalf of Recipient under this Agreement, including the activities and outputs described in clause 3 of Part A, and any other activities that are necessarily or customarily provided as part of those activities and/or outputs.

Agreement means this funding agreement, including Part A, Part B and this Schedule 1.

Background Material means any property, information, documentation or other material in any form:

- (a) in existence prior to the Start Date; or
- (b) developed outside the scope of this Agreement.

Business Day means any day, excluding Saturdays, Sundays and statutory public holidays in Wellington or Region of the Recipient.

Start Date means the date this Agreement commences, as set out in clause 6.1 of Part A.

Confidential information means

- (a) in the case of MPI, MPI Data; and
- (b) in the case of both MPI and the Recipient, any other information obtained from the other party in the course of, or in connection with this Agreement that is marked confidential or ought to reasonably be considered to be confidential, regardless of value or importance.

Control Measures has meaning set out in regulation 3 of the HSWA (GR&WM) Regulations 2016.

Contract Manager means the person identified in clause 7 of Part A as a contract manager for the relevant party, as that person may be replaced from time to time in accordance with this Agreement.

End Date means the date this Agreement ends as set out in clause 6.1 of Part A, as may be extended in accordance with clause 6.1 of Part A.

Financial Year means the twelve months from 1 July to 30 June.

Force Majeure Event means, in relation to a party, an event or circumstance which is beyond the reasonable control of that party, but does not include any event or circumstance which could have been avoided, prevented or circumvented by that party taking reasonable steps, including ensuring business continuity by implementing an appropriate disaster recovery plan.

Fund Manager means the Recipient's fund manager, as notified by the Recipient to MPI and may be replaced from time to time.

Funded Activity has the meaning set out in clause 3.1(a) of Part A.

Funding means the amounts paid or payable to the Recipient by MPI under this Agreement.

Governance Group means the group which governs the National Programme, whose members are MPI (Chair), Land Information New Zealand (LINZ), New Zealand Defence Force (NZDF), Department of Conservation (DOC), and Local Government New Zealand (LGNZ).

National Wilding Coniler Control Programme

GST means goods and services tax chargeable under the GST Act.

GST Act means the Goods and Services Tax Act 1985.

Health and Safety Incident means any event that falls under clause 3.1(k)(i) - (iv) of Part A.

Health and Safety Risk Management Plan or HSRMP means a plan agreed in writing between the Recipient and MPI for the management of risks and hazards relating to the Recipient's performance of the Activities in the relevant Management Unit.

HSWA means the Health and Safety at Work Act 2015.

HSWA Legislation means the Health and Safety at Work Act 2015 and includes all regulations made under that Act (including but not limited to the HSWA (GR&WM) (General Risk Regulations), and any other health and safety-related legislation relevant to the Recipient's performance of the Activities and its other obligations under this Agreement.

HSWA (GR&WM) Regulations 2016 means the Health and Safety at Work (General Risk and Workplace Management) Regulations 2016.

Incident means an unplanned event that results in injury but does not require time off work.

In Kind Contributions means funds, labour, or materials donated by other stakeholders or interested groups.

Intellectual Property Rights includes copyright, and all rights conferred under statute, common law or equity in relation to inventions (including patents), registered and unregistered trade marks, registered and unregistered designs, circuit layouts, Confidential Information, know-how, trade secrets, computer program codes and all other proprietary rights, whether registered or unregistered, and all equivalent rights and forms of protection anywhere in the world, together with all rights, interests or licences in or to any of the foregoing.

Management Unit or MU means an area designated by MPI as a management unit as set out in WCIS, and has a fixed name, number and boundary (area).

Management Unit Operational Plan or MUOP means the management unit, operational plan for a Management Unit, which is prepared by or on behalf of the Recipient and endorsed in writing by the relevant RCG.

MPI Coordinator means the MPI coordinator, as notified by MPI to the Recipient and may be replaced from time to time.

#### MPI Data means:

- (a) any data owned, held, used or created by MPI, and provided to the Recipient;
- (b) any data processed by the Recipient as a direct result of this Agreement; and
- (c) all records, data and other information (other than software object code) provided or made available to the Recipient by MPI.

MPI Programme Manager means the MPI programme manager, as that person may be replaced from time to time in accordance with this Agreement.

MPI's Operations Lead means the MPI operations lead, as that person may be replaced from time to time in accordance with this Agreement.

MU Manager means the Recipient's MU manager, as notified by MPI to the Recipient and may be replaced from time to time.

National Programme means the national wilding conifer programme, as described in Recital A of the Background.

Near Miss means an unplanned event that did not result in injury, illness or damage but had the potential to do so.

New Material means any property, information, documentation or other material in any form created:

- (a) by, for or on behalf of the Recipient;
- (b) on or following the Start Date; and
- (c) for the purpose of, or as a result of, performing its obligations under this Agreement.

National Wilding Conifer Control Programme

.33

Notifiable Event means a notifiable event within the meaning of section 25 of the HSWA, Including but not limited to the following situations as a result of work:

- (a) a death;
- (b) a notifiable illness or injury; or
- (c) a Notifiable Incident.

Notifiable Incident means a notifiable incident within the meaning of section 24 of the HWSA.

Operational Area or OA means part of the Management Unit designated as an operational area by MPI, as set out in WCIS, and has a fixed name, number and boundary (area).

Outcomes has the meaning in clause 1(a) of Part A.

Output means all or any output of the Activities provided or to be provided by or on behalf of the Recipient under this Agreement, including the outputs described in clause 3 of Part A (as amended from time to time in accordance with this Agreement).

Part A means Part A of this Agreement.

Part B means Part B of this Agreement.

PCBU means a person conducting a business or undertaking within the meaning of section 17 of the HWSA.

Personal Information has the meaning given to that term in section 2 of the Privacy Act 1993.

Personnel of any person, means all individuals directly or indirectly engaged by that person, including but not limited to directors, employees, contract staff, agents, consultants, specialists, support staff and co-opted or seconded staff and, in the case of the Recipient, includes its Subcontractors.

Privacy Laws means the Privacy Act 1993 (or any future replacement statue) and any other applicable legislation, principles, industry codes and policies.

Region means the region designated as a region by MPI, as set out in WCIS, and has a fixed name, number and boundary (area).

Regional Coordination Group or RCG means the regional coordination group for the relevant Region, as described in row (b) of the table at clause 3.1 of Part A.

Serious Incident means any Health and Safety Incident that:

- (a) has the potential for causing a fatality, or long-term injury or illness; or
- (b) does not meet the 'Notifiable Event' threshold, but still results in a lost time injury or illness.

Significant Event has the meaning given to that term in clause 5 of Part B.

Start Date means the start date set out in clause 6.1 of Part A.

Subcontractor means:

- any person engaged by the Recipient to perform all or any part of the Activities or in respect of this Agreement; and
- (b) any subcontractor of a person described in (a) of this definition.
- Tax Invoice means a tax invoice as defined in the Goods and Services Tax Act 1985.

Technical Liaison means the person identified in clause 7 of Part A as the technical liaison for the relevant party, as that person may be replaced from time to time in accordance with this Agreement.

Third Party Material means any property, information, documentation or other material in any form owned by a third party that is:

- (a) included, embodied in or attached to the New Material; or
- (b) used in undertaking the Activities.

National Wilding Conifer Control Programme

Use means to use, copy, store, process, transfer, view, handle, modify, support and/or maintain. WCIS means the 'wilding conifer information system' provided by MPI from time to time.

Worker means an individual who carries out work for a PCBU within the meaning of section 19 of the HWSA.

# 2. Interpretation

In this Agreement, unless the context requires otherwise:

- the following order of precedence (in order of decreasing priority) applies to the documents forming part of, or incorporated into, this Agreement if any conflict or inconsistency arises;
  - (i) Part B;
  - (ii) Part A; and
  - (iii) the Schedules;
- (b) headings are for guidance only and do not affect interpretation;
- (c) the singular includes the plural and vice versa;
- (d) where a word or phrase is defined, its other grammatical forms have a corresponding meaning;
- (e) a reference to:
  - (i) a party to this Agreement includes that party's permitted assigns and successors;
  - (ii) \$ or dollars is a reference to New Zealand currency;
  - (iii) monetary amounts are the amounts stated before the addition of GST (if any),
  - (iv) including and similar words do not imply any limitation;
  - (v) a person includes:
    - a company, body of persons (corporate or unincorporated) or any state, regional or local government body or agency; and
    - (B) that person's representatives, successors and assigns;
  - a statute includes references to regulations, orders or notices made under or in connection with the statute or regulations, as it may be amended or replaced from time to time;
  - a document includes references to that document as it may be amended or replaced from time to time;
  - this Agreement includes all attachments to this Agreement and any document incorporated into this Agreement by reference;
  - (ix) a clause is a reference to a clause in this Agreement;
  - (x) a Part or Schedule, are references to a part or schedule of this Agreement; and
  - (xi) a time means New Zealand standard time, as adjusted for daylights saving;
- subject to clause 20 of Part B, anything that this Agreement requires to be done in writing, may be done by email; and
- (g) no rule of construction applies to the disadvantage of MPI on the basis that MPI put forward this Agreement or any part of it.

National Wilding Conifer Control Programme

Ministry for Primary Industries Manatū Ahu Matua



# **Funding Agreement**

in relation to Wallaby Management Programme

The Sovereign in right of New Zealand acting by and through the Ministry of Primary Industries (**MPI**) and

Otago Regional Council (Recipient)

Council Meeting 2021.04.14

# **Funding Agreement**

# Wallaby Management Programme

| Deta | ails   | 5  |
|------|--|----|
| Back | kground  | 5  |
| Agr  | eed terms  | 7  |
| PAR  | TA: OPERATIONAL AND COMMERCIAL DETAILS                       | 7  |
| 1    | Outcomes   | 7  |
| 2    | Acknowledgement of appropriation                             | 7  |
| 3    | Activities   | 7  |
| 3.1  | In scope   | 7  |
| 32   | Restricted and/or Prohibited Activities                      | 15 |
| 4.   | Funding  | 15 |
| 4.1  | Purpose  | 15 |
| 4.2  | Maximum amount   | 15 |
| 4.3  | Method of payment  | 15 |
| 4.4  | Other provisions   | 15 |
| 5.   | Governance   | 15 |
| 5.1  | Role and responsibilities                                    | 15 |
| 6.   | Term and termination   | 16 |
| 6.1  | Term   | 16 |
| 6.2  | Termination of this Agreement                                | 16 |
| 7.   | Contact details  | 16 |
| PAR  | T B: GENERAL TERMS AND CONDITIONS                            | 18 |
| 1.   | Activities   | 18 |
| 2.   | Obligations  | 18 |
| 2.1  | Mutual obligations of the parties                            | 18 |
| 2.2  | Recipient's general obligations                              | 18 |
| 2.3  | MPI's obligations  | 18 |
| 3,   | Health and safety  | 19 |
| 3.1  | Compliance with health and safety legislation and directions | 19 |
| 4.   | Subcontractors and Personnel                                 | 19 |
| 4.1  | Subcontractors   | 19 |
| 4.2  | Personnel  | 20 |
| 5.   | Significant Events process                                   | 20 |
| 5.1  | Notification of Significant Event                            | 20 |
| 5.2  | Rectification process  | 20 |
| 5.3  | Cost of remedying  | 21 |
| 5.4  | Other remedies   | 21 |
| 6.   | Payment  | 21 |
| 6.1  | MPI's obligation to pay the Funding                          | 21 |
| 6.2  | Total amount payable   | 21 |
| 6.3  | Invoicing  | 22 |
| 6.4  | Disputed involces  | 22 |
| 0.5  | Eligiple Experionale   | 22 |
| 6.7  | Financial Management   | 22 |

Page 2

| 6.8  | Repayment                                   | 23 |
|------|---|----|
| 6.9  | Insufficient Funding                        | 23 |
| 6.10 | Tax   | 24 |
| 7.   | Change control procedure                    | 24 |
| 7.1  | Change request                              | 24 |
| 7.2  | Understanding consequence of change request | 24 |
| 8.   | Information management                      | 25 |
| 8.1  | Provision of information                    | 25 |
| 8.2  | Privacy                                     | 25 |
| 8.3  | Recordkeeping                               | 25 |
| 9.   | Confidentiality                             | 25 |
| 9.1  | Security and non-disclosure                 | 25 |
| 9.2  | Permitted disclosure                        | 26 |
| 9.3  | Information sharing                         | 26 |
| 9.4  | Mandatory disclosures                       | 26 |
| 9.5  | Publicity and announcements                 | 26 |
| 10.  | Warranties                                  | 27 |
| 10.1 | Warranties                                  | 27 |
| 11.  | Intellectual Property and data              | 27 |
| 11.1 | Background Intellectual Property            | 27 |
| 11.2 | New Intellectual Property                   | 27 |
| 11.3 | Licence                                     | 27 |
| 11.4 | Wall-IS                                     | 27 |
| 11.5 | Know How                                    | 28 |
| 12.  | Third party rights                          | 28 |
| 12,1 | Third Party Claim                           | 28 |
| 13.  | Conflicts of interest                       | 28 |
| 13.1 | Avoiding conflicts                          | 28 |
| 13.2 | Notifying conflicts                         | 28 |
| 14.  | Audit                                       | 28 |
| 14.1 | Audit                                       | 28 |
| 15.  | Dispute resolution                          | 29 |
| 15.1 | Negotiation                                 | 29 |
| 15.2 | Escalation                                  | 29 |
| 15.3 | Mediation                                   | 29 |
| 15.4 | Right to seek relief                        | 29 |
| 16,  | Liability                                   | 29 |
| 16.1 | Maximum liability                           | 29 |
| 16.2 | Exclusions                                  | 30 |
| 16.3 | Uncapped liability                          | 30 |
| 16.4 | Miligation                                  | 30 |
| 17.  | Force Majeure                               | 30 |
| 18.  | Insurance                                   | 30 |
| 19.  | Term and termination                        | 31 |
| 19.1 | Term  | 31 |
| 19.2 | Termination for default and insolvency      | 31 |
| 19.3 | Effects of expiry or termination            | 31 |
| 19.4 | Survival clauses                            | 31 |
| 20.  | Notices                                     | 31 |
| 21.  | General                                     | 31 |
|      |   |    |

| 33 |
|----|
| 34 |
| 39 |
| 43 |
|    |

Hage #

# Details

| Date            |   |
|-----------------|---|
| Parties         |   |
| Name            | The Sovereign in right of New Zealand acting by and through The Ministry of<br>Primary Industries with delegated financial authorisation to enter into the<br>agreement, by the Chief Operational Officer & Head of Biosecurity, Ministry for<br>Primary Industries |
| Short form name | MPI   |
| Notice details  | Address: PO Box 2526, Wellington 6140   |
|                 | Email: Wallaby@mpi.govt.nz  |
|                 | Attention: National Wallaby Management Programme  |
| Name            | Otago Regional Council, a body corporate under the Local Government Act 2002<br>having its head office at 70 Stafford Street, Dunedin   |
| Short form name | Recipient   |
| Notice details  | Address: 70 Stafford Street, Dunedin  |
|                 | Email: Sarah.Gardner@orc.govt.nz  |
|                 | Attention: Sarah Gardner  |

# Background

- A. Wallables are a serious pest in New Zealand and populations have been expanding numerically and geographically for a number of years. They have a significant impact on indigenous forest through browsing on undergrowth and regenerating plants. They compete with stock for pasture and fodder species, consequently reducing stocking rates. Wallables have also been known to damage fences, contribute to erosion and cause damage to plantation forestry seedlings.
- B. The issue is one that can only be addressed by landowners, community groups, iwi, industry, local and central government working together. This was recognised by the New Zealand Government in Budget 2020 providing funding of \$27 million over 4 years to establish a collaborative National Wallaby Management Programme (the National Programme).
- C. The New Zealand Government has allocated funding for managing wallables to be distributed through MPI.
- D. MPI wishes to distribute funding on a regional basis to selected regional councils for the purpose of that regional council carrying out activities that help manage wallables, as part of the National Programme. Each Region(s) will comprise specific Management Units and within those Management Units there will be Operational Areas, within which operational activities for wallaby management will be carried out.
- E. The Recipient is a regional council who wishes to manage wallables in its Region(s).
  - F. The establishment of the National Programme means a rapid expansion of the operational capacity of the Recipient's current wallaby management programmes. In addition to providing funding for the operational work to be carried out by the Recipient in relation to the National Programme, MPI intends to provide the Recipient with funding to cover other specified costs associated with programme management, such as contract management and reporting requirements. Funding for these programme management costs will be temporary, and it is anticipated by the parties that over time the Recipient will bare these costs itself and build these costs in to the Recipient's long-term plans.

G. As MPI and the Recipient are part of the public sector, each wishes to ensure that there is appropriate oversight across the Recipient's activities and use of public funding, focus on health and safety, and sharing of useful information. The partles wish that the "Regional Coordination Groups", of which the Recipient is a member, help provide oversight of the National Programme in its Region(s).

Hage 6

# Agreed terms

# PART A: OPERATIONAL AND COMMERCIAL DETAILS

# 1. Outcomes

- (a) The intended outcomes for this Agreement (the Outcomes) are to:
  - help to contain and reduce the number of wallables in New Zealand such that progressive containment is achieved;
  - support the development and implementation of a strategy for the National Programme to be approved by the Governance Group for the National Programme (New Zealand Wallaby Management Strategy), with initial outcomes expected to be, in particular, to:
    - (A) co-ordinate operations by faciliating co-ordinated control amongst land owners and other stakeholders; and
    - (B) promote information sharing of best practice and technological gains in control methods;
  - support the Recipient to tailor Outcomes to reflect the "Regional Pest Management Plans" for the Region(s) for which the Recipient is responsible;
  - (iv) encourage and, where reasonably practicable, implement feedback from the Regional Coordination Group(s) for the Region(s) for which the Recipient is responsible;
  - build and maintain a collaborative and constructive relationship between the parties and other stakeholders to address and manage the critical overarching issues caused by wallables;
  - (vi) support the Recipient to ensure the health and safety of Workers and others involved in and/or affected by the Activities, as required by law; and
  - (vii) for the Region(s) in which the Recipient operates, enable oversight of the operational programme and use of the Funding.
- (b) In performing the Activities, the Recipient will use all reasonable endeavours to contribute to the achievement of the Outcomes.

# Acknowledgement of appropriation

Each party acknowledges that the Funding under this Agreement is provided under the 'Border and Domestic Biosecurity Risk Management MCA appropriation of Vote Agriculture, Biosecurity, Fisheries and Food Safety'. The scope of the appropriation is 'Biosecurity Incursion Response and Long-Term Post Management' (Scope of Appropriation). Regardless of any other clause in this Agreement, the Recipient agrees that MPI does not provide the Funding, and the Recipient will not use the Funding, for any purpose outside the Scope of Appropriation.

# 3. Activities

#### 3.1 In scope

The Recipient will perform the Activities and provide the corresponding Outputs, as each is set out in the table below. Where indicated in that table, the Recipient:

- may use the Funding to perform the Activity, where indicated as "Funded" in the corresponding column (a Funded Activity); and
- (b) will perform each Activity at its cost, where indicated as "Recipient's cost" in the corresponding column.

|     | N310F   | Papu   | n for al felt - Calle<br>Ca |
|-----|---|--|-----------------------------|
|     | Operation   | al Planning  |                             |
| (a) | For each Financial Year during the term of this Agreement, and for each<br>Management Unit in the Region(s) that the Recipient is responsible for,<br>prepare (or procure the preparation of) a draft Management Unit Operational<br>Plan using the template provided by MPI.   | Each drait Management Unit Operational Plan is reviewed and,<br>by 1 May each year, is endorsed in writing by the:<br>1. RCG; and<br>2. MPI Programme Manager.             | Recipient's cost            |
| (b) | <ul> <li>If one does not already exist, set up a Regional Coordination Group for the Region(s) that the Recipient is responsible for and:</li> <li>(i) ensure that the RCG membars adequately represent partner organisations, including local iwi and stakeholders with an interest in the National Programme; and</li> <li>(ii) ensure the RCG meets periodically (and at the frequency to be datermined by the RCG) for the purpose of ensuring:</li> <li>that operational activities (to be performed under the MUOP) are planned in a cohesive and coordinated way:</li> <li>that operational activities (to be performed under the MUOP) achieve the objectives in New Zealand Wallaby Management Strategy (once approved by the Governance Group);</li> <li>alignment between the Aotivities carried out in relation to the National Programme and any other wallaby management work funded through other mechanisms;</li> <li>effective communication between the organisations responsible forwallaby management in the Region(s);</li> <li>that progress on the operational activities (to be performed under the MUOP) are communicated with local communities including relevant land occupiers; and</li> </ul> | The Regional Coordination Group operates effectively to<br>represent the key stakeholders of that particular Region(s)<br>and the Management Unit(s) within that Region(s. | Recipient's cost            |
|     | <ul> <li>Issues and risks are identified and dealt with or escalated to MPI.</li> </ul>   |  | (*                          |

|     | 10-1   | 0-2-   | Fund-Salari Proc. — II.<br>Kati   |
|-----|--|--|---|
| (c) | Implement (or procure the implementation of) the Management Unit<br>Operational Plan in accordance with its terms, including by engaging<br>Personnel and Subcontractors to implement that plan and help deliver the<br>Outcomes.  |  | Funded  |
| (d) | Managing its Personnel angaged for the purpose of implementing the<br>relevant Management Unit Operational Plan to ensure the Personnel carries<br>out their work and engagement in accordance with the terms of that plan and<br>this Agreement.  |  | Funded up to \$27,000<br>for the first Financial<br>Year. Any further cost<br>is at the Recipient's<br>cost.<br>MPI in its sole<br>discretion may review<br>whether to continue<br>that Funding (or any<br>other amount) for any<br>subsequent Financial<br>Year. |
|     | Financial and Ope  | erational Reporting  |   |
| (e) | Comply with any Wall-IS user guide provided by MPI to the Recipients, as<br>may be amended from time to time (Wall-IS User Guide).   | Keep accurate, up to date records showing the delivery of the<br>Activities against the relevant plan, including by taking minutes<br>and complete actions arising from regular meetings.          | Recipient's cost  |
| (I) | Ensure that all data is uploaded to Wall-IS in an accurate and timely manner,<br>in each case, as described and in accordance with any Wall-IS User Guide<br>and MPI's reasonable instructions given from time to time. For each<br>Region(s) the Recipient is responsible for providing reports to MPI (or if<br>directed by MPI, into Wall-IS) on the following: | Provide adequate information and reports to MPI at meetings.<br>Take minutes and complete actions arising from meetings held<br>with MPI ("Operational Meeting & Reporting Cycle" table<br>below). |   |
|     | (i) all operational activities (to be performed under the MUOP) completed in<br>each month (including method, location, area (length, if lencing),<br>contractor, start date, end date, lotal person hours worked, costs<br>expended), to be reported within 5 Business Days of the end of that<br>month; and  |  | 1   |

Page 1

|     | Asteria  | 6.5mm   | Constant de la Francisca de la constante |
|-----|--|---|--|
| -   | <ul> <li>(ii) number of people who are involved in paid work, to be reported within 5<br/>Business Days of the end of that month; and</li> </ul>   |   |  |
|     | (iii) location and number of wallaby sightings, to be reported within the<br>timeframe reasonably requested by MPI.  |   |  |
| (g) | Input (or procuring the inputting of) the financial data into Wall-IS in<br>accordance with any Wall-IS User Gulde and MPI's reasonable instructions<br>given from time to time, and do so accurately to reflect (amongst other things)<br>the costs of completed operations, and: |   |  |
|     | (i) Funding  |   |  |
|     | (ii) any other funding; and  |   |  |
|     | (iii) In Kind Contributions.   |   |  |
|     | If that inputting functionality is not available on Wall-IS, the Recipient must<br>provide that data to MPI in the form and forma that MPI reasonably requests.  |   |  |
| (h) | Comply with any audit under clause 14.1 of Part B.   |   | Funded (subject to<br>clause 14.1(e))    |
|     | Health and Safe  | ety Management  |  |
|     | Appendix A sets out the Recipient's health and safety obligations in further data<br>below and Appendix A, Appendix A will prevail.  | all. In the event there is any inconsistency between item (i) to (m)                        | 1.1                                      |
| (i) | Work with MPI to ensure a Health and Safety Management Plan (HSMP) is in<br>place and agreed prior to the commencement of any Funded Activity.   | HSMP is documented and agreed with MPI prior to the<br>commencement of any Funded Activity. | Recipient's cost                         |
| (1) | For each POBU, ensure a HSMP is in place prior to the commencement of<br>any Funded Activity.  |   | Recipient's cost                         |
| (k) | Ensure each Health and Safety Incident is reported to MPI within the<br>timeframes below:  |   | Recipient's cost                         |

Page 10

|     | 1 alway   | (C)(I)   | $\frac{1}{2} \frac{1}{2} \frac{1}$ |
|-----|---|--|--|
|     | <ul> <li>Notifiable Event within 24 hours of becoming aware of the Notifiable<br/>Event and following confirmation of event being notified to WorkSafe NZ;</li> <li>Serious Incident as soon as practicable and within 5 Business Days of<br/>becoming aware of the Serious Incident;</li> <li>Incident within 5 Business Days of becoming aware of the Incident; and</li> <li>Near Miss within 5 Business Days of becoming aware of the Near Miss.<br/>Notifiable Events will be notified to the MPI Operations Lead (or delegate) by<br/>teleptione in order to confirm investigation roles, responsibilities and actions.</li> </ul> |  |  |
|     | Serious Incidents, Incidents and Near Misses should include the results of<br>the health and satety investigation that followed the Serious Incident, Incident<br>or Near Miss and be submitted by inputting the relevant information into Wall-<br>IS in accordance with the Wall-IS User Guide. It that inputting functionality is<br>not available on Wall-IS, the Recipient will provide the results to MPI in the<br>form and format that MPI reasonably requests.   |  |  |
| (1) | Ensure health and safety audits are completed and any required<br>improvements that are identified during an audit are implemented.   |  | Recipient's cost   |
|     | Make the results of audits available to MPI when requested.   |  |  |
| (m) | Review and update the HSMP agreed under this Agreement in periodic<br>meetings with MPI.  | Meet at frequency agreed with MPI in order to cooperatively<br>ensure that health and safety risks and hazards related to<br>operations are being managed appropriately. | Recipient's cost   |
|     | Reporting a   | nd meetings  |  |
| (n) | Attend any meeting with MPI that MPI reasonably request.  |  | Recipient's cost   |
| (0) | Report to MPI in accordance with the "Operational Meeting & Reporting<br>Cycle" table below (or as otherwise agreed in writing between the parties), At<br>any "face to face" meeting, each party will endeavour to meet physically in<br>person or, where a physical meeting is not possible or desirable, by video-<br>conterence instead.  |  | Recipient's cost   |

Poge 11

Page 12

| Month     | Туре            | Meeting Objectives   |  |
|-----------|-----------------|--|--|
| September | Conference call | Confirm forecast financial contributions from partners.<br>Confirm Subcontractor/s for each operation (or agree dates when this will be confirmed).<br>Confirm that a HSMP is in place for each PCBU (or agree dates when this can be completed).<br>Current operations and spend - % complete.<br>Identify and agree resolution actions for key operational issues.<br>Agree actions and next meeting agenda. |  |
| December  | Conference call | Current operations and spend - % complete.<br>Review risks/issues (includes health and safety).<br>Confirm financial contributions received from partners (YTD).<br>Confirm In Kind Contributions made by partners (YTD).<br>Agree actions and next meeting agenda.  |  |
| March     | Face to Face    | Agree Management Unit operational planning process for upcoming Financial Year.<br>Confirm date for operational plans to be submitted.<br>Current operations and spend - % complete.<br>Review risks/issues (includes health and safety).<br>Forecast budget for upcoming Financial Year.  |  |
| Мау       | Conterence call | Confirm approved operations and budget for upcoming Financial Year (or actions required to complete).<br>Confirm actions to complete sign off for any variations to this Agreement and the supporting HSMP.<br>Agree upcoming Financial Year meetings schedule.<br>Agree annual debrief date and attendees.  |  |

Page 13
| Month     | Туре            | Meeting Objectives   |  |
|-----------|-----------------|--|--|
| Late June | Conference call | Confirm whether all previous Financial Year data is in Wall-IS or, if the relevant functionality to input that data in Wall-IS is not available, that the data has been provided to MPI. |  |
|           |                 | Confirm complete operations and spend for Financial Year (includes journals for expected invoices).  |  |
|           |                 | Confirm financial contributions from each co-funding partner and land-occupier, as identified in the MUOP (Partners),  |  |
|           |                 | Confirm In Kind Contributions.   |  |

#### 3.2 Restricted and/or Prohibited Activities

The Recipient must not use the Funding for the following prohibited activities (**Prohibited** Activities). MPI may add, remove and/or amend any of the Prohibited Activities from time to time by giving written notice to the Recipient.

| Prohibited Activity | Comments |  |
|---------------------|----------|--|
|                     |          |  |

# 4. Funding

#### 4.1 Purpose

The Recipient will only use the Funding for the purpose of carrying out the Funded Activities.

#### 4.2 Maximum amount

Regardless of anything else in this Agreement, MPI will not be obliged to provide Funding under this Agreement which, in total, exceeds the following amount for the corresponding Region. The amounts in the table may change from Financial Year to Financial Year by written notice from MPI, which may be given only after consultation with the Governance Group:

| Region | Financial Year | Maximum amount of<br>Funding (plus GST if any) |
|--------|----------------|--|
| Otago  | 2020/21        | \$373,000                                      |
| Otago  | 2021/22        | TBC  |
| Otago  | 2022/23        | TBC  |
| Otago  | 2023/24        | TBC  |

#### 4.3 Method of payment

The Recipient will invoice MPI by sending a valid Tax Invoice to the MPI Operations Lead in accordance with clause 6 of Part B. Each Tax Invoice may only be issued when the involcing criteria (as notified by MPI to the Recipient from time to time) for that Tax Invoice has been met to MPI's reasonable satisfaction.

# 4.4 Other provisions

Further detail regarding the method of payment, eligible expenditure, financial management, funding freezes and repayment is set out in clause 6 of Part B.

5. Governance

#### 5.1 Role and responsibilities

The party described as "appointor" in the table below will engage and keep engaged suitably qualified, skilled and experienced Personnel in the roles set out in the table below. Each party will ensure its appointee performs the responsibilities for that role, as described in the table below and elsewhere in this Agreement.

| Name of Role  | Responsibilities   | Appointer     |
|---|--|---------------|
| Representative on the<br>Regional Coordination<br>Group | <ol> <li>Have oversight and involvement<br/>in the development,<br/>endorsement-seeking process,<br/>implementation and monitoring<br/>of each relevant MUOP.</li> </ol> | The Recipient |

| Name of Role                              | Responsibilities  | Appointer  |  |
|---|---|--|--|
|   | <ol> <li>Keeps up to date on progress<br/>towards delivering the MUOP.</li> </ol>   |  |  |
|   | <ol> <li>Is involved in risk mitigation (all<br/>informed and involved).</li> </ol>   |  |  |
|   | <ol> <li>Assist in coordination of control<br/>work (and other related work).</li> </ol>  |  |  |
| Representation on the<br>Governance Group | To help govern the National<br>Programme, and oversee associated<br>Crown funding and National<br>Programme operational delivery. | MPI and, where the<br>Recipient is part of LGNZ,<br>through the Recipient's<br>LGNZ representative |  |

# 6. Term and termination

#### 6.1 Term

| Start Date             | 13 October 2020   |  |
|------------------------|---|--|
| End Date               | 30 June 2024  |  |
| Extending the End Date | The parties may extend the term of this Agreement by written agreement. |  |

#### 6.2 Termination of this Agreement

In addition to the rights contained in clause 19 of Part B, MPI or the Recipient may terminate this Agreement at any time for convenience, without cause, by giving at least four (4) months' prior written notice to the other party.

#### 7. Contact details

- (a) All matters or enquiries regarding the technical implementation or operation of this Agreement will be directed to MPI's Technical Liaison or the Recipient's Technical Liaison, as appropriate.
- (b) All general matters relating to this Agreement (including concerning the interpretation of this Agreement) will be directed to MPI's Agreement Manager or the Recipient's Agreement Manager, as appropriate.
- (c) If the Recipient cannot contact MPI's Agreement Manager or MPI's Technical Liaison within a reasonable timeframe in the circumstances, enquiries can be directed to the person for the time being holding the office of "MPI Manager Recovery and Pest Management".
- (d) If MPI cannot contact the Recipient's Agreement Manager or the Recipient's Technical Liaison, enquiries can be directed to the Recipient's Chief Executive or Managing Director.
- (e) The initial contact persons for each party are below. If a party's contact persons or their details change, that party must notify the other party in writing 10 Business Days in advance.

| Party     | Role                    | Details   |
|-----------|-------------------------|---|
| MPI       | Manager Pest Management | John Sanson<br>04 894 0836<br>John,Sanson@mpi.govt.nz                     |
| MPI       | Technical Liaison       | Andrew (AJ) Stewart<br>029 943 1512<br>Andrew.Stewart@mpi.govt.nz         |
| MPI       | Agreement Manager       | Sam Beaumont<br>022 011 5880<br>Samuel.Beaumont@mpl.govt.nz               |
| Recipient | Chief Executive         | Sarah Gardner<br>Sarah.Gardner@orc.govt.nz                                |
| Recipient | Managing Director       | Gavin Palmer<br>General Manager of Operations<br>Gavin.Palmer@orc.govt.nz |
| Recipient | Technical Liaison       | Richard Lord<br>03 474 0827<br>Richard.Lord@orc.govt.nz                   |
| Recipient | Agreement Manager       | Andrea Howard<br>Andrea.Howard@orc.govt.nz                                |

# PART B: GENERAL TERMS AND CONDITIONS

1. Activities

MPI agrees to provide the Funding to the Recipient, and the Recipient agrees to use the Funding and carry out the Activities (or ensure the Activities are carried out), subject to and in accordance with the terms of this Agreement, and will use the Funding solely for the purpose of performing the Funded Activities under the National Programme.

2. Obligations

#### 2.1 Mutual obligations of the parties

- (a) Each party agrees to:
  - act in good faith in all matters relating to this Agreement and, without abandoning their own interests, to demonstrate honesty, integrity, openness, reasonableness, and accountability in their dealings with each other; and
  - discuss any matters affecting this Agreement or the performance of the Aclivities, whenever necessary.
- (b) Each party will ensure it performs its obligations under this Agreement:
  - with due diligence, care and skill and using a standard of skill, care and diligence no lower than that expected of an experienced and well-resourced organisation performing that obligation;
  - (ii) in accordance with the specified timeframes or, if no timeframe is specified, then within a reasonable period; and
  - (iii) in accordance with all applicable laws, regulations and, where legally binding upon that party, codes and standards of practice.

#### 2.2 Recipient's general obligations

- (a) The Recipient will ensure that it provides the Activities, and will ensure that any Personnel engaged to provide the Activities does so:
  - (i) using appropriately trained, qualified, experienced and supervised persons;
  - (ii) in accordance with all Government or MPI internal policies and procedures relevant to this Agreement, as notified in writing to the Recipient at or before the Start Date;
  - (iii) in accordance with the Standards of Integrity and Conduct issued by the Public Service Commission (see <u>www.publicservice.govt.nz</u>); and
  - in a manner that:
  - (iv) is culturally appropriate for Maori, Pacific and other ethnic or indigenous groups;
  - (v) respects the personal privacy and dignity of all participants and stakeholders; and
  - meets MPI's satisfaction, as reasonably described by MPI in writing from time to time.
- (b) Unless otherwise agreed in writing, the Recipient will ensure that it obtains all necessary and prudent authorisations, consents, permits, licences and the like, to ensure the Recipient can perform its obligations under this Agreement in compliance with the law and any third party's rights, including in relation to performance of any Activities carried out on MPI's or a third party's premises.

#### 2.3 MPI's obligations

If reasonably requested by the Recipient, and without limiting the Recipient's obligations under this Agreement, MPI will:

 cooperate and collaborate with the Recipient in the same way that MPI generally cooperates and collaborates with other recipients similar to the Recipient in the relevant circumstances;

- (b) provide or give access to MPI's resources and information, including templates, which MPI generally makes available to recipients similar to the Recipient, for the purpose of the National Programme; and
- (c) endeavour to find opportunities to share best practices and lessons learned across the National Programme's delivery.

#### Health and safety

#### 3.1 Compliance with health and safety legislation and directions

(a) During the term of this Agreement, the Recipient will comply with its health and safety obligations as set out at Appendix A and Appendix B.

#### Subcontractors and Personnel

#### 4.1 Subcontractors

- (a) The Recipient must ensure that:
  - each Subcontractor is fully aware of the Recipient's obligations under this Agreement to the extent necessary for the Subcontractor to properly perform its obligations;
  - each subcontract it enters into with a Subcontractor is consistent with this Agreement, to the extent relevant and material for the performance of the Subcontractor's obligations; and
  - each subcontract with a Subcontractor requires that further subcontracting is also consistent with this Agreement.
- (b) The Recipient must ensure that, prior to the performance of the Activities by any Subcontractor, the Subcontractor has been advised of, and agrees to comply with:
  - (i) the Recipient's health and safety policy and procedures:
  - the health and safety risk management plan for the relevant PCBU (as described in (j) of the Activities table in Part A);
  - the plan to manage any specific site within the relevant Management Unit based on health and safety issues in performing the Activities; and
  - (iv) any health and safety directions given by MPI to the Recipient under paragraph 2.1(c) of Appendix A.
- (c) The Recipient will be responsible for the acts and omissions of each Subcontractor as if the act or omission was the Recipient's. The Recipient will not be relieved of any of its liabilities or obligations under this Agreement by entering into any subcontract.
- (d) The Recipient will ensure that:
  - (i) its contract with each Subcontractor it engages in relation to the National Programme contains generally the same obligations and it gives MPI the same rights as are set out in this Agreement to the extent that those obligations and rights relate to the Subcontractor's performance of the Activities; and
  - MPI can directly enforce those obligations and exercise its rights against the Subcontractor pursuant to the Agreement and Commercial Law Act 2017, provided that MPI will not:
    - enforce its rights under 4.1(d)(i) without seeking to resolve any issues in the first instance with the Recipient; or
    - (B) engage with Subcontractors without the prior written consent of the Recipient, such consent not to be unreasonably withheld.

# 4.2 Personnel

If MPI is at any time dissatisfied on reasonable grounds with the performance of the Recipient's Personnal in its performance of the Recipient's obligations under this Agreement:

- (a) MPI may notify the Recipient in writing of that dissatisfaction; and
- (b) the parties, in good faith, will discuss an appropriate course of action in relation to that Personnel.

If the Personnel performs a role that involves the accounting or allocation of Funding and has been arrested for or committed a dishonesty offence, the Recipient will remove that Personnel from that role immediately, if requested by MPI.

#### Significant Events process

#### 5.1 Notification of Significant Event

Each party will notify the other party as soon as reasonably practicable upon becoming aware of an actual or likely:

- Serious failure: significant failure (or series of related failures which together are significant) of the Recipient to perform its obligations in accordance with this Agreement;
- (b) MPI Breach: significant event, breach or likely breach by MPI of its obligations under this Agreement;
- (c) Reputational Event: any event relating to the Agreement which is likely to cause MPI, the government, the Recipient, its Personnel and/or Subcontractors significant adverse media or parliamentary attention, and in any event, if there is a change in the criminal record of the Recipient, and/or its Personnel;
- (d) Security Event: security event, breach or likely breach relating to the Agreement that compromises or is likely to compromise the integrity or security of MPI, the Recipient, the Activities, Funding, MPI assets (including any technology system) or MPI Data (such event, breach or likely breach being a Security Event); and
- (e) Regulatory Event: any event, or a near miss, relating to the Agreement that would cause the Recipient to notity a regulator, including any health and safety regulator, the office of the Privacy Commissioner or a financial services regulator,

(each such event, circumstance, breach or failure being a Significant Event).

#### 5.2 Rectification process

If a party advises the other of a Significant Event, and MPI requests, the parties will endeavour to agree a course of action. If no course of action is agreed within the timeframes advised by MPI (which, depending on the nature of the Significant Event, could be required within an hour), the Recipient must:

- (a) Co-operate: co-operate with MPI and any other third party specified by MPI;
- (b) Follow instructions: follow any instructions given by MPI, including any instruction given:
  - (i) under clause 6.7(b) of this Part B; and/or
  - (ii) to stop following a previous instruction (or any part of it),

and, if the Significant Event is related to a Notifiable Event, near miss, or breach, MPI's instruction may include:

- (iii) to stop performing the Activities (or any part of them):
- (iv) to suspend any work being carried out under or in relation to this Agreement by the Recipient, and/or its Personnel until MPI (in its sole discretion) considers the health and safety risk has been eliminated or adequately mitigated; and/or
- (v) to amend the HSMP in the manner MPI instructs.

- (c) Investigate: investigate the cause of the Significant Event and give a report to MPI (in writing if MPI requests) which describes (to the extent reasonably known):
  - (i) the cause of the Significant Event;
  - where the Significant Event is a Security Event and to the extent known, the identity of the person or persons who have gained or attempted to gain unauthorised access; and
  - (iii) the actions proposed to be taken by the Recipient.

If MPI requests, allow MPI to be involved in relation to that investigation to the level MPI requests.

- (d) Update: update MPI on regular basis of any subsequent developments relating to the Significant Event.
- (e) Remedy/Mitigate: remedy or mitigate the immediate effects of the Significant Event, or the circumstances or issue giving rise to the Significant Event, as soon as reasonably practicable, in the manner reasonably requested by MPI and provide MPI with a report (in writing, if MPI requests) of the action taken. MPI will take reasonable steps to remedy or mitigate the effects of the Significant Event on MPI.
- (f) Prevent Recurrence: take all reasonable action to prevent a recurrence of the relevant Significant Event and notify MPI (in writing if MPI requests) of any such action taken.
- (g) Remedy plan: within the period reasonably requested by MPI, prepare and submit to MPI for its approval a remedy plan setting out how the Recipient considers the Significant Event could be optimally remedied, the relevant timeframes for such remediation, what the Recipient considers would be required of the Recipient, MPI and/or any third party and any other details reasonably required in order for the Significant Event to be remedied in an optimal manner. If MPI approves that submitted plan, each party will comply with the requirements of that party as set out in that plan and any other requirements notified by MPI (acting reasonably).

#### 5.3 Cost of remedying

MPI will pay the Recipient for the actual reasonable costs incurred by the Recipient in complying with clause 5.2 of this Part B but only:

- up to the amount of the costs that was advised by the Recipient to MPI in advance of the Recipient incurring those costs; and
- (b) to the extent that the Significant Event is not due to the Recipient's or its Personnel's breach or misconduct.
- 5.4 Other remedies

The issuing of a notice under clause 5.1 of this Part B does not limit either party's other rights and remedies,

6. Payment

#### 6.1 MPI's obligation to pay the Funding

MPI's obligation to pay the Funding is subject to:

- (a) MPI continuing to have sufficient funding within its budget for the Funded Activities;
- (b) there being no un-remedied breach of this Agreement by the Recipient; and
- (c) MPI being satisfied on reasonable grounds that the Funding is being appropriately expended on the Funded Activities and is not being used in breach of this Agreement.

#### 6.2 Total amount payable

The Funding is the total amount payable by MPI for the provision of the Funded Activities. All other costs, disbursements and other expenses incurred by the Recipient in relation to this Agreement are at the Recipient's cost and not to be paid for by the Funding.

# 6.3 Invoicing

- (a) The Recipient will invoice the Funding In accordance with this clause 6.3 of Part B. Each Tax Invoice submitted by the Recipient (whether submitted monthly or otherwise) will include sufficient details to enable MPI to Identify:
  - (i) the Agreement number (if any);
  - (ii) the period covered by the invoice;
  - (III) the particular Funded Activities which are the subject of the invoice; and
  - (iv) the amount of Funding payable.
- (b) Unless provided otherwise in this Agreement, the Recipient:
  - (i) will provide MPI with aTax Invoice for the aggregate amount of Funding for each Management Unit for the relevant Financial Year, as set out in the table at clause 4.2 of Part A (as those amounts may be amended in accordance with clause 4.2 of Part A); and
  - may provide that Tax Invoice annually in advance for the immediately following Financial Year.
- (c) Without limiting MPI's other rights under this Agreement, MPI will pay the Recipient's Tax Invoices by the within 20 Business Days of receiving that invoice or before the beginning of the Financial Year to which the invoice relates, whichever is the later date. All payments by MPI will be deemed to have been made in Wellington, New Zealand. Payment by MPI is not evidence that the Activities to which the invoice relates have been provided in accordance with this Agreement.

#### 6.4 Disputed invoices

If MPI has a bona fide dispute in relation to all or any portion of any Tax Invoice, MPI may withhold payment of the amount subject to the dispute, provided that:

- (a) MPI will pay the undisputed amount when it becomes due and payable; and
- (b) the Recipient will continue to perform its obligations under this Agreement while the dispute is resolved.

#### 6.5 Eligible Expenditure

The Recipient must apply the Funding only to expenditure that:

- (a) is incurred by the Recipient;
- (b) for the purpose of the Funded Activities or for purposes incidental to the Funded Activities;
- (c) is incurred during the term of this Agreement; and
- (d) is otherwise approved in writing by MPI (in its absolute discretion) as eligible expenditure for the purposes of this Agreement,

#### (Eligible Expenditure).

## 6.6 Financial Management

- The Recipient must:
- ensure that any payments of Funding made to a third party in connection with this Agreement (including to its Personnel) are correctly made and properly authorised;
- (b) maintain proper and diligent control over the incurring of all liabilities in relation to the Funding;
- (c) maintain an appropriate linancial management system to ensure that the Funding is separately identified and managed within its accounts; and
- (d) except with MPI's prior written approval, not use the Funding for the purposes of a guarantee or security for any loan, credit, payment or other interest, or in the context of any litigation.

# 6.7 Funding Freeze

- (a) Without limiting any other right or remedy of MPI, if the Recipient materially breaches this Agreement, then MPI may issue the Recipient with a notice directing the Recipient not to spend the Funding (Funding Freeze Notice).
- (b) If the Recipient receives a Funding Freeze Notice under this clause 6.7 of this Part B, then, until the breach is remedied to MPI's satisfaction, the Recipient must not spend any Funding:
  - (I) unless and until MPI notifies the Recipient otherwise; and
  - except to the extent required to pay for any bona lide and unavoidable pre-existing contractual commitments to spend the Funding in accordance with this Agreement.

#### 6.8 Repayment

- (a) Without limiting any other right or remedy of MPI, MPI may recover Funding from the Recipient in the following circumstances (and the Recipient will pay that Funding within 5 Business Days of MPI's written notice requiring payment):
  - (i) Misspent Funding: At any time MPI may recover the amount of any Funding that has been spent or used other than in accordance with this Agreement, together with interest on all such amounts calculated at 10% per annum from the date of the misspending to the date the money is repaid.
  - Uncommitted Funding: On expiry or termination of this Agreement, MPI may recover any Funding which the Recipient:
    - has not spent or contractually committed to spend in accordance with this Agreement; and/or
    - (B) has spent or contractually committed to spend in accordance with this Agreement but which the Recipient can have refunded or released from that commitment, and the Recipient must use all reasonable endeavours to obtain such refund or release.
  - (iii) Activities Abandoned: MPI may recover an amount up to the total value of the Funding less the amount that the Recipient spent in accordance with this Agreement if the Recipient:
    - states an intention not to perform the Activities in accordance with this Agreement; and/or
    - (B) within 10 Business Days of being requested to do so by MPI, does not demonstrate (to MPI's reasonable satisfaction) that the Recipient will proceed with performing the Activities in accordance with this Agreement.
  - (iv) Excess Funding: Where the total Funding paid under this Agreement and any other money received by the Recipient to carry out the Funded Activities exceeds or is likely to exceed the funding required to perform the Activities, the Recipient must notify MPI upon becoming aware and, if MPI requests, promptly refund to MPI the excess amount.
- (b) If the Recipient does not pay, as required by this clause, the Recipient must pay MPI interest calculated at 10% per annum from the date payment is due until the date the money is repaid.

#### 6.9 Insufficient Funding

If the Recipient uses (or reasonably expects to use) all of the Funding in accordance with this Agreement without completing its performance of the Funded Activities during the term of this Agreement, then:

(a) the Recipient will give MPI written notice as soon as reasonably practicable (but the Recipient does not need to give more than six (6) months advance notice), detailing the amount of additional funding requested by the Recipient (Additional Funding) and

supporting evidence as to why the Additional Funding will be sufficient for the Recipient to complete the performance of the Funded Activities; and

- (b) within a reasonable period of time after receiving the Recipient's request for Additional Funding under clause 6.9(a)above, MPI will give the Recipient written notice as to whether it will provide:
  - the Additional Funding in full (in which case the Recipient will be obliged to complete the performance of the Funded Activities in accordance with this Agreement); or
  - part of the Additional Funding (in which case the parties will follow the change control procedure set out at clause 7 to agree upon a reduction in the scope of Funded Activities to be performed by the Recipient); or
  - (iii) no Additional Funding (in which case the Recipient will not be obliged to continue to perform the Activities under this Agreement after all of the funding has been spent on the Funded Activities in accordance with this Agreement).

# 6.10 Tax

All amounts set out in this Agreement are stated before the addition of GST, which, it payable, will be paid at the same time as the supply to which the GST relates. Subject to the previous sentence, MPI will not be obliged to pay any of the Recipient's and/or its Personnel:

- (a) taxes, duties, excess, levies, carbon credits;
- (b) holiday pay, sick pay or other payment under the Holidays Act 2003;
- (c) redundancy or any other form of severance pay; and/or
- (d) other amounts in relation to the performance of its obligations under this Agreement that are not the performance of the Funded Activities.

MPI will not be required to gross up any payment or Funding to the Recipient where MPI deducts tax due to non-resident contractors' tax.

7. Change control procedure

#### 7.1 Change request

A party may request an amendment to the terms of this Agreement at any time by providing a written change request to the other. Unless the parties' respective Agreement Managers agree in writing to use a truncated process, each party agrees to follow the process set out in this clause 7 when requesting a change to the terms of this Agreement.

#### 7.2 Understanding consequence of change request

- (a) Where MPI submits a change request, it will provide a reasonable description and explanation of the change sought.
- (b) Where the Recipient submits a change request, or within a reasonable period of time of receiving a change request from MPI, the Recipient must provide MPI with a reasonable description of:
  - how the proposed change would be implemented, including details of how the requested change would affect the Activities and any other term of this Agreement;
  - the cost of implementing the change and the Recipient's calculation of changes to the Funding (if any), which must be reasonable and calculated in good faith;
  - (iii) any information reasonably requested by MPI; and
  - any other relevant information, which may include how any change request can be most optimally implemented.

No change request will be binding unless each party accepts the change request in writing.

## 8. Information management

#### 8.1 Provision of information

Each party will:

- provide the other party with all information relating to the Activities as reasonably requested from time to time, within the timeframe reasonably requested; and
- (b) as soon as reasonably practicable, provide the information requested by the other party to comply with its statutory, parliamentary or other public sector reporting obligations; and
- (c) ensure that all information provided in writing to the other party under or in connection with this Agreement is materially accurate, complete and not misleading at the time the information was provided, for the purpose for which it was provided.

#### 8.2 Privacy

- (a) Each party must act in accordance with the Privacy Act 1993.
- (b) Each party will not use any Personal Information acquired in the course of performing its obligations under this Agreement, or disclose that Personal Information to any person other than MPI or the Recipient (as applicable) or the individual to whom the information relates, except:
  - (i) to the extent reasonably required to perform its obligations under this Agreement;
  - (ii) In accordance with the Privacy Laws; or
  - (iii) with the other party's or person's written consent.

#### 8.3 Recordkeeping

The Recipient will (and will ensure that each of its Subcontractors under and in relation this Agreement):

- (a) keep full, accurate and up-to-date records, including:
  - (i) financial records relating to the performance of the Activities and all monies paid and payable by MPI under or in relation to this Agreement sufficient to enable MPI to meet its obligations under the Public Finance Act 1989; and
  - any health and safety event which has been investigated internally by the Recipient, or a health and safety regulator, relating to the Activities (and the outcome of such investigations); and
  - (iii) any health and safety due diligence processes, compliance records, and any associated documents, relating to the health and safety compliance and performance of Workers and other Personnel involved in the Activities;
- (b) retain such records for 7 years from the expiry or termination of this Agreement, and
- (c) on request by MPI, and at no cost to MPI, provide to MPI copies of any such records reasonably requested by MPI and any information or materials (including financial information relating to the provision of Activities).

#### 9. Confidentiality

#### 9.1 Security and non-disclosure

Subject to clauses 9.2 and 9.3 of this Part B, each party agrees that it will:

- use all reasonable endeavours to keep the Confidential Information of the other party confidential at all times;
- (b) not Use, communicate, make available or re-supply the other party's Confidential Information to any person, other than for the purposes of performing its obligations and/or exercising its rights in accordance with this Agreement;
- effect and maintain all reasonable security measures to safeguard the other party's. Confidential Information from access, damage or use by unauthorised persons;

- (d) store all Confidential Information in New Zealand; and
- (e) ensure that any Personnel or professional adviser to whom a party discloses the other party's Confidential Information is aware of, and complies with, the provisions of this clause 9 of this Part B.

#### 9.2 Permitted disclosure

A party will not be bound by the obligations of confidentiality in clause 9.1 of this Part B to the extent that any disclosure of Confidential Information:

- (a) is required by law, convention, parliamentary rules or the rules of any stock exchange (Requirement), in which case, if permitted by the Requirement and as soon as reasonably practicable, the disclosing party will notify the other party of the Requirement, the information it will disclose and the date on which it will disclose the information, and will comply with that notice:
- (b) is required for the purposes of performing and enforcing this Agreement;
- (c) is of information that is publicly available through no fault of that party;
- (d) is made with the other party's prior written approval of the disclosure or (in the case of clause 9.1(d)) the transfer outside New Zealand;
- (e) which was rightfully received from a third party without restriction or without breach of this Agreement; or
- (f) was developed independently of the Confidential Information.

#### 9.3 Information sharing

In addition to clause 9.2 of this Part B, and in recognition of the mutual benefit of information sharing for the National Programme as a whole, the obligations of confidentiality in clause 9.1 of this Part B do not apply to the extent that any disclosure of Confidential Information is for the purpose of consulting, cooperating or coordinating with other then current recipients of funding from MPI under the National Programme, including in relation to:

- current best practice in relation to the performance of the Activities; and/or
- (b) a party's health and safety management under this Agreement.

#### 9.4 Mandatory disclosures

If any party is required to disclose any Confidential Information (as contemplated by clauses 8,1(a) or 9,2 of this Part B), that party will use its reasonable endeavours to obtain the recipient's commitment that it will treat the confidential information confidentially.

#### 9.5 Publicity and announcements

- (a) The Recipient will acknowledge MPI as a source of funding in all publications and publicity regarding the Activities, and will obtain MPI's approval of the form of the acknowledgement.
- (b) MPI may publicise and report on its distribution of Funding to the Recipient on any website, in any media releases, general announcements or annual reports. MPI may disclose the name of the Recipient, any of its Subcontractors, the amount of the Funding and a brief description of Activities.
- (c) Neither party may post on websites, social networking sites or publicly display objectionable or derogatory comments about the Activities, this Agreement, each other, or any of their Personnel.
- (d) The Recipient will promptly refer any enquiries from the media or any other person about the terms or performance of this Agreement to the MPI Programme Manager. If the MPI Programme Manager cannot be contacted, the Recipient will instead contact the person holding the office of "MPI National Procurement and Agreements Manager".

# 10. Warranties

# 10.1 Warranties

The Recipient warrants that at the date of this Agreement:

- (a) it is not insolvent or bankrupt and no action has been taken to initiate any form of insolvency or administration in relation to the Recipient;
- (b) there are no circumstances or matters that are likely to have a material adverse effect on the performance of its obligations under this Agreement;
- (c) it is not aware of any information that has not been disclosed to MPI which may, if disclosed, materially adversely affect the decision of MPI to provide the Funding;
- (d) all information relating to the Activities that was provided by the Recipient to MPI prior to MPI's execution of this Agreement, including in any proposal or presentation by the Recipient, was accurate, complete and true at the time it was provided. The Recipient acknowledges that MPI is entering into this Agreement in reliance on such information; and

on a continuing basis during the term of this Agreement (to the best of the Recipient's knowledge and belief):

- (e) the performance of the Activities under this Agreement will not constitute or give rise to a breach of:
  - the rights of any person (including any Intellectual Property Rights or physical property rights); or
  - (ii) any law with which the Recipient or Personnel is required to comply.

#### 11. Intellectual Property and data

#### 11.1 Background Intellectual Property

Except to the extent specified in this Agreement, nothing in this Agreement confers on a party any right, title or interest in the other party's Background Materials.

#### 11.2 New Intellectual Property

Unless otherwise agreed between the parties, all Intellectual Property Rights in the New Materials arising under or in relation to this Agreement are the property of MPI from the time of creation or will vest in MPI immediately afterwards.

#### 11.3 Licence

Subject to clause 11.4, each party (Licensor) grants the other (Licensee) a perpetual, irrevocable, world-wide, royalty-free, non-exclusive licence to Use any of the Licensor's:

- (a) Background Materials;
- (b) New Materials;
- (c) Third Party Materials (provided that the Licensor holds the rights to grant this licence in relation to Third Party Materials); and/or
- (d) any of the Licensor's data,

provided or made available by the Licensor to the Licensee under or in connection with this Agreement for the purpose of performing its obligations or exercising its rights under this Agreement. The Licensee may sub-license its rights under this clause 11.3 to its Subcontractors, but only to the extent necessary for that Subcontractor to perform the Activities or provide the Outputs in accordance with this Agreement.

11.4 Wall-IS

The Recipient must not (and will ensure its Personnel does not) copy, reverse engineer, rent, decompile or attempt to circumvent any security measures on Wall-IS or any part of it.

# 11.5 Know How

Notwithstanding any other provision of this Agreement, MPI and the Recipient will not be prevented or restricted by this Agreement from using any technique, idea, concept or know-how relating to this Agreement.

#### 12. Third party rights

#### 12.1 Third Party Claim

If a third party claims that a party (the Infringing Party) has infringed that third party's Intellectual Property Rights, other rights of that third party or the law (Third Party Claim), then:

- the other party will allow the Infringing Party to control settlement or defence of the Third Party Claim. The Infringing Party will consult with the other party on the settlement or defence of the Third Party Claim;
- (b) at the Infringing Party's cost and reasonable request, the other party will co-operate with the Infringing Party in the defence and any related settlement negotiations concerning any Third Party Claim; and
- (c) the other party will not enter into any settlement, negotiation or compromise with the third party without the Infringing Party's written consent (not to be unreasonably withheld).

#### 13. Conflicts of interest

# 13.1 Avoiding conflicts

Each party must maintain a high standard of honesty and integrity at all times in the performance of this Agreement and must use all reasonable endeavours to avoid any actual or perceived conflict of interest, being:

- (a) a situation which might create an actual or perceived conflict of interest, or compromise the Recipient's or MPI's integrity; and/or
- (b) any financial or other interest or undertaking that might interfere with or compromise the performance of the other party's obligations under this Agreement, or the integrity or standing of either party.

#### 13.2 Notifying conflicts

Each party must advise the other party in writing of any actual or perceived conflict of interest that might arise in the performance of its obligations under, or in connection with, this Agreement. Each party will assist the other party to manage, avoid or mitigate that conflict and follow any reasonable instructions (which may include stopping the performance of certain work) that the parties consider necessary or appropriate to manage the conflict of interest.

#### 14. Audit

- 14.1 Audit
  - (a) As part of the public sector, each party acknowledges the need to ensure transparency and accountability in the use of public funds, and also the cost and resource involved in carrying out an audit. Accordingly, MPI will work with the Recipient to identify an appropriate audit frequency and scope but, as the funder, MPI must have default rights of audit that it can use at any stage.
  - (b) The Recipient must allow MPI and any auditor nominated by MPI to conduct an audit from time to time, for the purpose of:
    - determining the level of compliance with this Agreement (including whether there has been a breach of this Agreement) by the Recipient and/or its Personnel;
    - determining whether Activities invoiced for or by the Recipient have been performed and invoiced according to this Agreement; and/or
    - (III) assisting in resolving a matter in dispute between the parties.

- (c) The Recipient will (and will ensure that each of its Personnel will) provide reasonable access and assistance to MPI (or an auditor appointed by MPI) in relation to an audit conducted under this clause 14.1 of this Part B.
- (d) In conducting an audit, MPI must:
  - give the Recipient reasonable notice of the audit, which in any event does not need to be more than 5 Business Days; and
  - (ii) comply with the Recipient's reasonable security requirements.
- (e) The Recipient may use the Funding for the reasonable costs of any audit unless a material non-compliance or charging error with this Agreement is discovered through the audit, in which case, the Recipient must reimburse MPI for the amount of any Funding that has been used to pay for the costs of the audit.
- (I) MPI will promptly notify the Recipient of the draft and final results of any audit conducted under this clause 14.1 of this Part B.
- 15. Dispute resolution

#### 15.1 Negotiation

Both parties will endeavour to resolve any dispute that may arise under or in connection with this Agreement through negotilation.

#### 15.2 Escalation

If the parties are unable to resolve a dispute by negotiation within 10 Business Days after the dispute was first notified in writing by one party to the other (**Dispute Notice**), and either party gives written notice to the other, then:

- the dispute will be referred for resolution by their respective senior responsible officers with oversight of this Agreement; and
- (b) if the dispute cannot be settled by the senior responsible officers within 20 Business Days of delivery of the Dispute Notice, the dispute will be referred for resolution by their respective Chief Executives.

#### 15.3 Mediation

Subject to clause 15.4 of this Part B, if a dispute cannot be settled within 30 Business Days of delivery of the Dispute Notice under clause 15.2 of this Part B (or such other timeframe agreed in writing by the parties), the dispute may be referred by either party to non-binding mediation in which case:

- (a) the parties agree to the mediation, which will be held in Wellington, New Zealand;
- (b) the mediation will be mediated by a sole mediator agreed in writing by the parties; or
- (c) if the parties cannot agree on a mediator within 10 Business Days of either party referring the dispute to mediation, either party may request the President of the New Zealand Law Society to appoint a suitably gualified independent mediator.

#### 15.4 Right to seek relief

This clause 15 of this Part B does not affect either party's right to seek urgent interlocutory and/or injunctive relief from any New Zealand Court of competent jurisdiction.

16. Liability

#### 16.1 Maximum liability

Without limiting clauses 16.2 and 16.3 of this Part B, the maximum aggregate liability, whether in contract, tort (including for negligence), breach of statutory duty or otherwise, of one party to the other party under or in connection with this Agreement is limited to the total amount of Funding paid by MPI under this Agreement.

# 16.2 Exclusions

- (a) Neither party is liable for any indirect or consequential loss or damage arising under or in connection with this Agreement, even if foreseeable and even if advised of the possibility of such loss or damages.
- (b) The Recipient will not be liable under this Agreement to the extent the Recipient's breach of this Agreement is a direct and unavoidable result of MPI's express written instruction (given knowingly).
- (c) Notwithstanding any other clause in this Agreement, this clause 16 applies only to the extent permitted by law.

# 16.3 Uncapped liability

- Clauses 16.1 and 16.2 of this Part B will not limit either party's liability under or in connection with this Agreement:
- (a) for damage to the other's or a third party's physical property;
- (b) for personal injury or death;
- (c) for fraud or wilful default; or
- (d) an obligation to pay any amount due and owing under this Agreement.

# 16.4 Mitigation

Each party must take reasonable steps to mitigate any loss or damage, cost or expense it may suffer or incur arising out of anything done or not done by the other party under or in connection with this Agreement.

- 17. Force Majeure
  - (a) Where either MPI or the Recipient (the Affected Party) is unable to carry out its obligations in accordance with this Agreement due to a Force Majeure Event, and:
    - the Affected Party gives the other party immediate written notice of the nature, the expected duration of, and the obligation affected by, the Force Majeure Event;
    - the Affected Party gives regular updates of the expected duration of and effects of the Force Majeure Event; and
    - (iii) the Affected Party uses all reasonable endeavours to:
      - mitigate the effects of the Force Majeure Event on the Affected Party's obligations under this Agreement; and
      - (B) perform the Affected Party's obligations under this Agreement despite the Force Majeure Event,

that obligation is suspended for so long as, and to the extent that, the Affected Party is affected by the Force Majeure Event.

(b) If the Recipient's obligation is relieved under this clause 17 of Part B, MPI will not be obliged to provide Funding for the Activities that have not been provided in accordance with this Agreement due to the Force Majeure Event.

#### 18. Insurance

For the term of this Agreement, and for at least three years after the term of this Agreement, the Recipient must effect and maintain adequate insurance to cover standard commercial risks, including professional indemnity, public liability and property damage (including fire) with sufficient coverage to cover its obligations under this Agreement. The Recipient must, upon the request of MPI, provide MPI with evidence of its compliance with this clause 18 of Part B.

# 19. Term and termination

#### 19.1 Term

This Agreement commences on the Start Date and continues until the End Date unless terminated earlier in accordance with its terms including under clause 6.2 of Part A.

#### 19.2 Termination for default

Either party may immediately terminate this Agreement, by written notice to the other party, if the other party breaches any material provision of this Agreement and the breach is not remedied within 20 Business Days of the receipt of written notice from the first party requiring the other to remedy the breach.

#### 19.3 Effects of expiry or termination

- (a) Termination of this Agreement is without prejudice to other rights and obligations of the parties accrued up to and including the date of termination.
- (b) On expiry or termination of this Agreement:
  - MPI may require the Recipient to provide evidence of how the Funding has been spent; and
  - any Funding that has not yet been distributed by MPI will not be distributed to the Recipient.

#### 19.4 Survival clauses

Clauses 6.8, 8.2, 8.3, 9, 10.1(e), 11, 12, 14, 15, 16, 17, 18, 19, 20 and 21 of Parl B any other clauses that expressly or by necessary implication survive this Agreement, continue after expiry or termination of this Agreement or part of it.

#### 20. Notices

- (a) Any notice or other communication under this Agreement will be deemed to be validly given if in writing and delivered by hand, registered mail, national post or international post, or email (subject to the remainder of this clause 20) to the MPI Programme Manager (for MPI) or Fund Manager (for the Recipient).
- (b) Any notice will be deemed to have been given on the date when actually delivered personally, by registered mail or following international posting, on the third Business Day following posting to a national address, and on the date that receipt of an emailed notice is acknowledged by the recipient personally (that is, not by any automatically generated email). However, if delivery occurs after 5:00 pm on a Business Day, it will be deemed delivered at 9:00 am on the next Business Day.
- (c) The parties agree that no notice required or permitted to be given under clause 15 (Dispute resolution) or clause 19 (Term and termination) may be given by email alone.

#### 21. General

- (a) Subject to clause 21(b) of this Part B, neither party will assign or transfer this Agreement or any rights or obligations under this Agreement without the prior written consent of the other party (such consent not to be unreasonably withheld).
- (b) MPI may assign, transfer or subcontract this Agreement or any rights or obligations under this Agreement to another public sector agency who has appropriation for the National Programme without the prior written consent of the Recipient.
- (c) No third party is entitled to the benefit of, or to enforce, this Agreement.
- (d) Each party will bear its own legal and other costs relating to the preparation and execution of this Agreement, unless agreed otherwise in writing.
- (e) This Agreement constitutes the entire agreement between the parties in connection with its subject matter and supersedes all previous agreements or understandings between the parties in connection with its subject matter.

- (I) The provisions of this Agreement are severable. If any one provision is determined to be judicially unenforceable, the remaining provisions will continue to be binding and enforceable.
- (g) This Agreement may be executed in any number of counterparts (including by executed, scanned or emailed copies) each of which will be deemed an original. Once each party has received a copy of the executed counterpart from the other party, each counterpart will be deemed to be as valid and binding on the party executing it as if it had been executed by all the parties.
- (h) This Agreement will be governed by, and construed in accordance with, the laws of New Zealand and the parties submit to the non-exclusive jurisdiction of the courts of New Zealand in relation to this Agreement.

PAga 32

# Execution

EXECUTED as an Agreement

Signed by The Sovereign in right of New Zealand acting by and through the Ministry of Primary Industries by:

41 rela

Signature

Penny Nelson

Name

Title

Deputy Director-General Biosecurity New Zealand

13 October 2020

Date

Signed for and on behalf of Otago Regional Council by:

Signature SARAH IARON Name

CHIEF EXECUTIVE

6. 11. 2020

Date

Page 33

Council Meeting 2021.04.14

# Schedule 1 – Definitions and Interpretation

# 1. Defined terms

#### 1.1 Defined terms

In this document, unless the context requires otherwise, the following terms have the following meanings:

Activities means the activities performed or to be performed (and any outputs provided or to be provided) by or on behalf of Recipient under this Agreement, including the activities and outputs described in clause 3 of Part A, and any other activities that are necessarily or customarily provided as part of those activities and/or outputs.

Agreement means this funding agreement, including Part A, Part B and this Schedule 1,

Agreement Manager means the person identified in clause 7 of Part A as a agreement manager for the relevant party, as that person may be replaced from time to time in accordance with this Agreement.

Background Material means any property, information, documentation or other material in any form:

- (a) in existence prior to the Start Date; or
- (b) developed outside the scope of this Agreement.

Business Day means any day, excluding Saturdays, Sundays and statutory public holidays in Wellington or Region(s) of the Recipient.

Start Date means the date this Agreement commences, as set out in clause 6.1 of Part A.

Confidential information means

- (a) in the case of MPI, MPI Data; and
- (b) in the case of both MPI and the Recipient, any other information obtained from the other party in the course of, or in connection with this Agreement that is marked confidential or ought to reasonably be considered to be confidential, regardless of value or importance.

End Date means the date this Agreement ends as set out in clause 6.1 of Part A, as may be extended in accordance with clause 6.1 of Part A.

Financial Year means the twelve months from 1 July to 30 June.

Force Majeure Event means, in relation to a party, an event or circumstance which is beyond the reasonable control of that party, but does not include any event or circumstance which could have been avoided, prevented or circumvented by that party taking reasonable steps, including ensuring business continuity by implementing an appropriate disaster recovery plan.

Fund Manager means the Recipient's fund manager, as notified by the Recipient to MPI and may be replaced from time to time.

Funded Activity has the meaning set out in clause 3.1(a) of Part A.

Funding means the amounts paid or payable to the Recipient by MPI under this Agreement.

Governance Group means the group which governs the National Programme, whose members are MPI (Chair), Land Information New Zealand (LINZ), New Zealand Defence Force (NZDF), Department of Conservation (DOC), and Local Government New Zealand (LGNZ).

GST means goods and services tax chargeable under the GST Act.

GST Act means the Goods and Services Tax Act 1985.

Health and Safety Incident means any event that falls under clause 3.1((k)(i) - (iv)) of Part A.

Health and Safety Management Plan means a written plan agreed between the Recipient and MPI for the management of risks and hazards relating to the Recipient's performance of the Activities. A plan may be documented on a template supplied by MPI, or on the Recipient's own

template. If not completed on MPI's template, then the plan must at a minimum include the following matters:

- the health and safety risk register, specific to the contracted work (including identified risks and mitigations, and assigned responsibilities for managing those risks);
- (b) confirmation of how those risks will be communicated to Workers;
- (c) clearly assigned health and safety roles & responsibilities for the Activities being performed (the name of the person responsible, rather than position titles);
- (d) details of regular meetings to discuss health and safety matters relating to the Activities being performed (this is a requirement for all high risk contracts);
- details of regular (min. monthly) reporting on Health and Safety Incidents (incl. Near Misses) and corrective actions;
- required personal protective equipment, and assigned responsibility for checking equipment;
- (g) any applicable standard operating procedures (step-by-step instructions on key tasks) in place and how Workers access these;
- the name of any and all contractors, subcontractors and if applicable volunteers engaged to assist with the work (and how they will be managed by the Recipient);
- (I) the health and safety induction process for Workers involved in the Agreement;
- identification of required training and/or certification to carry out the work (and confirmation of who checks training and/or certification has been completed);
- (k) emergency procedures, specific to the Activities;
- (I) details on how the Recipient's health and safety performance will be monitored; and
- (m) the process agreed with MPI for regular consultation, co-ordination and co-operation with MPI on health and safety matters, related to the Activities to be performed.

HSWA means the Health and Safety at Work Act 2015.

Incident means an unplanned event that results in injury but does not require time off work.

In Kind Contributions means funds, labour, or materials donated by other stakeholders or interested groups.

Intellectual Property Rights includes copyright, and all rights conferred under statute, common law or equity in relation to inventions (including patents), registered and unregistered trade marks, registered and unregistered designs, circuit layouts, Confidential Information, know-how, trade secrets, computer program codes and all other proprietary rights, whether registered or unregistered, and all equivalent rights and forms of protection anywhere in the world, together with all rights, interests or licences in or to any of the foregoing.

Management Unit or MU means an area designated by MPI as a management unit as set out in Wall-IS, or as otherwise notified by MPI, and has a fixed name, number and boundary (area).

Management Unit Operational Plan or MUOP means the management unit, operational plan for a Management Unit, which is prepared by or on behalf of the Recipient and endorsed in writing by the relevant RCG.

MPI Coordinator means the MPI coordinator, as notified by MPI to the Recipient and may be replaced from time to time.

MPI Data means:

- (a) any data owned, held, used or created by MPI, and provided to the Recipient:
- (b) any data processed by the Recipient as a direct result of this Agreement; and
- (c) all records, data and other information (other than software object code) provided or made available to the Recipient by MPI.

MPI Programme Manager means the MPI programme manager, as that person may be replaced from time to time in accordance with this Agreement.

MPI's Operations Lead means the MPI operations lead, as that person may be replaced from time to time in accordance with this Agreement.

National Programme means the national wallaby management programme, as described in Recital B of the Background.

Near Miss means an unplanned event that did not result in injury, illness or damage but had the potential to do so.

New Material means any property, information, documentation or other material in any form created:

- (a) by, for or on behalf of the Recipient;
- (b) on or following the Start Date; and
- (c) for the purpose of, or as a result of, performing its obligations under this Agreement.

Notifiable Event means a notifiable event within the meaning of section 25 of the HSWA, including but not limited to the following situations as a result of work:

- (a) a death;
- (b) a notifiable illness or injury; or
- (c) a Notifiable Incident.

Notifiable Incident means a notifiable incident within the meaning of section 24 of the HWSA.

Operational Area or OA means part of the Management Unit designated as an operational area by MPI, as set out in Wall-IS, or as otherwise notified by MPI, and has a fixed name, number and boundary (area).

Outcomes has the meaning in clause 1(a) of Part A.

Output means all or any output of the Activities provided or to be provided by or on behalf of the Recipient under this Agreement, including the outputs described in clause 3 of Part A (as amended from time to time in accordance with this Agreement).

Part A means Part A of this Agreement.

Part B means Part B of this Agreement.

PCBU means a person conducting a business or undertaking within the meaning of section 17 of the HWSA. For the purposes of this Agreement, both MPI and the Recipient are PCBUs and are required to consult, cooperate and coordinate their activities to meet their health and safety obligations to workers and others affected by the work in the provision of the Activities.

Personal Information has the meaning given to that term in section 2 of the Privacy Act 1993.

Personnel of any person, means all individuals directly or indirectly engaged by that person, including but not limited to directors, employees, contract staff, agents, consultants, specialists, support staff and co-opted or seconded staff and, in the case of the Recipient, includes its Subcontractors.

Privacy Laws means the Privacy Act 1993 (or any future replacement statue) and any other applicable legislation, principles, industry codes and policies.

Region(s) means the region(s) designated as a region(s) by MPI, as set out in Wall-IS or as otherwise notified by MPI, and has a fixed name, number and boundary (area).

Regional Coordination Group or RCG means the regional coordination group for the relevant Region(s), as described in row (b) of the table at clause 3.1 of Part A.

Serious Incident means any Health and Salety Incident that:

- (a) has the potential for causing a fatality, or long-term injury or illness; or
- (b) does not meet the 'Notifiable Event' threshold, but still results in a lost time injury or illness.

Significant Event has the meaning given to that term in clause 5 of Part B.

Start Date means the start date set out in clause 6.1 of Part A.

Subcontractor means:

- any person engaged by the Recipient to perform all or any part of the Activities or in respect of this Agreement; and
- (b) any subcontractor of a person described in (a) of this definition.

Tax Invoice means a tax invoice as defined in the Goods and Services Tax Act 1985.

Technical Liaison means the person identified in clause 7 of Part A as the technical liaison for the relevant party, as that person may be replaced from time to time in accordance with this Agreement.

Third Party Material means any property, information, documentation or other material in any form owned by a third party that is:

- (a) included, embodied in or attached to the New Material; or
- (b) used in undertaking the Activities.

Use means to use, copy, store, process, transfer, view, handle, modify, support and/or maintain.

Wall-IS means the 'Wallaby Information System' provided or made available by MPI.

Worker means an individual who carries out work for a PCBU within the meaning of section 19 of the HWSA.

# 2. Interpretation

In this Agreement, unless the context requires otherwise:

- the following order of precedence (in order of decreasing priority) applies to the documents forming part of, or incorporated into, this Agreement if any conflict or inconsistency arises:
  - (i) Part B;
  - (ii) Part A; and
  - (iii) the Schedules;
- (b) headings are for guidance only and do not affect interpretation;
- (c) the singular includes the plural and vice versa;
- (d) where a word or phrase is defined, its other grammatical forms have a corresponding
- meaning; (e) a reference to:
  - (i) a party to this Agreement includes that party's permitted assigns and successors;
  - (ii) \$ or dollars is a reference to New Zealand currency;
  - (iii) monetary amounts are the amounts stated before the addition of GST (if any);
  - (iv) Including and similar words do not imply any limitation;
  - (v) a person includes:
    - (A) a company, body of persons (corporate or unincorporated) or any state, regional or local government body or agency; and
    - (B) that person's representatives, successors and assigns;
  - (vi) a statute includes references to regulations, orders or notices made under or in connection with the statute or regulations, as it may be amended or replaced from time to time.

- a document includes references to that document as it may be amended or replaced from time to time;
- (viii) this Agreement includes all attachments to this Agreement and any document incorporated into this Agreement by reference;
- (ix) a clause is a reference to a clause in this Agreement;
- (x) a Part or Schedule, are references to a part or schedule of this Agreement; and
- (xi) a time means New Zealand standard time, as adjusted for daylights saving;
- (f) subject to clause 20 of Part B, anything that this Agreement requires to be done in writing, may be done by email; and
- (g) no rule of construction applies to the disadvantage of MPI on the basis that MPI put forward this Agreement or any part of it.

# Appendix 1 - Health and Safety obligations

#### 1. Definitions

1.1 In this Appendix, unless the context requires otherwise:

Business Hours means the hours between 9:00am and 5:00pm on a Business Day.

Control Measures has meaning set out in regulation 3 of the HSWA (GRWM) Regulations 2016,

HSWA (GRWM) Regulations 2016 means the Health and Safety at Work (General Risk and Workplace Management) Regulations 2016.

HSWA Legislation means the Health and Safety at Work Act 2015 and includes all regulations made under that Act (including but not limited to the Health and Safety at Work (General Risk and Workplace Management Regulations 2016), and any other health and safety-related legislation relevant to the Recipient's performance of the Activities and other obligations under this Agreement.

- 1.2 Other terms used but not defined in this Appendix have the same meaning as in the HSWA.
- 2. Compliance with health and safety legislation and directions
- 2.1 During the term of this Agreement, MPI and the Recipient will:
  - (a) consult, cooperate and coordinate activities with MPI to ensure that the parties comply with their respective obligations under HSWA Legislation as they relate to this Agreement, including attending regular meetings with MPI to discuss health and safety matters; and
  - (b) perform its, and ensure its Workers perform their, obligations under this Agreement in compliance with HSWA Legislation, including but not limited to the:
    - (i) duties of a PCBU under ss36-43 of HSWA;
    - duties relating to the identification of hazards and implementation of Control Measures under the HSW (GR&WM) Regulations; and
    - (iii) duties relating to the keeping of records under s57 of HSWA.
- 2.2 The Recipient will comply with all reasonable directions of MPI relating to health and safety as notified from time to time on any matters not addressed in MPI or the Recipient's health and safety policy and procedures, or in the Health and Safety Management Plan.
- 2.3 MPI acknowledges that any Worker engaged by the Recipient in the performance of the Activities has a statutory right to cease or refuse to carry out unsafe work, as that term is defined under section 82 of the HSWA.
- 2.4 During the term of this Agreement, MPI may consult with the Recipient on any concerns it has on any matters not addressed in the Recipient's health and safety policy and procedures. This includes where MPI considers there are new or increased risks or hazards or WorkSafe issues, applicable best practice guidelines or policies.

- 3. Health and safety policies, practices and plans
- 3.1 During the term of this Agreement, the Recipient will:
  - (a) maintain a general health and safety policy and practices that are appropriate to the nature of the Activities carried out by the Recipient;
  - (b) comply with its health and safety policy and practices, and ensure its Workers also comply.
  - (c) consider any concerns raised by MPI under paragraph 2.4 above, and make changes to its health and safety documentation and procedures as it deems appropriate.
- 3.2 Within 10 Business Days after commencing the provision of the Activities contracted under this Agreement, the Recipient will submit to MPI a draft written Health and Safety Management Plan which includes the information required in clause 1.1 of schedule 1 of the Agreement, identification of all health and safety risks arising from the contracted activities, and outlining how the Recipient proposes to mitigate and manage these risks. MPI will have the opportunity to review and comment on the draft plan, and offer suggestions to the Recipient.
- 3.3 Having duly considered MPI's comments, the Recipient will finalise the Health and Safety Management Plan within a further 10 Business Days (or such longer period as the parties may agree in writing). The parties must not unreasonably withhold their agreement to the Health and Safety Management Plan.
- 3.4 Where a Health and Safety Management Plan is in place, the Recipient must comply, and ensure its Workers comply, with the plan. The Recipient must also review and if necessary update the Health and Safety Management Plan at intervals no longer than every six (6) months, and
  - (a) following any serious health and safety incident; and/or
  - (b) at any the Recipient or MPI reasonably considers that new or increased risks or hazards have arisen in the provision of the Activities.
- 3.5 If the parties are unable to agree on the content of the Health and Safety Management Plan, then MPI in its sole discretion may:
  - suspend or terminate all or part of the Activities under this Agreement, either permanently or until such time as the Health and Safety Management Plan has been agreed between the parties; or
  - (b) refer the matter to the dispute resolution clauses under this Agreement.
- 3.6 During the term of this Agreement, MPI may request any changes to Health and Satety Management Plan that MPI considers are reasonably necessary to manage any new or increased risks or hazards. Where new or increased costs would be incurred by the requested change, the Recipient may seek an agreed variation to cover those costs. If the change to the plan cannot be agreed, then MPI may exercise any of the options set out in paragraph 3.5 above.
- Notifying events to WorkSafe NZ
- 4.1 During the term of this Agreement, the Recipient will ensure that all Notifiable Events occurring during delivery of the Activities are duly notified to:
  - WorkSale NZ in accordance with the requirements of section 56 of the HSWA; and
  - (b) MPI within one Business Day of becoming aware of the event.

- 4.2 Where the Recipient notifies WorkSafe NZ of a Notifiable Event, the Recipient must report the event to MPI within one Business Day of becoming aware of the event (in accordance with paragraph 4.1) and in addition promptly provide a written summary of that Notifiable Event and corrective actions identified to MPI.
- Regular health and safety reporting to MPI
- 5.1 During the term of this Agreement the Recipient will comply with the health and safety reporting requirements outlined Appendix B to this Agreement.
- 6. Non-compliance and corrective actions
- 6.1 If MPI becomes aware of a breach of HSWA Legislation or these health and safety clauses by the Recipient or its Personnel or Subcontractors in relation to the Activities, MPI may, in its reasonable discretion, require the Recipient to do any one or more of the following:
  - (a) remedy any breach of the HSWA Legislation or this Agreement;
  - (b) suspend or terminate all or part of the Activities under this Agreement, until such time as the breach has been remedied and/or the health and safety risk eliminated or managed (and during any such period of suspension MPI is not required to make payment to the Recipient for the Activities);
  - (c) if the breach relates to the acts or omissions of a Subcontractor, direct the Recipient to terminate or suspend the relevant subcontract;
  - (d) make an amendment to the Health and Safety Management Plan to manage health and safety risks; and/or
  - (e) refer the matter to dispute resolution under clause 15 of this Agreement.
- 6.2 In the event that the Recipient fails to remedy a breach or take any other action required by MPI under paragraph 6.1, MPI may terminate this Agreement under clause 19 of this Agreement. In this event, MPI's liability is limited to payment for Activities performed in accordance with this Agreement up to and including the date on which MPI gave notice of the action required to be taken.
  - 7. Subcontractors
  - 7.1 This paragraph 7 applies in addition to any other clauses in this Agreement relating to Subcontractors.
  - 7.2 MPI may, at any time during the term of this Agreement, require the Recipient to:
    - conduct a health and safety pre-qualification process for all, or any particular, Subcontractors the Recipient proposes to use to carry out the Activities; and/or
    - (b) submit a Health and Safety Management Plan, or amend an existing Health and Safety Management Plan, to address specific health and safety risks posed by the proposed use of the Subcontractor.
  - 7.3 The Recipient must ensure that prior to the commencement of the Activities by any Subcontractor, the Subcontractor has been advised of, and complies with:
    - (a) the Recipient's health and safety policy and procedures;
    - (b) the Health and Safety Management Plan relevant to the Activities to be carried out;
    - (c) any specific site-based health and safety issues in performing the Activities; and

Páge 41

- (d) any health and satety directions given by MPI to the Recipient under paragraph 2.2.
- 8. Records
- 8.1 The Recipient must keep all records required under the HSWA (and promptly provide them on request to MPI).
- 8.2 In addition to any records required under HSWA, the Recipient must keep records (and provide them on request to MPI) of:
  - any Health and Safety Incidents which have been investigated internally by the Recipient, or by WorkSafe NZ, relating to the Activities (and the outcome of such investigations); and
  - (b) health and safety due diligence processes, and any associated documents, relating to the health and safety compliance and performance of Subcontractors used for the Activities.

# 9. Health and safety audit and inspection

- 9.1 MPI may, at any time during the term of this Agreement:
  - (a) require the Recipient to provide information or documentation in relation to any matter concerning the Recipient's health and safety performance or compliance in relation to the Activities, or relating to a Health and Safety Incident or risk; or
  - (b) carry out a paper-based audit of the Recipient's health and safety system as it relates to the Activities.
- 9.2 At any reasonable time during Business Hours, MPI may carry out a physical inspection of any place of work that the Recipient is using or intends to use in connection with the provision of the Activities to MPI.

# Appendix 2 – Regular health and safety reporting

The Recipient health and safety reporting is to fall under the following three categories:

#### 1. Notifiable Events

As set out in paragraph 4.1 (b) of Appendix A, Notifiable Events must be reported to MPI within one Business Day of the Recipient becoming aware of the event. To clarify, under paragraph 4.2 of Appendix A where WorkSafe has been notified, the Recipient must also promptly provide a written summary of that Notifiable Event and corrective actions to MPI.

#### Contemporaneous reporting of Serious Incidents that do not meet the 'Notifiable Event' threshold

The Recipient must report any Serious Incident to MPI as soon as practicable.

MPI considers a "Serious Incident" to be any Health and Safety Incident relating to the Activities provided under this Agreement that:

- a) has a potential for causing a tatality, or a long-term injury or illness; or
- b) does not meet the 'Notifiable Event' threshold, but still results in a lost time injury or illness.

Serious incidents must be reported contemporaneously to MPI by inputting the relevant information into Wall-IS in accordance with any Wall-IS User Guide. If that inputting functionality is not available on Wall-IS, the Recipient will provide the results to MPI in the form and format that MPI reasonably requests.

#### 3. Monthly summary of all incidents

In addition to other reporting requirements, the Recipient will deliver a written monthly report to MPI summarising all Health and Safety Incidents relating to the Activities carried out under this Agreement. The monthly report will include:

- a) the identity of the project, response, long-term programme or other piece of work the incident related to. Where this is not possible the incidents will be classified generally (i.e. general laboratory incidents);
- b) corrective actions identified for each Serious Incident and whether these have been implemented – including any corrective actions taken as a result of any MPI or WorkSate NZ audit or investigation, or any internal investigation resulting from a Notifiable Event;
- c) any instances it becomes aware of where its Workers, in carrying out the Activities, have not complied with the requirements of this Agreement or HSWA in relation to health and safety, or have not complied with MPI's health and safety policy or practices, or those of the Recipient;
- d) current status of any injured Workers; and
- e) any damaged property or environmental damage or pollution to the extent that it relates to a health or safety risk of the Activities.

The monthly report will be compiled from information entered into Wall-IS, if any additional information is required to satisfy the information requirements set out in 3. a)-e), that information will be delivered to your MPI Technical Lialson/Contract Manager by the 5th Business Day of each month during the term of this Agreement.

# 7.2. SoE Water Quality Report

| Prepared for: | Council   |
|---------------|---|
| Report No.    | SPS2116   |
| Activity:     | Environmental: Water Quality SOE – Rivers and Lakes             |
| Author:       | Rachel Ozanne, Scientist  |
| Endorsed by:  | Anita Dawe, Acting General Manager Strategy, Policy and Science |
| Date:         | 1 April 2021  |

# PURPOSE

- [1] This paper presents a report on the state (2015-2020) and trends (2000-2020) of lake and river water quality in the Otago Region. The report also identifies the baseline state (2012-2017) as required by the National Policy Statement for Freshwater Management 2020 (NPS-FM 2020). The full report is attached in the Appendix.
- [2] Understanding the current state of water quality is a key component informing the Land and Water Regional Plan (LWRP) as well as upcoming action plans.

# **EXECUTIVE SUMMARY**

- [3] This study analysed the available water quality data in the Otago Region. The state of water quality in the region is reported, on a site-by-site basis, relative to targets specified in the National Objectives Framework (NOF) of the National Policy Statement Freshwater Management (NPSFM, 2020). In addition, the study assessed water quality trends site by site, and across Otago as a whole. ORC engaged Land Water People (LWP) to evaluate water quality state (LWP, 2020a and undertake trend analysis (LWP, 2020b).
- [4] State analysis was based on water quality samples collected over a five-year period from 1 July 2015 to 30 June 2020 and compared to the five-year period 1 July 2012 to 30 June 2017, which is defined as the baseline state (NPSFM, 2020).
- [5] Water quality analysed represented 10 physico-chemical and microbiological variables and biological indicators, for 124 monitoring sites in the region. Sites included ORC monitored river sites (110), NIWA monitored National River Water Quality Network (NRWQN) sites (5) and ORC monitored lake sites (9 lakes, 22 sites/depths). All variables were evaluated for state and trends at all sites (when sufficient data was available). This report describes river and lake state and trends for the variables that specifically relate to the NPSFM (2020); chlorophyll-a, total nitrogen, total phosphorus, ammoniacalnitrogen, nitrate, suspended fine sediment, macroinvertebrate community index (MCI), macroinvertebrate average score per metric (ASPM), dissolved reactive phosphorus and E. coli.
- [6] Sites were graded as a NOF Band (A, B, C, D, and Band E for E. coli) for each variable based on a comparison of the assessed state with the relevant criteria.
- [7] Trend analysis was carried out for 10-year and 20-year periods ending on 1 September 2020 for all site and water quality variable combinations that met a minimum

Council Meeting 2021.04.14

requirement for numbers of observations. The methods used for statistical trend analyses were Kendall's test of rank correlation and the Sen slope estimator (SSE), which have both been used for trend analysis of water quality for several decades (Hirsch et al., 1982). LWP (2020b) considered flow adjustment as part of the trend assessment.

- [8] Individual site trend estimates were aggregated, to provide an overall picture of trends for the region. This was done graphically using stacked bar charts showing proportions of sites for each variable that fall into different trend direction confidence categories. For the 10-year trend period the predominant trend direction was variable by water quality analyte, but 20-year trends were predominantly degrading for all variables apart from ammoniacal nitrogen.
- [9] The most obvious pattern associated with the assessment of water quality state was that almost all sites passed the NOF criteria for ammoniacal-N toxicity and nitrate toxicity. There are obvious spatial patterns associated with the variation in grades, and water quality is best at river and stream reaches located at high or mountainous elevations under predominantly native flora cover. These sites tend to be associated with the upper catchments of larger rivers (e.g. Clutha River/Matau-Au) and the outlets from large lakes (i.e. Hawea, Wakatipu and Wanaka). Water quality is generally poorer at sites located on smaller, low-elevation streams that drain pastoral or urban catchments.
- [10] There has previously been a lack of detailed information held by ORC on local or catchment scale land use change or land management practice changes. Land use activity and land resources have not been monitored by ORC, thus significantly restricting analyses that could be performed to investigate the effect of land use activity on water quality. ORC's Science team is addressing this data deficiency and is developing a land science programme to enable ORC to comment on drivers of water quality trends across Otago in coming years. This is also being addressed by requirements in the NPSFM (2020), which requires freshwater to be managed in an integrated way and considers the effects of the use and development of land on a whole-of-catchment basis, including the effects on receiving environments. Part of that response includes managing land in the proposed LWRP.

# RECOMMENDATION

That the Council:

1) **Receives** this report.

# BACKGROUND

- [11] Otago Regional Council (ORC) operates a State of Environment (SoE) water quality monitoring network in lakes and rivers throughout the region for monitoring the state and trends in water quality and reporting on policy effectiveness. Prior to mid-2018, there were fewer monitoring sites in the region. Following a review (NIWA, 2017), a more extensive monitoring programme commenced in mid-2018 to better represent environmental classes in the Otago region, based largely on the River Environment Classification (REC).
- [12] This study analysed the available water quality data for rivers in the Otago Region. The state of water quality in the region is reported, on a site-by-site basis, relative the

Council Meeting 2021.04.14

attribute states specified in the NOF of the NPSFM (2020). In addition, the study assessed water quality trends site by site and across the Otago region.

- [13] For each numeric attribute, the NOF defines categorical numeric attribute states as four (or five) attribute bands, which are designated A to D (or A to E, in the case of the *E. coli* attribute). The attribute bands represent a graduated range to support achievement of environmental outcomes, from high (A band) to low (D or E band). The ranges for numeric attribute states that define each attribute band are defined in Appendix 2 of the NPSFM (2020). For most attributes, the D band represents a condition that is unacceptable (with the threshold between the C and the D band being referred to as 'bottom line' in any waterbody nationally.
- [14] The trend direction and the confidence in the trend direction were evaluated using either the Mann Kendall assessment or the Seasonal Kendall assessment. The method used to assess trend rate is based on non-parametric Sen slope regressions of water quality observations against time. The trend for each site/variable combination was assigned a categorical level of confidence that the trend was improving according to its evaluated confidence. Outputs from the trend analyses were also classified into four direction categories: improving, degrading, indeterminate, and not analysed.
- [15] Water quality state and trend results are organised in FMU/Rohe chapters, giving a detailed assessment of state and trends for water quality in that region. A summary of Otago water quality is also given.

# RESULTS

[16] Figure 1 presents results of river state analysis and Figure 2 presents results of lake state analysis. The maps show Otago river sites coloured according to their state grading based on NOF attribute bands.

Council Meeting 2021.04.14



Figure 1 Water quality river state results (2015-2020) coloured according to their state grading as indicated by NOF attribute bands.



Figure 2 Water quality lake state results (2015-2020) coloured according to their state grading as indicated by NOF attribute bands.

- [17] The plots show spatial patterns associated with the variation in grades. Water quality is best at river and stream reaches located at high or mountainous elevations under predominantly native flora cover. These sites tend to be associated with the upper catchments of larger rivers (e.g. Clutha River/Matau-Au) and the outlets from large lakes (e.g. Hawea, Wakatipu and Wanaka). Water quality is generally poorer at sites located on smaller, low-elevation streams that drain pastoral or urban catchments.
- [18] Figure 3 gives results of trend analysis across Otago. Colour coded bar charts represent the proportions of sites with trends indicating improving and degrading water quality for the 10- and 20-year trend periods, respectively. Green colours indicate sites with improving trends, and red-orange colours indicate sites with degrading trends. Trends used in the graphs are not flow-adjusted and have a hi-censor filter applied.

Council Meeting 2021.04.14

[19] These plots show that 20-year trends were predominantly degrading for all variables apart from ammoniacal nitrogen and for the 10-year trend period the predominant trend direction was variable by water quality variable.



*Figure 3 Summary plots representing the proportion of river sites with improving 10-year and 20-year time period trends at each categorical level of confidence.* 

[20] Figure 4 presents results of lake trend analysis across Otago. The lake sites are classified according to confidence that their 18-year water quality trend direction indicated improving water quality. Green colours indicate sites with improving trends, and red-orange colours indicate sites with degrading trends for each of the attributes measured.

Council Meeting 2021.04.14


*Figure 4 Map of lake sites classified by confidence that their 18-year water quality trend direction indicated improving water quality.* 

# ISSUE

[21] Some NPSFM 2020 Appendix 2A attributes are not yet monitored (i.e. dissolved oxygen below point-source) and some attributes have a limited dataset (i.e., planktonic cyanobacteria). Of the attributes in Appendix 2B of the NPSFM, monitoring has commenced for submerged plants (native and invasive) with the first surveys undertaken in 2020. Monitoring is also underway for deposited fine sediment, dissolved oxygen and ecosystem metabolism, but datasets are limited.

Council Meeting 2021.04.14

[22] The SoE network was set up in 2018 to be representative of the river network in Otago (based on REC). In 2023 it is intended to revise the network to ensure that the water quality network is representative of each FMU.

# DISCUSSION

- [23] In nearly all cases, monitoring sites identified as degraded in previous reports (ORC, 2012, ORC, 2017) or from targeted catchment studies, remain degraded. There is very little change in the pattern of water quality throughout Otago. The lack of detailed information on local or catchment scale land use change or land management practice changes severely limits ability to comment on drivers of trends evident in the data set. To better interpret the reasons for improvements or degradation in water quality, information on land type, management and use are required. Science's recently formed Land Team are building knowledge in this area.
- [24] In relation to state compliance with the NPSFM (2020) most lakes return an 'A' band for all attribute states; except for Lake Tuakitoto, Lake Hayes and Lake Waihola. For rivers, a clear pattern of good water quality across all attributes is associated with the Upper Lakes Rohe and to some extent the Dunstan Rohe. Water quality is generally poorer at sites located on smaller, low-elevation streams that drain agricultural or urban catchments, including Dunedin Coast Rohe and North Otago Rohe.
- [25] The historic data series for the SoE lake monitoring sites, particularly for Lakes Wanaka, Wakatipu, Hawea and Lake Dunstan, have high numbers of results at less than the detection limit (censored values) reflecting nutrient and algal concentrations below those able to be measured by the laboratory. These sites may have a reasonably highlevel of confidence in trend direction, however the rate of the trend cannot be quantified.
- [26] A policy is being developed on how ORC will make decisions on its non-regulatory interventions for freshwater, which includes Action Plans. The implementation of that policy and the FMU process will address Otago's degraded freshwater resources. The policy will include an approach to prioritise degraded sites. Sites that are degraded or degrading will be considered in the prioritisation process which will result in a programme of work.

# OPTIONS

[27] Not applicable

# CONSIDERATIONS

# **Policy Considerations**

[28] Issues with the current planning approach for land use and freshwater management will be addressed, as far as practicable, through the new LWRP or non-regulatory action plans. For example, land use controls which have not been utilised beyond beds of lakes and rivers, will now form part of the regulatory approach where appropriate.

# **Financial Considerations**

[29] In the future, further investment is required to ensure the SoE Water Quality monitoring network is compliant with national direction and is representative of FMUs. This work

Council Meeting 2021.04.14

will be completed as scheduled in the proposed long-term plan for the 2023/2024 financial year.

# Significance and Engagement

[30] Not applicable.

# **Legislative Considerations**

[31] Monitoring networks must comply with national legislation and effectively evaluate objectives in regional plans. However, as policies can change rapidly, there is generally a lag for network changes to be implemented and then further delay until sufficient data is collected to enable analysis.

# **Risk Considerations**

- [32] Providing this information assists ministries, agencies, regulators, decision makers, resource users and community groups, to better understand and manage the risks associated with degraded and degrading water quality.
- [33] Continued water quality degradation is a significant risk for Otago impacting on Te Mana o Te wai and community resilience. This can be mitigated by ensuring key stakeholders are aware and take ownership in the problem, assisting land managers and ensuring access to tools and knowledge to adopt best management practices. ORC's action plan policy will greatly assist with mitigating risk of continued degradation. However, until this is implemented and supported with adequate resourcing, continued degradation is likely.
- [34] The NPS-FM (2020) requires freshwater is managed in an integrated way, considering the effects of use and development of land on a whole-of-catchment basis, including effects on receiving environments. This will help in understanding drivers of trends evident across Otago.

# **NEXT STEPS**

- [35] A SoE report is a statutory requirement which is undertaken every five years and provides a detailed review of water reporting on regional state and trends in water quality and performance against the NPS-FM. The next report will cover the period up to June 2022 to align with notification of the proposed LWRP. Water quality will be updated and included in this report.
- [36] The communications team will provide a high-level communications plan to support the SOE water quality report. The key message within the communications plan is to clearly define the purpose of the SOE report which is to support ORC's FMU community consultations.

# ATTACHMENTS

1. Attachment SoE Water Quality 2015 to 2020 [7.2.1 - 155 pages]

Council Meeting 2021.04.14

2000-2020

# State and Trends of River and Lake Water Quality in the Otago Region 2000-2020

Page 1 | 155

2000-2020

Otago Regional Council Private Bag 1954, Dunedin 9054 70 Stafford Street, Dunedin 9016 Phone 03 474 0827 www.orc.govt.nz

© Copyright for this publication is held by the Otago Regional Council. This publication may be reproduced in whole or in part, provided the source is fully and clearly acknowledged. ISBN Report writer: Rachel Ozanne, Freshwater Science Team, Otago Regional Council. Data and statistical analyses performed by: LandwaterpeopleLtd (LWP Ltd) Reviewed by: LWP Ltd Published

Page 2 | 155

2000-2020

Page 3 | 155

2000-2020

# Contents

| E> | Executive Summary15              |       |   |  |  |
|----|----------------------------------|-------|---|--|--|
| 1  | Introduction17                   |       |   |  |  |
| 2  | 2 Otago Region                   |       |   |  |  |
|    | 2.1                              | Reg   | ion Description                           |  |  |
|    | 2.2                              | Free  | shwater management units                  |  |  |
|    | 2.3                              | Upp   | er Lakes Rohe (Clutha Matu/Au FMU)20      |  |  |
|    | 2.4                              | Dur   | stan Rohe (Clutha Matu/Au FMU)20          |  |  |
|    | 2.5                              | Mai   | nuherekia Rohe (Clutha Matu/Au FMU)20     |  |  |
|    | 2.6                              | Rox   | burgh Rohe (Clutha Matu/Au FMU)21         |  |  |
|    | 2.7                              | Low   | er Clutha Rohe (Clutha Matu/Au FMU)21     |  |  |
|    | 2.8                              | Taie  | ri FMU21                                  |  |  |
|    | 2.9                              | Dur   | edin Coast FMU22                          |  |  |
|    | 2.10 North Otago FMU             |       | th Otago FMU23                            |  |  |
|    | 2.11                             | Catl  | ins FMU23                                 |  |  |
| 3  | OF                               | RC mo | nitoring programme                        |  |  |
|    | 3.1                              | Wat   | ter Quality Sites                         |  |  |
|    | 3.2                              | Wat   | ter quality variables25                   |  |  |
|    | 3.2                              | 2.1   | Phytoplankton, Periphyton and Nutrients25 |  |  |
|    | 3.2                              | 2.2   | Toxicants                                 |  |  |
|    | 3.2                              | 2.3   | Suspended sediment                        |  |  |
|    | 3.2                              | 2.4   | Aquatic Life27                            |  |  |
|    | 3.2                              | 2.5   | Escherichia coli (E. coli)                |  |  |
| 4  | M                                | ethod | s29                                       |  |  |
|    | 4.1 Water Quality State Analysis |       | ter Quality State Analysis                |  |  |
|    | 4.1                              | 1.1   | Grading of monitoring sites               |  |  |
|    | 4.1                              | 1.2   | Handling censored values                  |  |  |
|    | 4.1                              | 1.3   | Time-period for assessments               |  |  |
|    | 4.1                              | 1.4   | Calculation of water clarity              |  |  |
|    | 4.1                              | 1.5   | pH Adjustment of Ammonia33                |  |  |
|    | 4.1                              | 1.6   | Evaluation of compliance statistics       |  |  |
|    | 4.2                              | Wat   | ter Quality Trend Analysis                |  |  |
|    | 4.2                              | 2.1   | River water quality data35                |  |  |

Page 4 | 155

## 2000-2020

|   | 4.2. | 2      | Lake quality data                                      |              |
|---|------|--------|--|--------------|
|   | 4.2. | 3      | Flow data  |              |
|   | 4.2. | 4      | Sampling dates, seasons, and time-periods for analyses |              |
|   | 4.2. | 5      | Handling censored values                               |              |
|   | 4.2. | 6      | Flow adjustment  |              |
|   | 4.2. | 7      | Seasonality assessment                                 |              |
|   | 4.2. | 8      | Analysis of trends                                     | 40           |
|   | 4.2. | 9      | Trend direction assessment                             | 40           |
|   | 4.2. | 10     | Assessment of trend rate                               | 41           |
|   | 4.2. | 11     | Evaluating changes in discontinuous data               | 42           |
|   | 4.2. | 12     | Interpretation of trends                               | 43           |
|   | 4.2. | 13     | River data availability                                | 45           |
| 5 | Res  | ults ( | Dverview   | 46           |
| 6 | Upp  | oer La | akes Rohe (Clutha Matu/Au FMU)                         | 47           |
|   | 6.1  | Stat   | e Analysis Results                                     | 48           |
|   | 6.1. | 1      | Periphyton and Nutrients                               | 51           |
|   | Тох  | icant  | s  | 51           |
|   | 6.1. | 2      | Suspended fine sediment (Rivers)                       | 51           |
|   | 6.1. | 3      | Aquatic Life (Rivers)                                  | 52           |
|   | 6.1. | 4      | Human health for recreation                            | 52           |
|   | 6.2  | Trer   | nd Analysis  | 53           |
|   | 6.3  | Wat    | er quality summary and discussion: Upper lakes Rohe    | 54           |
| 7 | Dur  | nstan  | Rohe (Clutha Matu/Au FMU)                              | 57           |
|   | 7.1  | Stat   | e Analysis Results                                     |              |
|   | 7.1. | 1      | Periphyton and Nutrients                               | 61           |
|   | 7.1. | 2      | Toxicants  | 62           |
|   | 7.1. | 3      | Suspended fine sediment (Rivers)                       | 62           |
|   | 7.1. | 4      | Aquatic Life (Rivers)                                  | 62           |
|   | 7.1. | 5      | Human health for recreation (Rivers)                   | 62           |
|   | 7.2  | Trer   | nd Analysis  | 63           |
|   | 7.3  | Wat    | er quality summary and discussion: Dunstan Rohe        | 64           |
| 8 | Ma   | nuhe   | rekia Rohe (Clutha Matu/Au FMU)                        | 69           |
|   | 8.1  | Stat   | e Analysis Results                                     | 70           |
|   | Per  | iphyt  | on and Nutrients                                       | 72           |
|   | 8.1. | .1     | Toxicants  | 72           |
|   |      |        |  | Page 5   155 |

## 2000-2020

| 8.1.2 |          | Suspended fine sediment (Rivers)                      | 72           |
|-------|----------|---|--------------|
| 8.1.3 |          | Aquatic Life (Rivers)                                 | 73           |
| 8.1.4 |          | Human health for recreation (Rivers)                  | 73           |
| 8.2   | Trei     | nd Analysis   | 74           |
| 8.3   | Wa       | ter quality summary and discussion: Manuherekia Rohe  | 75           |
| 9 R   | Roxburg  | h Rohe (Clutha Matu/Au FMU)                           | 78           |
| 9.1   | Stat     | e Analysis Results                                    | 79           |
| 9     | 9.1.1    | Periphyton and Nutrients                              | 81           |
| 9     | 9.1.2    | Toxicants (Rivers)                                    | 81           |
| 9     | 9.1.3    | Suspended fine sediment                               | 81           |
| 9     | 9.1.4    | Human health for recreation                           | 81           |
| 9.2   | Trei     | nd Analysis   | 82           |
| 9.3   | Wa       | ter quality summary Roxburgh Rohe                     | 83           |
| 10 L  | ower C   | lutha Rohe (Clutha Matu/Au FMU)                       | 85           |
| 10.1  | 1 Stat   | e Analysis Results                                    | 86           |
| 1     | .0.1.1   | Periphyton and Nutrients                              | 88           |
| 1     | .0.1.2   | Toxicants   | 88           |
| 1     | .0.1.3   | Suspended fine sediment                               | 89           |
| 1     | .0.1.4   | Aquatic Life (Rivers)                                 | 89           |
| 1     | .0.1.5   | Human health for recreation (Rivers)                  | 89           |
| 10.2  | 2 Trei   | nd Analysis   | 90           |
| 10.3  | 3 Wa     | ter quality summary and discussion: Lower Clutha Rohe | 92           |
| 11 T  | aieri FN | ЛU  | 97           |
| 11.1  | 1 Stat   | e Analysis Results                                    |              |
| 1     | 1.1.1    | Periphyton and Nutrients                              |              |
| 1     | 1.1.2    | Toxicants (Rivers)                                    |              |
| 1     | 1.1.3    | Suspended fine sediment                               |              |
| 1     | 1.1.4    | Aquatic Life (Rivers)                                 |              |
| 1     | 1.1.5    | Human health for recreation (Rivers)                  |              |
| 11.2  | 2 Trei   | nd Analysis   |              |
| 11.3  | 3 Wa     | ter quality summary Taieri FMU                        |              |
| 12 D  | Dunedin  | Coast FMU   | 110          |
| 12.1  | 1 Stat   | e Analysis Results                                    | 111          |
| 1     | .2.1.1   | Periphyton and Nutrients                              | 113          |
| 1     | 2.1.2    | Toxicants (Rivers)                                    |              |
|       |          |   | Page 6   155 |

## 2000-2020

| 12.1.3  |            | Suspended fine sediment (Rivers)       | 113 |  |
|---------|------------|--|-----|--|
| 12.1.4  |            | Aquatic Life (Rivers)                  |     |  |
| 12.1.5  |            | Human health for recreation (Rivers)   | 114 |  |
| 12.2    | Tre        | end Analysis: Rivers                   | 115 |  |
| 12.3    | Wa         | ater quality summary Dunedin Coast FMU | 116 |  |
| 13 No   | rth O      | Dtago FMU                              | 118 |  |
| 13.1    | Stat       | ate Analysis Results                   | 119 |  |
| 13      | 1.1        | Periphyton and Nutrients               | 121 |  |
| 13      | 1.2        | Toxicants (Rivers)                     | 121 |  |
| 13      | 1.3        | Suspended fine sediment (Rivers)       | 121 |  |
| 13      | 1.4        | Aquatic Life (Rivers)                  | 121 |  |
| 13      | 1.5        | Human health for recreation            | 122 |  |
| 13.2    | Tre        | end Analysis: Rivers                   | 122 |  |
| 13.3    | Wa         | ater quality summary North Otago FMU   | 125 |  |
| 14 Ca   | tlins F    | FMU                                    | 129 |  |
| 14.1    | Stat       | ate Analysis Results                   | 130 |  |
| 14      | 1.1        | Periphyton and Nutrients               | 132 |  |
| 14      | 1.2        | Toxicants (Rivers)                     | 132 |  |
| 14      | 1.3        | Suspended fine sediment (Rivers)       | 132 |  |
| 14      | 1.4        | Aquatic Life (Rivers)                  | 132 |  |
| 14      | 1.5        | Health for recreation (Rivers)         | 133 |  |
| 14.2    | Tre        | end Analysis Results – Rivers          | 134 |  |
| 14.3    | Wa         | ater quality summary Catlins FMU       | 134 |  |
| 15 Ota  | ago Si     | Summary                                | 137 |  |
| 15.1    | Ota        | ago Overview: Water Quality State      | 137 |  |
| 15.2    | Ota        | ago Overview: Water Quality Trends     | 140 |  |
| Referen | References |  |     |  |

Page 7 | 155

## 2000-2020

# List of Figures

| Figure 1 Map showing the FMU and Rohe boundaries, State of Environment monitoring site locations are also shown   |
|---|
| Figure 2.The Kendall S statistic is calculated by first evaluating the difference between all pairs of water quality observations (Figure 1, A and B)41   |
| Figure 3 Pictogram of the calculation of the Sen slope, which is used to characterise trend rate.<br>   |
| Figure 4 Location of water quality monitoring sites in the Upper Lakes Rohe47   |
| Figure 5 Grading of the river sites of the Upper Clutha Rohe based on the NOF criteria48  |
| Figure 6 Grading of the lake sites of the Upper Clutha Rohe based on the NOF criteria   |
| Figure 7 Maps showing Upper Clutha Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines   |
| Figure 8 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement   |
| Figure 9 Location of water quality monitoring sites in the Dunstan Rohe   |
| Figure 10 Grading of the river sites of the Dunstan Rohe based on the NOF criteria58  |
| Figure 11 Grading of the lake sites of the Dunstan Rohe based on the NOF criteria59   |
| Figure 12 Maps showing Dunstan Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines   |
| Figure 13 Summary of Dunstan Rohe sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement  |
| Figure 14 Location of water quality monitoring sites in the Manuherekia Rohe69  |
| Figure 15 Grading of the river sites of the Manuherekia Rohe based on the NOF criteria.<br>Grades for sites that did not meet the sample number requirements in Table 1 are shown as<br>white cells with coloured circles. The white cells indicate sites for which the variable was not<br>monitored. Small square in the upper left quadrant of the cells indicate the site grade for the<br>baseline |
| Figure 16 Maps showing Manuherekia Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines71   |
| Figure 17 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement  |
| Figure 18 Dunstan Creek, E.coli is 'exceptionally unlikely' to be improving (over 10-years)75   |

Page 8 | 155

## 2000-2020

| Figure 19 Location of water quality monitoring sites in the Roxburgh Rohe78   |
|---|
| Figure 20 Grading of the river and lake sites in the Roxburgh based on the NOF criteria79   |
| Figure 21 Maps showing Roxburgh Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines              |
| Figure 22 Summary of Roxburgh Rohe sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement   |
| Figure 23 Location of water quality monitoring sites in the Lower Clutha Rohe85   |
| Figure 24 Grading of River and Lake sites in the Lower Clutha Rohe, based on the NOF criteria.<br>  |
| Figure 25 Maps showing Lower Clutha Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines          |
| Figure 26 Summary of Lower Clutha Rohe sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement90   |
| Figure 27 Summary of Lake Tuakitoto trends, categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement91  |
| Figure 28 Heriot Burn. E.coli is 'virtually certain' to be improving (over 10-years)92  |
| Figure 29 Location of water quality monitoring sites in the Taieri FMU97  |
| Figure 30 Grading of the river sites of the Taieri FMU based on the NOF criteria  |
| Figure 31 Grading of the lake sites of the Taieri FMU based on the NOF criteria   |
| Figure 32 Maps showing Taieri FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines                 |
| Figure 33 Summary of Taieri FMU river sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement103   |
| Figure 34 Summary of Taieri FMU lake sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement104  |
| Figure 35 Summary of river state, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve a band 'D' |
| Figure 36 Summary of river sites where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in Table 2   |
| Figure 37 Summary lake sites where trends (continuous data) are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in Table 2108  |

Page 9 | 155

## 2000-2020

| Figure 38 Overall summary of trends for the Taieri FMU using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in Table 2   |  |  |  |  |  |
|--|--|--|--|--|--|
| Figure 39 Location of water quality monitoring sites in the Dunedin Coast FMU110   |  |  |  |  |  |
| Figure 40 Grading of the river sites of the Dunedin Coast FMU based on the NOF criteria11  |  |  |  |  |  |
| Figure 41 Maps showing Dunedin Coast FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines |  |  |  |  |  |
| Figure 42 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement   |  |  |  |  |  |
| Figure 43 Location of water quality monitoring sites in the North Otago FMU118   |  |  |  |  |  |
| Figure 44 Grading of the river sites of the North Otago FMU based on the NOF criteria119   |  |  |  |  |  |
| Figure 45 Maps showing North Otago FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines   |  |  |  |  |  |
| Figure 46 Summary of North Otago FMU sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement  |  |  |  |  |  |
| Figure 47 Waianakarua River, 10-year trend in Nitrite/Nitrate Nitrogen   |  |  |  |  |  |
| Figure 48 Location of water quality monitoring sites in the Catlins FMU129   |  |  |  |  |  |
| Figure 49 Grading of the river sites of the Catlins FMU based on the NOF criteria130   |  |  |  |  |  |
| Figure 50 Maps showing Catlins FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines       |  |  |  |  |  |
| Figure 51 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement   |  |  |  |  |  |
| Figure 53 Summary plot representing the proportion of river sites with improving 10-year time-period trends at each categorical level of confidence140   |  |  |  |  |  |
| Figure 54 Summary plot representing the proportion of river sites with improving 20-year time-period trends at each categorical level of confidence14  |  |  |  |  |  |
| Figure 55 Map of river sites classified by confidence that their 10-year raw water quality trend direction indicated improving water quality   |  |  |  |  |  |
| Figure 56 Map of river sites classified by confidence that their 20-year raw water quality trend direction indicated improving water quality   |  |  |  |  |  |
| Figure 57 Map of lake sites classified by confidence that their 10-year water quality trend direction indicated improving water quality  |  |  |  |  |  |
|  |  |  |  |  |  |

Page 10 | 155

## 2000-2020

| Figure 58 Map of lake sites classified by confidence that their 18-year water quality trend |     |
|---|-----|
| direction indicated improving water quality. Confidence is expressed categorically based o  | on  |
| the levels defined in Table 2. LWP (2020b)  | 145 |

Page 11 | 155

## 2000-2020

# List of Tables

| Table 1 River water quality variables included in this report, including NPS-FM reference and water body type  |
|--|
| Table 2. Level of confidence categories used to convey the confidence that the trend (or stepchange) indicated improving water quality   |
| Table 3 River water quality variables, measurement units and site numbers for which 10- and20-year trends (Raw, and Flow Adjusted FA) were analysed by this study.45   |
| Table 4 Summary of river state, where state does not meet the national bottom line cells arecoloured red. No lake variables were below the national bottom line  |
| Table 5 Summary of river sites (raw data) where trends are greater than 'likely' or 'unlikely'.Confidence is expressed categorically based on the levels defined in Table 2  |
| Table 6 Summary lake sites where trends (continuous data) are greater than 'likely' or'unlikely'. Confidence is expressed categorically based on the levels defined in Table 2   |
| Table 7 Summary of trends using raw data for rivers and continuous data for lakes. Confidenceis expressed categorically based on the levels defined in Table 2   |
| Table 8 Summary of river and lake state, red cells show where state does not meet thenational bottom line in one or more variable  |
| Table 9 Summary of river sites where trends are greater than 'likely' or 'unlikely' (raw data).Confidence is expressed categorically based on the levels defined in Table 2  |
| Table 10 Summary lake sites where trends (continuous data) are greater than 'likely' or'unlikely'. Confidence is expressed categorically based on the levels defined in Table 2  |
| Table 11 Overall summary of trends for the Dunstan Rohe using raw data for rivers andcontinuous data for lakes. Confidence is expressed categorically based on the levels defined inTable 2  |
| Table 12 Summary of river, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve a band 'D' |
| Table 13 Summary of river sites where trends are greater than 'likely' or 'unlikely'. Confidenceis expressed categorically based on the levels defined in Table 2  |
| Table 14 Overall summary of trends for the Manuherekia Rohe using raw data for rivers andcontinuous data for lakes. Confidence is expressed categorically based on the levels defined inTable 2  |
| Table 15 Summary of river state, red cells show where state does not meet the national   bottom line in one or more variable   |
| Table 16 Summary of river sites where trends are greater than 'likely' or 'unlikely' (raw data).Confidence is expressed categorically based on the levels defined in Table 2   |

Page 12 | 155

## 2000-2020

| Table 17 Summary of Lake Or<br>Confidence is expressed cate                                | low trends when they are greater than 'likely' or 'unlikely'.<br>prically based on the levels defined in Table 2  |
|--|---|
| Table 18 Overall summary of expressed categorically based                                  | rends for Lake Onslow (continuous data). Confidence is on the levels defined in Table 2.  |
| Table 19 Summary of river ar<br>national bottom line in one o<br>DRP (median and Q95) have | lake state, red cells show where state does not meet the<br>more variable. There is no national bottom line for DRP, but<br>een included in the table when sites achieve a band 'D'92   |
| Table 20 Summary of Lake Tu  | kitoto State where attributes are graded D or below93   |
| Table 21 Summary of river sit<br>is expressed categorically bas                            | s where trends are greater than 'likely' or 'unlikely'. Confidence<br>d on the levels defined in Table 294  |
| Table 22 Summary lake sites<br>'unlikely'. Confidence is expre                             | here trends (continuous data) are greater than 'likely' or sed categorically based on the levels defined in Table 295   |
| Table 23 Overall summary of<br>continuous data for lakes. Co<br>Table 2                    | rends for the Lower Clutha Rohe using raw data for rivers and fidence is expressed categorically based on the levels defined in   |
| Table 24 Summary of river ar<br>national bottom line in one o<br>DRP (median and Q95) have | lake state, red cells show where state does not meet the<br>more variable. There is no national bottom line for DRP, but<br>een included in the table when sites achieve a band 'D'116  |
| Table 25 Summary of river sit<br>'unlikely'. Confidence is expre                           | s (not flow adjusted) where trends are greater than 'likely' or sed categorically based on the levels defined in Table 2116   |
| Table 26 Overall summary of<br>continuous data for lakes. Co<br>Table 2                    | rends for the Dunedin Coast FMU using raw data for rivers and fidence is expressed categorically based on the levels defined in   |
| Table 27 Summary of river ar<br>national bottom line in one o<br>DRP (median and Q95) have | lake state, red cells show where state does not meet the<br>more variable . There is no national bottom line for DRP, but<br>een included in the table when sites achieve a band 'D'125 |
| Table 28 Summary of river sit<br>is expressed categorically bas                            | s where trends are greater than 'likely' or 'unlikely'. Confidence<br>d on the levels defined in Table 2126   |
| Table 29 Overall summary of<br>continuous data for lakes. Co<br>Table 2                    | rends for the North Otago FMU using raw data for rivers and fidence is expressed categorically based on the levels defined in   |
| Table 30 Summary of river an<br>national bottom line in one o                              | d lake state, red cells show where state does not meet the more variable134   |
| Table 31 Summary of river sit<br>Confidence is expressed cate                              | s where trends (raw data) are greater than 'likely' or 'unlikely'.<br>prically based on the levels defined in Table 2135  |
| Table 32 Overall summary of<br>continuous data for lakes. Co<br>Table 2                    | rends for the Catlins FMU using raw data for rivers and fidence is expressed categorically based on the levels defined in   |
|  | Page 13   155   |

## 2000-2020

| Table 33 Summary of lake sites where attributes fall below the NPSFM national bottom line   |
|---|
|   |
| Table 34 Summary of river sites where attributes fall below the NPSFM national bottom line. |
| There is no national bottom line for DRP, but DRP (median and Q95) have been included in    |
| the table when sites achieve a band 'D'   |

Page 14 | 155

2000-2020

# **Executive Summary**

This study analysed the available water quality data in the Otago Region. The state of water quality in the region is reported, on a site by site basis, relative to targets specified in the National Objectives Framework (NOF) of the National Policy Statement-Freshwater Management (NPSFM, 2020). In addition, the study assessed water quality trends site by site, and across Otago as a whole. ORC engaged Landwaterpeople (LWP) to evaluate water quality state (LWP, 2020a) and undertake trend analysis (LWP, 2020b).

State analysis was based on water quality samples collected over a five-year period from 1 July 2015 to 30 June 2020 and compared to the five-year period 1 July 2012 to 30 June 2017, which is defined as the baseline state (NPSFM, 2020).

Water quality analysed represented 10 physico-chemical and microbiological variables and biological indicators for 124 monitoring sites in the region; the sites included ORC monitored river sites (110), NIWA monitored National River Water Quality Network (NRWQN) sites (5) and ORC monitored lake sites (9 lakes, 22 sites/depths). All variables were evaluated for state and trends at all sites (when sufficient data was available), this report describes only river state and trends for the variables that specifically relate to the NPSFM (2020); chlorophyll-a, total nitrogen, total phosphorus, ammoniacal-nitrogen, nitrate, suspended fine sediment, macroinvertebrate community index (MCI), macroinvertebrate average score per metric (ASPM), dissolved reactive phosphorus and E. coli. The state and trend outputs for all sites and variables are provided in supplementary files (a full list of these files is provided in Appendix A). Sites were graded as a NOF Band (A, B, C, D, and for E. coli) (for NOF Criteria) for each variable based on a comparison of the assessed state with the relevant criteria.

Trend analysis was carried out for 10-year and 20-year periods ending on 1 September 2020 for all site and water quality variable combinations that met a minimum requirement for numbers of observations. The methods used for statistical trend analyses were Kendall's test of rank correlation and the Sen slope estimator (SSE), which have both been used for trend analysis of water quality for several decades (Hirsch et al., 1982). LWP (2020b) considered flow adjustment as part of the trend assessment.

Individual site trend estimates were aggregated, to provide an overall picture of trends for the region. This was done graphically using stacked bar charts showing proportions of sites for each variable that fall into different trend direction confidence categories. For the 10-year trend period the predominant trend direction was variable by water quality analyte but that the 20-year trends were predominantly degrading for all variables apart from ammoniacal nitrogen.

The most obvious pattern associated with the assessment of water quality state was that almost all sites passed the NOF criteria for ammoniacal-N toxicity and nitrate toxicity. There are obvious spatial patterns associated with the variation in grades, water quality is best at river and stream reaches located at high or mountainous elevations under predominantly native cover. These sites tend to be associated with the upper catchments of larger rivers (e.g. Clutha River/Matau-Au) and the outlets from large lakes (e.g. Hawea, Wakatipu and Wanaka). Water quality is generally poorer at sites located on smaller, low-elevation streams that drain pastoral or urban catchments.

There is a lack of detailed information held by Otago Regional Council on local or catchment scale land use change or land management practice changes. This severely limits Council's ability to comment on

Page 15 | 155

## 2000-2020

drivers of trends evident across Otago. This is likely to be addressed by requirements in the NPSFM (2020), which requires that freshwater is managed in an integrated way that considers the effects of the use and development of land on a whole-of-catchment basis, including the effects on receiving environments.

2000-2020

# 1 Introduction

Otago Regional Council (ORC) operates a State of Environment (SoE) water quality monitoring network in lakes and rivers throughout the region for monitoring the state and trends in water quality and reporting on policy effectiveness. Prior to mid-2018, there were fewer monitoring sites in the Region, following a review (NIWA, 2017), a more extensive monitoring programme commenced in mid-2018 to better represent environmental classes in the Otago region, based largely on the River Environment Classification (REC; Snelder et al. 2004).

This study analysed the available water quality data for rivers in the Otago Region. The state of water quality in the region is reported, on a site by site basis, relative to the attribute states specified in the National Objectives Framework (NOF) of the National Policy Statement-Freshwater Management (NPSFM, 2020). In addition, the study assessed water quality trends site by site and across the Otago region as a whole. All results are presented in Freshwater Management Unit (FMU) chapters.

The aims of this report are to:

- Report on the state of water quality and ecology indicators in rivers and lakes relative to attribute states in the NPSFM 2020 and to each FMU/Rohe;
- Identify significant trends in water quality and apply level of confidence categories to convey the confidence that the trend (or step change) indicates improving water quality.
- Meet Council's RMA obligations reporting the State of the Environment of Otago's rivers and lakes.

Page 17 | 155

2000-2020

# 2 Otago Region

# 2.1 Region Description

The Otago region covers a land area of 32,000 km<sup>2</sup>, from the Waitaki River in the north to Brothers Point in the south, and inland to Lake Wakatipu, Queenstown, Hawea, Haast Pass and Lindis Pass. The distinctive and characteristic landscape of Otago includes the Southern Alps and alpine lakes; large high- country stations; dry central areas, with tussock grassland and tors; and dramatic coastlines around the Otago Peninsula and the Catlins. Lowland pasture country is common in the west. The character of the region's water bodies is diverse, reflecting the variation in environmental conditions throughout the region.

The Clutha River/Mata-Au drains much of the Otago region. Its catchment area totals 21,000 km<sup>2</sup>, and 75% of the total flow of the river at Balclutha comes from the outflows of Lakes Hawea, Wanaka and Wakatipu. Larger rivers feeding into the Clutha catchment include the Cardrona, Lindis, Shotover, Nevis, Fraser, Manuherekia, Teviot, Pomahaka, Waitahuna and Waiwera rivers. The Clutha and its principal tributary, the Kawarau River, pass through gorges, two of which are dammed for hydro-electricity generation.

The second largest catchment in Otago is the Taieri River (5,060 km<sup>2</sup>). It rises in the uplands of Central Otago and meanders between mountain ranges before passing through an incised gorge and crossing the Taieri Plain, where it joins the waters of the Lake Waipori and Waihola catchments and becomes tidal before making its way through another gorge to the sea at Taieri Mouth.

Other significant Otago rivers drain the coastal hills in catchments of varying character. In the north, the Kakanui, Waianakarua, Shag and Waikouaiti rivers rise in high country and pass through mainly dry downlands. The Tokomairiro River, which flows through Milton, south of Dunedin, drains rolling country between the Taieri and Clutha catchments. Rivers to the south of Otago, particularly the Catlins area, emerge from wetter, often forested hills.

The environmental context in which Otago's water bodies exist is characterised by high rainfall in the Southern Alps and occasional very low rainfall in the semi-arid central Otago valleys. Despite the large water volumes in the region, parts of Otago are among the driest areas in New Zealand. Several rivers are characterised as 'water-short', including the Lindis, Manuherekia, Taieri, Shag and Kakanui rivers and their tributaries (Regional Plan: Water 2004) (ORC, 2017)

# 2.2 Freshwater management units

To give effect to the NPSFM (2020) and take a more localised approach to water and land management, ORC developed Freshwater Management Unit (FMU) boundaries incorporating the concept of ki uta ki tai (from the mountains to the sea).

In Otago five FMUs have been recognised, **Error! Reference source not found.**; Clutha/Mata-Au, Taieri, North Otago, Dunedin Coastal and Catlins. The Clutha/Mata-Au FMU has been further divided in to five sub-areas, or 'Rohe', for a more tailored water management approach in these areas. These include the Upper Lakes Rohe, Dunstan Rohe, Manuherekia Rohe, Roxburgh Rohe and Lower Clutha Rohe.

Page 18 | 155

2000-2020



Figure 1 Map showing the FMU and Rohe boundaries, State of Environment monitoring site locations are also shown.

Page 19 | 155

2000-2020

# 2.3 Upper Lakes Rohe (Clutha Matu/Au FMU)

The Upper Lakes Rohe encompasses Lake Wakatipu, Lake Wanaka, and Lake Hawea and all the tributaries that flow into them, effectively acknowledging this is a pristine, high value area. The headwaters of the catchment are predominantly in rugged, steep terrain with the highest point, Mt. Aspiring, reaching 3027 m. Numerous headwater streams such as the Dart River and Matukituki River originate along the eastern boundary of the Southern Alps and are fed by permanent glaciers.

The rivers and streams of the Upper Lakes FMU cover a broad range of river types, however based on the River Environment Classification, cool/wet and cool/extremely wet rivers are a significant contributor to total river length throughout the Rohe. The REC classes 'wet' rivers as having a mean annual rainfall of 500 mm to 1500 mm; 'extremely wet' rivers are classed as having a mean annual rainfall greater than 1500 mm. Water yields from these streams and rivers is high. Also of significance is the proportion of glacial rivers. The high proportion of native cover in the upper catchments of the large lakes falls in areas of high to very high rain and snowfall. This provides large volumes flowing from pristine catchments of exceptional quality that feeds the Southern Great Lakes.

ORC monitors 23 river sites and three lakes in the Upper Lakes Rohe, many of the river sites were established in 2018.

# 2.4 Dunstan Rohe (Clutha Matu/Au FMU)

The Dunstan Rohe is essentially the mid-section of the Clutha FMU and encompasses the Kawarau River and the Clutha River from the outflows of Lakes Wanaka, Wakatipu and Hawea, the Rohe's boundary is at the Clyde Dam. The Rohe encompasses dryer catchments where water use is high and where there are water quality pressures and high growth (Reference Required). The major tributaries of the Dunstan Rohe include the Cardrona River (347 km2), Luggate Creek (123 km2), Lindis River (1039 km2) and the Shotover River (1091 km2). ORC monitors 16 river sites and three lakes in the Dunstan Rohe.

# 2.5 Manuherekia Rohe (Clutha Matu/Au FMU)

The Manuherekia catchment is located north-east of Alexandra, Central Otago, and is the largest subcatchment of the Clutha / Mata-au catchment. It has a distinct community, high water use and highly modified hydrology. The Manuherekia has a catchment area of approximately 3035 km<sup>2</sup>.

The Manuherekia catchment can be divided into two major sub-catchments. The eastern Ida Valley drains the eastern and south-eastern Otago uplands ('Rough Ridge'). The western Manuherekia Valley is separated from the Ida Valley by the central Raggedy Range, where the Idaburn River drains through a single gorge into the Manuherekia River. The river's headwaters are in the Hawkdun Range, and the catchment is surrounded by mountainous terrain, except to the south-west, where it joins the Clutha River/Mata-Au at Alexandra (Kiensle, 2008).

Low rainfall in the valley bottoms led to the early development of extensive water storage and irrigation schemes. Consequently, three reservoirs were established in the Manuherekia catchment to provide water for irrigation. Falls Dam was built in 1935 to capture the high rainfall water supply in the

Page 20 | 155

#### 2000-2020

northern high-altitude part of the catchment and has a capacity of 11 Mm<sup>3</sup>. Poolburn Reservoir was constructed in 1931, with a capacity of 26 Mm<sup>3</sup>, while Manorburn Reservoir was built in 1935 and has a capacity of 51 Mm<sup>3</sup> (Kiensle, 2008).

There are several major irrigation schemes. Flow of the Manuherekia River is partly controlled by Falls Dam. Blackstone Hill, Omakau, Manuherekia, and Galloway irrigation schemes take water out of the Manuherekia River and distribute the water through a network of open water channels to irrigate the Manuherekia Valley. The Poolburn Reservoir is used to store water to irrigate Ida Valley. Water from the Manorburn Reservoir is partly diverted into the Manuherekia Valley over an open water race to irrigate the upper Galloway Irrigation Scheme. The rest of the Manorburn water is used for irrigation in the Ida Valley (Kiensle, 2008).

The contribution of water from the Ida Valley to the mainstem Manuherekia is minimal during the summer months. For example, during the 2008-2009 irrigation season the Pool Burn at Cobb Cottage (after the confluence of the Idaburn and Poolburn, prior to entering the gorge) had a median flow of 50 l/s, compared to a median flow of 4433 l/s in the Manuherekia at Ophir.<sup>1</sup> This is important to note when considering the contribution of water quality contaminants in the lower Manuherekia River during the irrigation season. ORC monitors eight river sites in the Manuherekia Rohe.

# 2.6 Roxburgh Rohe (Clutha Matu/Au FMU)

The Roxburgh Rohe is bounded to the north by the Clyde Dam and to the South at Beaumont. The Rohe covers just over 1,000 square kilometres and encompasses catchments of the Fraser River (327 km2), the Teviot River (332 km<sup>2</sup>) and the Benger Burn (131 km<sup>2</sup>) as well as small tributaries entering the Clutha between Clyde and Beaumont. Lake Onslow is in the Rohe. Onslow is a man-made 830ha lake at the head of the Teviot River, 700m above sea level. ORC monitors four river sites and one lake in the Roxburgh Rohe.

# 2.7 Lower Clutha Rohe (Clutha Matu/Au FMU)

The Lower Clutha Rohe runs from Beaumont to the Pacific Ocean where the Clutha River/Mata-Au discharges to the sea near Balclutha. The Lower Clutha Rohe includes the catchments of the Tuapeka River (249 km<sup>2</sup>), Pomahaka River (2060 km<sup>2</sup>), Waipahi River (339 km<sup>2</sup>), Waiwera River (208 km<sup>2</sup>) and the Waitahuna River (406 km<sup>2</sup>).

The Pomahaka River is the largest catchment of the Lower Clutha Rohe. The upper reaches of the Pomahaka catchment are steep and dominated by tussock, while the lower reaches are primarily pastoral rolling hill country and intensively managed land. Soils are generally poorly drained requiring artificial drainage predominantly in the form of tile drains. If inappropriately managed, these tile and mole drains accelerate water and associated contaminant flows of nitrogen, phosphorous and bacteria to local watercourses. ORC monitors 14 river sites and one lake in the Lower Clutha Rohe.

Page 21 | 155

<sup>&</sup>lt;sup>1</sup> <u>https://www.orc.govt.nz/media/2437/web-version-manuherkiaa2-report2.pdf</u>.

2000-2020

# 2.8 Taieri FMU

The Taieri River rises in the Lammerlaw and Lammermoor Ranges at 1,150m above sea level. The river flows through the dry Maniototo Plain, Strath Taieri Plain and the low-lying Taieri Plain before reaching the Pacific Ocean about 30km south-west of Dunedin. The main tributaries of the Taieri River are the Kyeburn, Sutton Stream, Deep Stream, Lee Stream, Silverstream and the Waipori River.

The upper Taieri headwaters drain a relatively undeveloped area of native tussock country on the northern side of the Lammerlaw Range. The river then flows through the dry, 660km<sup>2</sup> area of the Maniototo Plain, west of the Rock and Pillar Range. The river follows an extremely meandering course through large Scroll Plains in the Maniototo. These are unique and scientifically important features of the upper catchment. Beyond the northern end of the Rock and Pillar Range, the Kye Burn flows into the Taieri and contributes high levels of sediment to the river. These high sediment loads may be in part due to historic gold mining activities in the Kye Burn Catchment.

The midreaches of the Taieri River flow through the smaller Strath Taieri Plain (occupying an area of 85km<sup>2</sup>), past Middlemarch, and through the Taieri Gorge. Many small tributaries join the main stem of the river along this sub-region.

Most of the human settlement is in the lower Taieri Catchment on the Taieri Plain (occupying an area of 180km<sup>2</sup>), where the town of Mosgiel is located. The floodplain area is intensively farmed (mostly dairying). The lower Taieri River is joined by the Silverstream, which provides high quality trout spawning and nursery habitats for the river fishery. A large floodplain and the associated Lake Waipori/Waihola wetland complex are the dominant features of the lower catchment. Part of the lower Taieri plain lies below sea level, and the potential for flooding has resulted in extensive flood protection works, including floodbank construction and channel straightening (e.g., the lower Silverstream) which has significantly altered the physical habitat quality of some river reaches. ORC monitors 19 river sites and one lake in the Taieri FMU.

# 2.9 Dunedin Coast FMU

The Dunedin Coast FMU recognises the middle coast of Otago, starting south of Karitane, encompassing Dunedin city and Otago Peninsula and the coastal catchments to Clutha/Mata-Au mouth. All of these areas have similar water quality issues and water quantity demands. ORC monitors the catchments of Lindsay's Creek, the Leith Stream and the Kaikorai Stream in the Dunedin City area and the Tokomairiro River and Akatore Creek to the South.

The Leith Stream catchment covers an area of 42 km<sup>2</sup>. The headwaters of the Leith Stream originate at the saddle between Mount Cargill and Swampy Hill and flow for 12 km in a south-easterly direction to discharge direct to the Otago Harbour, Dunedin. There are numerous tributary streams to the Leith, the principal of which is Lindsays Creek which flows for 7 km to its junction with the Leith Stream at the Dunedin Botanical Gardens. Significant areas of the lower catchment of the Leith Stream and Lindsays Creek flow through urban areas of the Dunedin City.

The Kaikorai Stream has a total catchment area of 55 km<sup>2</sup> and flows in a south westerly direction for approximately 15 km down the Kaikorai Valley into Kaikorai Estuary. The headwaters originate in the Kaikorai Hills to the north. The catchment includes the western flanks of Dunedin city and all of Green Island. The remaining area includes the communities of Fairfield and Waldronville (ORC, 2008). Fraser's Stream is a major tributary of the Kaikorai Stream and the Dunedin City Council discharges up to 560 litres per second of high-quality water from the Mt Grand Water Treatment Plant to MacLeod's Creek

Page 22 | 155

#### 2000-2020

(a tributary of Fraser's Stream). This flow significantly improves the water quality and instream values of the Kaikorai Stream downstream of the discharge point (ORC, 2008).

The Tokomairiro River is located about 48 km south-west of Dunedin and has a catchment area of 403 km<sup>2</sup>. The catchment has indistinct boundaries, with no dividing mountain ranges between it and neighbouring catchments. It is bordered to the east by tributaries of the Waihola-Waipori wetland complex (including Meggat Burn and Boundary Creek) and a number of coastal tributaries including Akatore Creek. ORC monitors eight river sites in the Dunedin Coast FMU.

## 2.10 North Otago FMU

The North Otago FMU contains parts of the lower Waitaki Plains; the Kakanui catchment that includes the Kakanui River, Kauru River and Waiareka Creek; the Waianakarua Stream and Trotters Creek along with river catchments that drain independently to the sea; and to the south, the Shag River and Waikouaiti River.

From its source in the Kakanui Mountains, the Kakanui River flows north-east for about 40 km, through gorges incised in rolling or downland country, before emerging onto plains at Clifton. The Kakanui River's water resource is heavily used for irrigation. In recent times concern has been expressed about agricultural intensification and subsequent degradation of water quality. The lower Kakanui River and Waiareka Creek are dominated by a mixture of beef, sheep, deer, cropping and particularly since the introduction of irrigation water into the Waiareka Creek catchment by the NOIC irrigation scheme dairy farming. In contrast, land use in the Kauru and upper Kakanui are typified by red tussock, native forest, plantation forestry or pasture for red deer, sheep and beef. The water quality in the alluvial gravels of the Kauru River and the main-stem Kakanui River, particularly upstream of Gemmels Crossing, is influenced by groundwater surface water interaction. There is very little groundwater surface water interaction and Wilson, 2013).

The Waianakarua River is a small river with a catchment area of 262 km<sup>2</sup> which rises in the Horse Range and Kakanui Mountains in North Otago. Much of the catchment consists of extensively grazed grasslands and scrub, native forest, and plantation forestry but intensification of land use in the lower catchment has occurred in recent years.

The Shag River catchment covers an area of 550 km<sup>2</sup>. The Shag River is a medium sized river with its headwaters originating on the south-western slopes of Kakanui Peak in the Kakanui Mountains. From here it flows 90km in a south-easterly direction past the township of Palmerston before entering the Pacific Ocean to the south of Shag Point. The Shag catchment is dominated by agriculture and forestry with some short-rotation cropping in the lower catchment.

The Waikouaiti catchment area covers 421 km<sup>2</sup>, the river has two main branches, the North Branch and South Branch. The North Branch has a catchment area of 283 km<sup>2</sup> and the South Branch a catchment area of 86 km<sup>2</sup>. The remaining 52 km<sup>2</sup> includes the area downstream of the confluence of the two main branches along with the Waikouaiti Estuary. The headwaters of the north branch originate in the Macraes Flat area, whereas the south branch drains the northern slopes of the Silver Peaks. ORC monitors 15 river sites the North Otago FMU.

# 2.11 Catlins FMU

In the south of Otago, the Catlins FMU contains a collection of smaller catchments that feed into the sea south of the Clutha / Mata-au catchment. This FMU contains Otago's area of the Catlins

Page 23 | 155

#### 2000-2020

Conservation Park. The coast is dominated by sandy bays and cliffs and from there, the land rises steadily from the south-east to north-west, reaching its maximum altitude (720 m) at Mt Pye, in the headwaters of the Tahakopa and Catlins Rivers, and then it falls again, through rolling country, towards the Mataura River (in Southland) and the Clinton lowlands. The forested ridges provide a contrast to the cleared valleys, where more intensive agricultural activities are present. Headwaters of all major rivers rising from within the Catlins have their vegetation intact.

ORC monitors four rivers in the Catlins FMU, the Catlins River, Owaka River, Maclennan River and the Tahakopa. The Catlins flows south-eastward. Its total length is 42 km and it shares its estuary with the Owaka River, which flows into the Pacific Ocean at Pounawea, 28 km south of Balclutha. The Owaka River is 30 km long and flows south-east. Its source is on the slopes of Mt Rosebery. The Tahakopa River flows south-east through the Catlins. Its total length is 32 km, and it flows into the Pacific Ocean 30 km east of Waikawa, close to the settlement of Papatowai. The Maclennan River is 17.5 km long and enters the Tahakopa River near Maclennan. ORC monitors four river sites Catlins FMU.

Page 24 | 155

#### 2000-2020

# **3 ORC monitoring programme**

# 3.1 Water Quality Sites

SoE monitoring sites covered in this report include 110 river sites<sup>2</sup> and 22 monitoring sites/depths for lakes. NIWA monitors an additional five sites in the Otago region as part of the National River Water Quality Network (NRWQN). **Error! Reference source not found.** Figure 1 shows the location of the river and lake monitoring sites. Significant changes to the SoE monitoring programme have occurred over the years, the main changes are:

- Up to June 2013, ORC collected surface water quality samples on a bi-monthly basis. From July 2013, sampling frequency increased to monthly sampling.
- Prior to mid-2018, there were fewer monitoring sites in Otago, following a review (NIWA, 2017), a more extensive monitoring programme commenced in mid-2018 to better represent environmental classes in the Otago region, based largely on the River Environment Classification (REC). The number of river sites monitored expanded from 65 to 106.
- Prior to mid-2018 SoE lake monitoring sites consisted of a mix of lake-outlet (lakes Wanaka, Wakatipu and Hawea) and lake-shore (lakes Dunstan, Hayes, Johnson, Onslow, Waihola and Tuakitoto) sampling sites. Following NIWA's review, all lake sites other than Tuakitoto and Onslow now are mid-lake sampled with full vertical water column profiled on every sampling occasion.
- A review of the biomonitoring programme (NIWA, 2017) to better represent environmental classes in Otago, meant the cessation of several macroinvertebrate sites in 2018 and the adoption of several new sites.

# 3.2 Water quality variables

River and lake water quality was assessed using variables that characterise physical, chemical and microbiological conditions, and macroinvertebrate community composition. All variables included are attributes described in NPSFM 2020 (Appendix 2A or 2B).

## 3.2.1 Phytoplankton, Periphyton and Nutrients

Healthy freshwater ecosystems have low (oligotrophic) to intermediate (mesotrophic) levels of living material and primary production (growth of plants or algae). High levels of nutrients, primarily nitrogen (nitrate) and phosphorus (phosphate), can lead to water bodies becoming eutrophic. Eutrophic states are associated with periodic high biomass (blooms) of plants or algae, including suspended algae (phytoplankton) in lakes and algae on the beds of streams and rivers (periphyton).

Chlorophyll-a is a common method for estimating stream periphyton biomass (Ministry for the Environment, 2000) because all types of algae contain chlorophyll-a, this metric reflects the total amount of live algae in a sample. The trophic state of a water body is the amount of living material (biomass) that it supports. The NPSFM2020 specifies attributes for tropic state based on

Page 25 | 155

<sup>&</sup>lt;sup>2</sup> SoE monitoring currently covers 106 river sites. Some sites were discontinued in 2018, but are reported.

#### 2000-2020

phytoplankton biomass in lakes (NPSFM 2020, , Appendix 2A, Table 1) and periphyton biomass in rivers (NPSFM 2020, Appendix 2A, Table 2), chlorophyll a is the measure of biomass that the NOF phytoplankton and periphyton attributes are based on.

Nitrate (NO3N), ammoniacal-N (NH4N), dissolved reactive phosphorus (DRP), total nitrogen (TN) and total phosphorus (TP) influence the growth of benthic river algae (periphyton), lake planktonic algae (phytoplankton) and vascular plants (macrophytes). The NPSFM specifies additional attributes for TN and TP in lakes (NPSFM 2020, Appendix 2A, Table 3 and Table 4).

The NPSFM 2020 does not specify nutrient concentration criteria to manage the trophic state of rivers, because the relationship between trophic state and nutrient concentrations varies between rivers even at the regional scale. The nutrient criteria to achieve periphyton biomass objectives in rivers are river specific and should be derived at the local level (MfE, 2018).

The Ministry for the Environment has produced guidance (MfE, 2020) for defining nutrient concentrations to manage the NPSFM 2020 periphyton attribute states in rivers. The guidance is centered around spatial exceedances for TN and DRP. Spatial exceedance is used because deriving nutrient targets to achieve a target periphyton growth cannot be 100% certain due to natural variability, complex interactions in the environment, and the complexity of the relationship between nutrients and periphyton abundance (MfE, 2020). Given the short record of chlorophyll-a observations in the region, these nutrient concentration criteria provide a useful alternative for estimating trophic state in the region's rivers.

In this report TN and DRP median concentrations are compared to the spatial exceedance criteria of 20% (as opposed to 10% or 30%). At this level there is some risk (i.e., 20%) that the chlorophyll-*a* response at some sites will exceed the desired chlorophyll *a* threshold, even if DRP or TN concentration targets are achieved.

In addition to the MfE guidance, the NPSFM 2020 provides an attribute table for DRP in rivers to protect ecosystem health. In combination with other conditions favouring eutrophication, DRP enrichment drives excessive primary production and significant changes in macroinvertebrate and fish communities, as taxa sensitive to hypoxia are lost. The NPSFM 2020 (Appendix 2B, Table 20) describes that at concentrations below the national bottom line, it is expected that ecological communities are impacted by substantial DRP elevated above natural reference conditions.

#### 3.2.2 Toxicants

When ammonia is present in water at high enough concentrations, it is difficult for aquatic organisms to sufficiently excrete the toxicant, leading to toxic build-up in internal tissues and blood, potentially leading to death. Environmental factors, such as pH and temperature, can affect ammonia toxicity to aquatic animals. The NPSFM 2020 has developed an ammonia toxicity risk framework (NPSFM 2020, Appendix 2A, Table 5) when toxicity concentrations are below the national bottom line, toxicity starts impacting regularly on the 20% of the most sensitive species.

Nitrate generally impacts on trophic state at much lower concentrations than those that are toxic. Because of this, nitrate will generally be managed well within toxic levels by the requirement to manage trophic state (e.g., periphyton). The NPSFM has developed a nitrate toxicity risk framework (NPSFM 2020, Appendix A, Table 6) when toxicity concentrations are below the national bottom line, toxicity has growth effects on up to 20% of species monitored.

Page 26 | 155

2000-2020

#### 3.2.3 Suspended sediment

Suspended fine sediment can severely affect values around water, particularly around ecosystem health. High concentrations of suspended sediment have a 'high impact on instream biota and ecological communities are significantly altered and sensitive fish and macroinvertebrate species are lost or at high risk of being lost' (NPSFM, 2020). Suspended fine sediment can be monitored by clarity or turbidity measurements.

Clarity is a measure of light attenuation due to absorption and scattering of dissolved and particulate material in the water column. Clarity is monitored because it affects primary production, plant distributions, animal behaviour, aesthetic quality and recreational values. Cl;arity is correlated with suspended solids, which can impede fish feeding and cause riverbed sedimentation. Clarity is the metric used in the NPSFM 2020 suspended fine sediment attribute table (NPSFM 2020, Appendix A, Table 8).

Turbidity refers to light scattering by suspended particles. Nephelometric turbidity is generally inversely correlated with visual water clarity (Davies-Colley and Smith, 2001), but unlike visual clarity, turbidity measurements do not account for optical effects (i.e., absorption) of dissolved materials. The NPSFM 2020 allows for the conversion of turbidity to visual clarity. ORC does not measure visual clarity and applies this conversion.

#### 3.2.4 Aquatic Life

Macroinvertebrates are an important component of streams and rivers because they aid ecosystem processes and provide food for fish and some birds. As macroinvertebrates have a relatively long-life span, they are good indicators of environmental conditions over a prolonged period. Macroinvertebrates are included in the NPSFM 2020 as attributes requiring an action plan (NPSFM 2020, Appendix 2B, Tables 14-15).

The main measure of macroinvertebrate communities, the MCI index, is designed specifically for stony-riffle substrates in flowing water. The MCI is responsive to multiple stressors, but not all stressors, and as such provides a good indicator of the overall condition of the macroinvertebrate component of stream ecosystem health.

MCI values can be affected by factors other than water quality, so it is more informative to consider changes in MCI values at the same site over a period, rather than among sites throughout the catchment. For example, a change in MCI value at a site may be due to human activities leading to increased nitrogen or sedimentation with resulting ecological consequences (Clapcott et al. 2018). Sites with an MCI score of less than 80 are classified as poor, those scoring 80-100 as fair, those scoring 100-120 as good, and those scoring higher than 120 as excellent (Stark and Maxted, 2007).

The NPSFM has attribute states for Macroinvertebrate Community Index (MCI) score; Quantitative Macroinvertebrate Community Index (QMCI) score and Macroinvertebrate Average Score Per Metric (ASPM). Historical monitoring by ORC has included the Semi-Quantitative Macroinvertebrate Community Index (SQMCI) score, rather than QMCI. As the two are not directly comparable the QMCI metric is not shown.

The Average Score Per Metric (ASPM) was introduced by Collier (2008). This is an aggregation method for assessing wadeable stream ecosystem health considering the relative responses of core metrics and is composed of three individual metrics, the MCI, EPT richness to the total taxa found and % EPT abundance. EPT Richness Index estimates water quality by the relative abundance of three major

Page 27 | 155

#### 2000-2020

orders of stream insects that have low tolerance to water pollution. EPT can be expressed as a percentage of the sensitive orders (E= Ephemeroptera, P= Plecoptera, T= Tricoptera) and % EPT is the total number of EPT individuals divided by the total number of insects in the sample).

#### 3.2.5 Escherichia coli (E. coli)

The concentration of the bacterium *E.coli* is used as an indicator of human or animal faecal contamination, from which the risk to humans arising from infection or illness from waterborne pathogens during contact-recreation may be estimated.

'Water contaminated by human or animal faeces may contain a range of pathogenic (disease-causing) micro-organisms. Viruses, bacteria, protozoa, or intestinal worms can pose a health hazard when the water is used for drinking or recreational activities. It is difficult and impractical to routinely measure the level of all pathogens that may be present in fresh water. Instead, indicator bacteria are used to indicate the likely presence of untreated sewage and effluent contamination.

*E. coli* is commonly found in the gut of warm blooded organisms and is relatively easy to measure which makes it a useful indicator of faecal presence and therefore of disease-causing organisms that may be present. *E. coli* is the attribute for specifying human health for recreation objectives for fresh water because it is moderately well correlated with Campylobacter bacteria and numeric health risk levels can be calculated. Campylobacteriosis has the highest reporting rate of all New Zealand's 'notifiable' diseases' (MfE, 2018).

The NPSFM 2020 assesses river swimmability and the attribute states uses four statistical measures of *E.coli* concentrations, the overall state is determined by satisfying all numeric attribute states (NPSFM 2020, Appendix 2A, Table 9).

Page 28 | 155

2000-2020

# 4 Methods

# 4.1 Water Quality State Analysis

LWP evaluated state at ORC's river and lake monitoring sites for nutrients and bacteria, this section details the methods LWP used for state analysis and is taken directly from LWP 2020a.<sup>3</sup>

#### 4.1.1 Grading of monitoring sites

The water quality state for river and lake monitoring sites is graded based on attributes and associated attribute state bands defined by the National Objectives Framework (NOF) of the NPSFM (2020) detailed in Table 1.

Each table of Appendix 2 of the NPSFM (2020) represents an attribute that must be used to define an objective that provides for a particular environmental value. For example, Appendix 2A, Table 6 defines the nitrate toxicity attribute, which is defined by nitrate-nitrogen concentrations that will ensure an acceptable level of support for "Ecosystem health (water quality)" value. Objectives are defined by one or more numeric attribute states associated with each attribute. For example, the nitrate-nitrogen attribute has two numeric attribute states defined by the annual median and the 95<sup>th</sup> percentile concentrations.

For each numeric attribute, the NOF defines categorical numeric attribute states as four (or five) attribute bands, which are designated A to D (or A to E, in the case of the *E. coli* attribute). The attribute bands represent a graduated range of support for environmental values from high (A band) to low (D or E band). The ranges for numeric attribute states that define each attribute band are defined in Appendix 2 of the NPSFM 2020. For most attributes, the D band represents a condition that is unacceptable (with the threshold between the C and the D band being referred to as "bottom line") in any waterbody nationally. In the case of the NO3N (toxicity) and NH4N (toxicity) attributes in the NPSFM 2020, the C band is unacceptable, and for the DRP attribute, no bottom line is specified.

The primary aim of the attribute bands designated in the NPSFM 2020 is as a basis for objective setting as part of the NOF process. The attribute bands are intended to be simple shorthand for communities and decision makers to discuss options and aspirations for acceptable water quality and to define objectives. Attribute bands avoid the need to discuss objectives in terms of technically complicated numeric attribute states and associated numeric ranges. Each band is associated with a narrative description of the outcomes for values that can be expected if that attribute band is chosen as the objective. However, it is also logical to use attribute bands to provide a grading of the current state of water quality; either as a starting point for objective setting or to track progress toward objectives.

Page 29 | 155

<sup>&</sup>lt;sup>3</sup> LWP Ltd., 2020a. State of Lake and River Water Quality in the Otago Region. For records up to 30 June 2020, prepared for Otago Regional Council.

2000-2020

| NPSFM 2020 Reference - NOF Attribute    | Water body type  | Minimum Sample<br>Requirements                                | Numeric attribute state description   | Units                    |
|---|------------------|---|---|--------------------------|
|   |                  | •   |   |                          |
| A2A; Table 1 - Phytoplankton            | Lakes            |   | Median of phytoplankton chlorophyll-a   | mg chl-a m <sup>-s</sup> |
|   |                  |   | Annual maximum of phytoplankton chlorophyll-a   | mg chl-a m⁻³             |
| A2A; Table 2 – Periphyton               | Rivers           | Minimum of 3 years of data                                    | 92nd percentile of periphyton chlorophyll- <i>a</i> for default river class <sup>2</sup>    | mg chl-a m <sup>-3</sup> |
|   |                  |   | 83rd percentile of periphyton chlorophyll- <i>a</i> for productive river class <sup>1</sup> | mg chl-a m <sup>-3</sup> |
| A2A; Table 3 – Total Nitrogen           | Lakes            |   | Median concentration of total nitrogen  | mg m⁻³                   |
| A2A; Table 4 – Total Phosphorus         | Lakes            |   | Median concentration of total phosphorus  | mg m <sup>-3</sup>       |
| A2A; Table 5 - Ammonia                  | Lakes and Rivers |   | Median concentration of Ammoniacal-N  | mg l <sup>-1</sup>       |
|   |                  |   | Maximum concentration of Ammoniacal-N   | mg l <sup>-1</sup>       |
| A2A; Table 6 - Nitrate                  | Rivers           |   | Median concentration of Nitrate   | mg l <sup>-1</sup>       |
|   |                  |   | 95th percentile concentration of Nitrate  | mg l <sup>-1</sup>       |
|   |                  | Median of 5-years of at least<br>monthly samples (at least 60 |   |                          |
| A2A.; Table 8 - Suspended fine sediment | Rivers           | samples)  | Median visual clarity   | m                        |
| A2A; Table 9 - Escherichia coli         | Rivers and Lakes | minimum of 60 samples<br>over a maximum of 5-years            | % exceedances over 260 cfu 100 mL <sup>-1</sup>   | %                        |
|   |                  |   | % exceedances over 540 cfu 100 mL <sup>-1</sup>   | %                        |
|   |                  |   | Median concentration of <i>E. coli</i>  | cfu 100 ml <sup>-1</sup> |
|   |                  |   | 95th percentile concentration of <i>E. coli</i>   | cfu 100 ml <sup>-1</sup> |
| A2B; Table 14 - Macroinvertebrates      | Rivers           | State calculated as 5-year median                             | Median MCI score  | -                        |
| A2B; Table 15 - Macroinvertebrates      | Rivers           |   | Median ASPM score   | -                        |
| A2B; Table 20 - DRP                     | Rivers           |   | Median concentration of DRP   | mg l <sup>-1</sup>       |

## Table 1 River water quality variables included in this report, including NPSFM 2020 reference and water body type

Page 30 | 155

2000-2020

|  | 95th percentile concentration of DRP | mg l <sup>-1</sup> |
|--|--------------------------------------|--------------------|
|--|--------------------------------------|--------------------|

Page 31 | 155

2000-2020

A site can be graded for each attribute by assigning it to attribute bands (e.g., a site can be assigned to the A band for the NO3N toxicity attribute). A site grading is done by using the numeric attribute state (e.g., annual median nitrate-nitrogen) as a compliance statistic. The value of the compliance statistic for a site is calculated from a record of the relevant water quality variable (e.g., the median value is calculated from the observed monthly NO3N concentrations). The site's compliance statistic is then compared against the numeric ranges associated with each attribute band and a grade assigned for the site (e.g., an annual median NO3N concentration of 1.3 mg/l would be graded as 'B -band, because it lies in the range >1.0 to  $\leq$ 2.4 mg/l). Note that for attributes with more than one numeric attribute state, a grade for each numeric attribute state has been provided (e.g., for the NO3N (toxicity) attribute, grades are defined for both the median and 95<sup>th</sup> percentile concentrations).

## 4.1.2 Handling censored values

Censored values were replaced by imputation for the purposes of calculating the compliance statistics. Left censored values (values below the detection limit(s)) were replaced with imputed values generated using ROS (Regression on Order Statistics; Helsel, 2012), following the procedure described in Larned *et al.* (2015). The ROS procedure produces estimated values for the censored data that are consistent with the distribution of the uncensored values and can accommodate multiple censoring limits. When there are insufficient non-censored data to evaluate a distribution from which to estimate values for the censored observations, censored values are replaced with half of their reported value.

Censored values above the detection limit were replaced with values estimated using a procedure based on "survival analysis" (Helsel, 2012). A parametric distribution is fitted to the uncensored observations and then values for the censored observations are estimated by randomly sampling values larger than the censored values from the distribution. The survival analysis requires a minimum number of observations for the distribution to be fitted; hence in the case that there were fewer than 24 observations, censored values above the detection limit were replaced with 1.1\* the detection limit.

#### 4.1.3 Time-period for assessments

When grading sites based on NPSFM 2020 attributes, it is general practice to define consistent timeperiods for all sites and to define the acceptable proportion of missing observations (i.e., data gaps) and how these are distributed across sample intervals so that site grades are assessed from comparable data. The time-period, acceptable proportion of gaps and representation of sample intervals by observations within the time-period are commonly referred to as site inclusion or filtering rules (Larned *et al.*, 2018).

The grading assessments were made for the 5-year time-period to end of June 2020. The start and end dates for this period were determined by the availability of quality assured data, reporting timeperiods and consideration of statistical precision of the compliance statistics used in the grading of sites. The statistical precision of the compliance statistics depends on the variability in the water quality observations and the number of observations. For a given level of variability, the precision of a compliance statistic increases with the number of observations. This is particularly important for sites that are close to a threshold defined by an attribute band because the confidence that the assessment of state is 'correct' (i.e., that the site has been correctly graded) increases with the

Page 32 | 155

#### 2000-2020

precision of the compliance statistics (and therefore with the number of observations). As a general rule, the rate of increase in the precision of compliance statistics slows for sample sizes greater than 30 (i.e., there are diminishing returns on increasing sample size with respect to precision (and therefore confidence in the assigned grade) above this number of observations (McBride, 2005).

In this study, a period of five years represents a reasonable trade-off for most of the attributes because it yielded a sample size of 30 or more observations for many sites and attribute combinations. The five-year period for the state analyses is also consistent with national water-quality state analyses (Larned *et al.*, 2015, 2018), as well as guidance for a number of specific attributes within the NPSFM 2020 (**Error! Reference source not found.**). Where no guidance was provided, a default filtering rule that required at least 30 observations in the 5-year time-period was used. For annually sampled macroinvertebrate variables, which are generally less variable than physical or chemical water quality variables, the nominated minimum sample size requirement was reduced to 5.

For grading the suspended fine sediment and *E. coli* attributes, the NPSFM 2020 requires 60 observations over 5-years. For monthly monitoring, this requires collection of all monthly observations (i.e., no missing data). All ORC records have at least one missing observation associated with the national COVID-19 lockdown in April 2020, and so no sites met this requirement for the selected time-periods. For this study, the rule to require observations for 90% of months over the 5-year period (54 observations) was relaxed. Therefore, within the supplementary files state assessments for all sites are provided regardless of whether they meet the filtering rules, as well as details about the number of observations and number of years with observations.

#### 4.1.4 Calculation of water clarity

The NPSFM 2020 suspended fine sediment attribute is based on observations of visual clarity. ORC river monitoring programme does not include visual clarity but does routinely collect turbidity observations. Franklin et al. (2020) define a relationship between median clarity and median turbidity, based on a regression of 582 sites across New Zealand as:

$$ln(CLAR) = 1.21 - 0.72 ln(TURB)$$

where CLAR is site median visual clarity (m) and TURB is site median turbidity (NTU). In this study, median turbidity values over the 5-year time-period was calculated, and then calculated median clarity using the above relationship in order to grade the sites against the NPSFM 2020 suspended fine sediment attribute.

Sites operated by NIWA as part of the national monitoring network include observations for clarity, and therefore for these sites performance against the NPSFM 2020 suspended fine sediment attribute has been evaluated with the observed (rather than modelled) clarity values.

#### 4.1.5 pH Adjustment of Ammonia

Ammonia is toxic to aquatic animals and is directly bioavailable. When in solution, ammonia occurs in two forms: the ammonium cation  $(NH_4^*)$  and unionised ammonia  $(NH_3)$ ; the relative proportions of the forms are strongly dependent on pH (and temperature). Unionised ammonia is significantly more toxic to fish than ammonium, hence the total ammonia toxicity increases with increasing pH (and/or temperature) (ANZECC, 2000). Standards related to ammoniacal-N concentrations in freshwater typically require a correction to account for pH and temperature. A pH correction to  $NH_4$ -N was applied to adjust values to equivalent pH 8 values, following the methodology outlined in Hickey

Page 33 | 155
### 2000-2020

(2014). For pH values outside the range of the correction relationship (pH 6-9), the maximum (pH<6) and minimum (pH>9) correction ratios were applied (Reference required).

# 4.1.6 Evaluation of compliance statistics

For compliance statistics specified and annual (maximum, median, 95<sup>th</sup> percentile) values in the NPSFM 2020, have been calculated over the entire 5-year state period.

Page 34 | 155

2000-2020

# 4.2 Water Quality Trend Analysis

LWP evaluated 10 and 20-year trends at ORC's river and lake monitoring sites for each measured variable (primarily nutrients and bacteria), this section details the methods LWP used for trend analysis and is taken directly from LWP 2020b.<sup>4</sup>

#### 4.2.1 River water quality data

The river water quality data used in this study was supplied by ORC (110 sites) and NIWA (8 sites) and comprised 114,600 observations at 115 monitoring sites (3 sites overlapped between the ORC and NIWA data) of the variables at shown in **Error! Reference source not found.** 

### 4.2.2 Lake quality data

The lake water quality data used in this study comprised 18,612 observations at 22 monitoring sites/depths of the 13 variables. Some sites had two depths associated with water quality sampling. The different depths were treated as independent sampling sites.

The ORC lake monitoring programme underwent major changes over the period in 2016-2018. Several new sites were introduced, and older sites were phased out. Many of these older sites had long term records (starting in approximately 2000) but were ceased by mid-2018. Many of the water quality variables at the new sites were also monitored at these locations during an intensive investigation period between 2006-2009. These data were extracted from physical records for use in this study. The extracted data was not associated with censoring information. Observations were reinstated as censored values as part of the pre-processing based on the detection limits in operation for the same variables in other lakes over the same time-period.

#### 4.2.3 Flow data

Many of the river water quality monitoring sites also had flow recorded. These records were obtained from the ORC database. Flows associated with the NIWA sites were a combination of measured and modelled flows. Water quality observations can be strongly associated with flow, and the effect of flow on water quality can be accounted for in analysis of trends. Mean daily flows were recorded with 51 of the 115 monitoring sites (and, of these sites, approximately 87% of all sample occasions had flow measured).

#### 4.2.4 Sampling dates, seasons, and time-periods for analyses

In trend assessments, there are several reasons why it is important to define the trend period and seasons and to assess whether observations are adequately distributed over time. First, because variation in many water quality variables is associated with the time of the year or season, the

Page 35 | 155

<sup>&</sup>lt;sup>4</sup> LWP Ltd., 2020b. ORC River, Groundwater and Lake water quality Trend analysis. For 10 and 20-year periods up to September 2020, prepared for Otago Regional Council.

#### 2000-2020

robustness of trend assessment is likely to be diminished if the observations are biased to certain times of the year. Second, a trend assessment will always represent a time-period; essentially that defined by the first and last observations. The assessment's characterisation of the change in the observations over the time-period is likely to be diminished if the observations are not reasonably evenly distributed across the time-period. For these reasons, important steps in the data compilation process include specifying the seasons, the time-period, and ensuring adequately distributed data.

Monitoring programmes are generally designed to sample with a set frequency, (e.g., monthly, quarterly). The trend analysis 'season' is generally specified to match this sampling frequency (e.g., seasons are months, bi-months, or quarters). There is therefore generally an observation for each sample interval (i.e., each season, such as month or quarter, within each year). Sampling frequency for some variables is annually. For example, annual sampling is common for biological sampling such as macro-invertebrates. In this case the 'season' is specified by the year.

Two common deviations from the prescribed sampling regime are (1) the collection of more than one observation in a sample interval (e.g., two observations within a month) and (2) a change in sampling interval within the time-period. Both of these deviations occurred in the ORC datasets, particularly type (2), as there was a network wide change in sampling frequency in 2013, largely moving from bimonthly to monthly monitoring for rivers, and from biannual to quarterly for groundwater in 2011. For type (1) deviations, the median within each sample interval was taken. For type (2) deviations, the coarser sampling interval to define seasons was used. For the part of the record with a higher frequency, the observations in each season were defined by taking the observation closest to the midpoint of the coarser season. The reason for not using the median value in this case is that it will induce a trend in variance, which will invalidate the null distribution of the test statistic (Helsel *et al.*, 2020).

The trend at all sites was characterised by the rate of change of the central tendency of the observations of each variable through time. Because water quality is constantly varying through time, the evaluated rate of change depends on the time-period over which it is assessed (Ballantine *et al.*, 2010; Larned *et al.*, 2016). Therefore, trend assessments are specific for a given period of analysis. Trend periods of 10 and 20-years were evaluated.

For a regional study that aims to allow robust comparison of trends between sites and to provide a synoptic assessment of trends across a whole region, such as the present study, it is important that trends are commensurate in terms of their statistical power and representativeness of the time-period. In these types of studies, it is general practice to define consistent time-periods (i.e., trend duration and start date) so that all sites are subjected to the same conditions (i.e., equivalent political, climate, economic conditions). It is also general practice to define the acceptable proportion of gaps and how these are distributed across sample intervals so that the reported trends are assessed from comparable data. The acceptable proportion of gaps and representation of sample intervals by observations within the time-period are commonly referred to as site inclusion or filtering rules (Larned *et al.*, 2018) but this is also termed 'site screening criteria' and 'completeness criteria'.

There are no specific data requirements or filtering rules for trend assessments performed over many sites and variables such as the present study. The definition of filtering rules is complicated by a trade-off: more restrictive rules increase the robustness of the individual trend analyses but will generally exclude a larger number of sites thereby reducing spatial coverage. In general, this trade-off is also affected by the duration of trend period. Steadily increasing monitoring effort in New Zealand over

Page 36 | 155

#### 2000-2020

the last two decades means that shorter and more recent trend periods will generally have a larger number of eligible sites.

The application of filtering rules for variables that are measured at quarterly intervals or more frequently requires two steps. Firstly, sites are retained for observations available for at least X% of the years in the time-period. Secondly, sites are retained for observations available for at least Y% of the sample intervals. For variables that are measured annually such as MCI, the filtering rules are applied by retaining sites for which values are available for at least X% of the years in the trend period.

In this study, filtering rules applied by Larned et al. (2019) were used, which set X and Y to 80%. Further, the definition of seasons was flexible in order to maximise the number of sites that were included. If the site failed to comply with filter rule (2) when seasons were set as months, coarsening of the data to quarterly seasons was applied and the filter rule (2) was reassessed. If the data then complied with filter rule (2), the trend results based on the coarser (i.e., quarterly) seasons were retained for reporting. Bi-months were also included as an intermediate coarseness between months and quarters, as this sampling interval was historically used.

Using these filter rules, the number of site/variable combinations that would be included in the analysis under varying trend period end dates was explored. While the intention was to provide the most recent possible trend assessments (up to the end of the observations dataset, August 2020), the possibility of having an earlier end date was also considered, if that would significantly increase the number of sites that would comply with the filtering rules. End dates were considered at the end of months from December 2019 through to August 2020. The results of this analysis are not included in this report as generally, there was little variation in the number of sites that complied with the filtering rules for end dates between February 2020 and August 2020. The exception was for the macroinvertebrate metrics, which had a large reduction in the number of sites that complied with the filtering rules from the December 2019 cut-off point to all end dates in 2020 (generally a reduction from 26 to 13 sites). This arises due the cessation of several macroinvertebrate sites in 2018. In the interest of providing the most up to date trend assessments, the trends for rivers presented in this study were for 10 and 20-year periods ending at 31 August 2020.

A slightly different approach has been applied to the lake monitoring data in order to maximise the assessment of trends for these sites due the irregularity of the monitoring and changes in monitoring sites. The most recent end date to examine long term, fixed period, trends across all sites was identified. This date coincided with the termination of monitoring at a number of long-term sites at the end of June 2018. We evaluated trends for 10- and 18-year periods up to the end of June 2018. The 18- year period was selected as there were no lake data available prior to 2000. For these fixed period trend assessments, the data were subjected to the same filtering rules as used for the river and groundwater sites.

Another deviation for the trend analysis at lake sites was for a group of sites that were monitored for a period between 2006-2009 after which there was no monitoring until the programme was reestablished in 2018. These sites have been analysed using alternative trend assessment procedures that evaluate the change between the two time-periods (see Section 4.2.11). However, it was important that the data still complied with the time-period requirements relating to representativeness of the time-periods, and that there was no bias toward any particular season in the records. Consequently, the two analysis time-periods for these site/variable combinations to be three complete years: 1 May 2006 to 30 April 2009, and 1 June 2017 to 31 May 2020 were set. It was also required that at least 80% of observations were available in each time-period.

Page 37 | 155

2000-2020

# 4.2.5 Handling censored values

For several water-quality variables, true values are occasionally too low or too high to be measured with precision. These measurements are called censored values. The detection limit is the lowest value that can be measured by an analytical method (either a laboratory measurement or a measurement made in the field) and the reporting limit is the greatest value of a variable that can be measured. Water-quality datasets from New Zealand rivers and lakes often include censorted DRP, TP and NH4N measurements because they are below detection limits, and censored *E. coli* and CLAR measurements because they are above reporting limits.

Censored values are managed in a special way, that is by the non-parametric trend assessment methods. It is therefore important that censored values are correctly identified in the data. Detection limits or reporting limits that have changed through the trend time-period (often due to analytical changes) can induce trends that are associated with the changing precision of the measurements rather than actual changes in the variable. This possibility needs to be accounted for in the trend analysis and this is another reason that it is important that censored values are correctly identified in the data.

A 'hi-censor' filter was applied in the trend assessments to minimise biases that might be introduced due to changes in detection limits through the trend assessment period. The hi-censor filter identifies the highest detection limit for each water quality variable in the trend assessment period and replaces all observations below this level with the highest detection limit and identifies these as censored values. This procedure generally had limited impact on the trend assessment, with the exception of ammoniacal nitrogen, as there was a significant shift in the detection limit, and most of the observations were generally very small (of similar magnitude to the detection limit).

#### 4.2.6 Flow adjustment

Where water quality data are colected in a river and are associated with a solute or particulate matter (e.g., a concentration or an optical measure such as clarity or turbidity) some of the variation can be associated with the river flow (i.e., discharge) at the time the observation was collected. The observed values can vary systematically with flow rate due to two kinds of physical processes. The water quality observations may decrease systematically with increasing flow due to the effect of dilution of the contaminant, or increase with increasing flow due to wash-off of the contaminant (Smith *et al.*, 1996). Different mechanisms may dominate at different sites so that the same water quality variable can exhibit positive or negative relationships with flow. Some water quality variables can be associated with a combination of dilution and wash off with increasing flow. For example, a portion of the *E. coli* load may come from point sources discharges such as sewage treatment plants (dilution effect), but another portion may be derived from surface wash-off. Increasing flow in this situation may result is an initial dilution at the low end of the discharge range, followed by an increase with discharge at higher values of discharge.

Trend analysis seeks to quantify the relationship between the water quality observations and time. In this context, flow can be considered as a covariate; a variable that is also related to the water quality observations but whose influence is confounding water quality – time relationship of interest. Statistical analysis can be used to remove the influence of the covariate on the water quality observations. For river data, this statistical analysis is termed flow adjustment. The same principle can be applied to other types of environment (e.g., lakes, groundwater) and other covariates (e.g., wind, precipitation) and so the general term is covariate adjustment.

Page 38 | 155

#### 2000-2020

Covariate adjustment has two purposes. First, it can increase the statistical power of the trend assessment (i.e., increase the confidence in the estimate of direction and rate of the trend) by removing some of the variability that is associated with the covariate. Second, it removes any component of the trend that can be attributed to a trend in the covariate (e.g., a trend in the flow on sample occasions, such as increasing or decreasing flow with time).

Covariate adjustment involves fitting a model that describes the relationship between the water quality observation and the covariate, and then using the residuals of the model instead of the original water quality observations in the subsequent trend assessment step. In the description of the covariate adjustment method below, flow adjustment was the focus (i.e., removing the influence of flow at from water quality observations made in a river). However, in principle, the method is the same for any other type of covariate adjustment.

Four alternative regression models were considered to describe the relationship between the water quality observations and flow: log-log regression, locally estimated scatterplot smoothing (LOESS, with spans of 0.7 and 0.9) and generalised additive models (GAM). Censored values were represented during model fitting by raw values (i.e., the numeric component of the censored values) multiplied by a 0.5 for detection limit censoring and 1.1 for reporting limit censoring.

The next step was to select the best model from the alternatives tested. Expert judgement was used to choose the most suitable model based considering the following: (1) the homoscedasticity (constant variance) of the regression residuals, (2) model goodness of fit measures and (3) plausibility of the shape of the fitted model. The model goodness of fit measure alone should not be relied on because they can indicate good model performance but describe unrealistic relationships. This is particularly likely when more flexible models are used such as LOESS and GAM models and therefore these models should be used with caution.

When the relationship between flow and the water quality variable was poor, it was concluded that that there was not a systematic relationship between the observations and flow. In this case, no model was selected, no flow adjustment was performed and the trend assessment was performed on the raw data. Choosing not to flow adjust took into consideration the balance between the potential to reduce variance in the observations, and the risk of selecting an implausible/inappropriate model of the relationship between the observations and flow.

#### 4.2.7 Seasonality assessment

For many site/variable combinations, observations vary systematically by season (e.g., by month or quarter). In cases where seasons are a major source in variability, accounting for the systematic seasonal variation should increase the statistical power of the trend assessment (i.e., increase the confidence in the estimate of direction and rate of the trend). The purpose of a seasonality assessment is to identify whether seasons explain variation in the water quality variable. If this is true, then it is appropriate to use the seasonal versions of the trend assessment procedures at the trend assessment step.

Seasonality was evaluated using the Kruskall-Wallis multi-sample test for identical populations. This is a non-parametric ANOVA that determines the extent to which season explains variation in the water quality observations. Following Hirsch *et al.* (1982), site/variable combinations were identified as being seasonal based on the *p*-value from the Kruskall-Wallis test with  $\alpha$ =0.05. For these sites/variable combinations, subsequent trend assessments followed the "seasonal" variants.

Page 39 | 155

#### 2000-2020

The choice of  $\alpha$  is subjective and a value of 0.05 is associated with a very high level of certainty (95%) that the data exhibit a seasonal pattern. In our experience there are generally diminishing differences between the seasonal and non-seasonal trend assessments for *p*-values values larger than 0.05 (Helsel *et al.*, 2020).

# 4.2.8 Analysis of trends

The purpose of trend assessment is to evaluate the direction (i.e., increasing or decreasing) and rate of the change in the central tendency of the observed water quality values over the period of analysis (i.e., the trend). Because the observations represent samples of the water quality over the period of analysis, there is uncertainty about the conclusions drawn from their analysis. Therefore, statistical models are used to determine the direction and rate of the trend and to evaluate the uncertainty of these determinations.

Trends were evaluated using the LWPTrends functions in the R statistical computing software. A brief description of the theoretical basis for these functions is described below.

# 4.2.9 Trend direction assessment

The trend direction and the confidence in the trend direction were evaluated using either the Mann Kendall assessment or the Seasonal Kendall assessment. Although the non-parametric Sen slope regression also provides information about trend direction and its confidence, the Mann Kendall assessment is recommended, rather than Sen slope regression, because the former more robustly handles censored values.

The Mann Kendall assessment requires no *a priori* assumptions about the distribution of the data but does require that the observations are randomly sampled and independent (no serial correlation) and that there is a sample size of  $\geq$  8. Both the Mann Kendall and Seasonal Kendall assessments are based on calculating the Kendall *S* statistic, which is explained diagrammatically in Figure 2.

```
2000-2020
```



Figure 2. The Kendall S statistic is calculated by first evaluating the difference between all pairs of water quality observations (Figure 2, A and B). Positive differences are termed 'concordant' (i.e., the observations increased with increasing time) and negative differences are termed discordant (i.e., the observations decreased with increasing time). The Kendall S statistic is the number of concordant pairs minus the number of discordant pairs (C1). The sign of S indicates the water quality trend direction with a positive or negative sign indicating that observations increased or decreased through time respectively (C2). In the special case that the z score is equal to zero, the trend would be pronounced "indeterminate", or equally likely to be increasing as decreasing.

#### 4.2.10 Assessment of trend rate

The method used to assess trend rate is based on non-parametric Sen slope regressions of water quality observations against time. The Sen slope estimator (SSE; Hirsch *et al.*, 1982) is the slope parameter of a non-parametric regression. SSE is calculated as the median of all possible interobservation slopes (i.e., the difference in the measured observations divided by the time between sample dates; **Error! Reference source not found.**).

Page 41 | 155

2000-2020



#### Figure 3 Pictogram of the calculation of the Sen slope, which is used to characterise trend rate.

The seasonal Sen slope estimator (SSSE) is calculated in two steps. First, for each season, the median of all possible inter-observation slopes is calculated in same manner as shown in Figure 3 but for data pertaining to observations in each individual season. Second, *SSSE* is the median of the seasonal values.

Uncertainty in the assessed trend rate is evaluated following a methodology outlined in Helsel and Hirsch (2002). To calculate the  $100(1-\alpha)$ % two-sided symmetrical confidence interval about the fitted slope parameter, the ranks of the upper and lower confidence limits are determined, and the slopes associated with these observations are applied as the confidence intervals.

The inter-observation slope cannot be definitively calculated between any combination of observations in which either one or both observations comprise censored values. Therefore, it is usual to remove the censor sign from the reported laboratory value and use just the 'raw' numeric component (i.e., <1 becomes 1) multiplied by a factor (such as 0.5 for left-censored and 1.1 for right-censored values). This ensures that in the Sen slope calculations, any left-censored observations are always treated as values that are less than their 'raw' values and right censored observations are always treated as values that are greater than their 'raw' values. As the proportion of censored values increase, the probability that the Sen slope is affected by censoring increases. The outputs from the trend assessment provide an 'analysis note' to identify Sen Slopes where one or both of the observations associated with the median interobservation slope is censored.

# 4.2.11 Evaluating changes in discontinuous data

Some of the monitoring data for lake sites is broken into two distinct time-periods, with a moderate gap (~ 4 years) between periods. Following the USGS guidelines (Helsel et al. 2020), these types of datasets have been analysed using a step change approach. The analysis procedure uses a rank-sum

Page 42 | 155

#### 2000-2020

test (and seasonal variant where appropriate) to test whether there is a change in the observations between the two periods, and the Hodges-Lehman (H-L) estimator to evaluate the magnitude, and direction of the change.

The rank-sum test is implemented in R using the functions: twoSampleLinearRankTest and twoSampleLinearRankTestCensored (both part of the EnvStats package). The second function explicitly accounts for censored data in either or both of the time periods. A seasonal rank sum test is applied when the observations are determined to be seasonal. For the seasonal case, the rank sum statistic and variance are evaluated for each season and these are then individually summed to determine the overall test statistic. As for the Mann Kendall test, the *p*-value from the rank sum test was converted to a measure of the confidence in direction of change in water quality between the two time-periods.

The H-L estimator is evaluated in a similar manner as the Sen Slope, with the exception that rather than evaluating the rate of change between all pairs of observations, only the differences are evaluated, and only between pairs from different periods. The H-L estimator is the median of all possible differences between the data in the before and after periods. A seasonal H-L estimator is evaluated when the observations are determined to be seasonal.

We also provide an estimate of the rate of change that the difference represents, by dividing the H-L estimator by the difference between the mid times of each time period. This measure is indicative only and should only be used as an approximation of the relative magnitude of the rate of change at these sitesonly be used as an approximation of the relative magnitude of the rate of change at these sites.

### 4.2.12 Interpretation of trends

The trend assessment procedure used here facilitates a more nuanced inference than the 'yes/no' output corresponding to the chosen acceptable misclassification error rate. The confidence in direction (*C*) can be transformed into a continuous scale of confidence the trend was decreasing ( $C_d$ ). For all trends with S < 0,  $C_d = C$ , and for all S > 0 a transformation is applied so that  $C_d = 1$ -*C*.  $C_d$  ranges from 0 to 1.0. When  $C_d$  is very small, a decreasing trend is highly unlikely, which because the outcomes are binary, is the same as an increasing trend is highly likely.

The trend for each site/variable combination was assigned a categorical level of confidence that the trend was improving according to its evaluated confidence, direction and the categories shown in Table 2. Improvement is indicated by decreasing trends for all the water quality variables in this study except for MCI, SQMCI, ASPM and dissolved oxygen (for which increasing trends indicate improvement).

Page 43 | 155

#### 2000-2020

Table 2. Level of confidence categories used to convey the confidence that the trend (or step change) indicated improving water quality. The confidence categories are used by the Intergovernmental Panel on Climate Change (IPCC; Stocker et al., 2014).

| Categorical level of confidence trend was decreasing | Descriptor used in report | Value of C <sub>d</sub> (%) |
|--|---------------------------|-----------------------------|
| Virtually certain                                    | <b>↑</b> ↑ <b>↑</b>       | 0.99–1.00                   |
| Extremely likely                                     | <u>↑</u> ↑↑               | 0.95–0.99                   |
| Very likely  | <b>↑</b> ↑                | 0.90–0.95                   |
| Likely   | 1                         | 0.67–0.90                   |
| About as likely as not                               | <b>↔</b>                  | 0.33–0.67                   |
| Unlikely   | Ļ                         | 0.10-0.33                   |
| Very unlikely  | <b>↓</b> ↓                | 0.05-0.10                   |
| Extremely unlikely                                   | 111                       | 0.01-0.05                   |
| Exceptionally unlikely                               | 1111                      | 0.0-0.01                    |

Outputs from the trend analyses were also classified into four direction categories: improving, degrading, indeterminate, and not analysed. An increasing or decreasing trend category was assigned based on the sign of the S statistic from the Mann Kendall test. An indeterminate trend category was assigned when the Z score equalled zeros. Trends were classified as "not analysed" for two reasons:

- When a large proportion of the values were censored (data has <5 non-censored values and/or <3 unique non-censored values). This arises because trend analysis is based on examining differences in the value of the variable under consideration between all pairs of sample occasions. When a value is censored, it cannot be compared with any other value and the comparison is treated as a "tie" (i.e., there is no change in the variable between the two sample occasions). When there are many ties there is little information content in the data and a meaningful statistic cannot be calculated.
- 2) When there is no, or very little, variation in the data because this also results in ties. This can occur because laboratory analysis of some variables has low precision (i.e., values have few or no significant figures). In this case, many samples have the same value, and this then results in ties.

Changes for discontinuous data were classified as "not analysed" when there were less than 3 unique observations in the entire record, or if seasonal, within any season.

Page 44 | 155

2000-2020

# 4.2.13 River data availability

Following the application of the filtering rules, the total number of sites that were included in the analyses was reduced, a summary of the site numbers that were included in the final trend assessment is presented in Table 3. Confidence that the trend direction indicated improving water quality, was mapped for the raw (with high censor filter) for the 10 and 20-year trend periods in Figure 54 and Figure 55.

Table 3 River water quality variables, measurement units and site numbers for which 10- and 20-year trends (Raw, and Flow Adjusted FA) were analysed by this study.

| Variable                      | Number<br>of sites | Number of sit<br>complied with<br>rules (10-year | es that<br>n filtering<br>s) | Number of sites that<br>complied with filtering<br>rules (20-years) |    |  |
|-------------------------------|--------------------|--|------------------------------|---|----|--|
|                               |                    | Raw  | FA                           | Raw   | FA |  |
| Ammoniacal Nitrogen           | 114                | 50   | 32                           | 34  | 18 |  |
| ASPM                          | 51                 | 10   | 6                            | 0   | 0  |  |
| Chlorophyll a                 | 44                 | 0  | 0                            | 0   | 0  |  |
| Dissolved Inorganic Nitrogen  | 108                | 0  | 0                            | 0   | 0  |  |
| Dissolved Reactive Phosphorus | 108                | 50   | 32                           | 33  | 18 |  |
| E. coli                       | 114                | 50   | 27                           | 28  | 13 |  |
| MCI                           | 54                 | 13   | 7                            | 0   | 0  |  |
| Nitrite/Nitrate Nitrogen      | 114                | 50   | 32                           | 34  | 18 |  |
| SQMCI Score                   | 53                 | 13   | 7                            | 0   | 0  |  |
| Total Nitrogen                | 114                | 50   | 32                           | 33  | 18 |  |
| Total Phosphorus              | 114                | 50   | 32                           | 32  | 18 |  |
| Turbidity                     | 114                | 50   | 32                           | 32  | 18 |  |

Page 45 | 155

#### 2000-2020

# 5 Results Overview

In Otago five FMUs have been recognised: Clutha/Mata-Au, Taieri, North Otago, Dunedin Coastal and Catlins. The Clutha/Mata-Au FMU has been divided in to five sub-areas, or Rohe comprising the Upper Lakes Rohe, Dunstan Rohe, Manuherekia Rohe, Roxburgh Rohe and Lower Clutha Rohe. The results section is organised in FMU or Rohe chapters, each chapter follows the same format:

- Figure of the FMU/Rohe showing location of water quality monitoring sites
- State analysis results of grading of river/lake sites based on the NOF criteria. Including a comparison with the baseline state (2012-2017)
- State analysis maps showing results of river/lake sites coloured according to their state grading as indicated by NOF attribute bands.
- Trend analysis summary of sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement.

At the end of each FMU/Rohe section a summary table is presented showing which sites and attributes fall below the national bottom line.

2000-2020



# 6 Upper Lakes Rohe (Clutha Matu/Au FMU)

Figure 4 Location of water quality monitoring sites in the Upper Lakes Rohe

Page 47 | 155

2000-2020

# 6.1 State Analysis Results

The results of grading the SoE sites in the Upper Lakes Rohe according to the NPSFM 2020 NOF criteria are summarised in Figure 5**Error! Reference source not found.** and Figure 6 and mapped in Figure 7. Many sites in the Upper Lakes Rohe did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Figure 5 Grading of the river sites of the Upper Clutha Rohe based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.

Page 48 | 155

2000-2020



Figure 6 Grading of the lake sites of the Upper Clutha Rohe based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.

Page 49 | 155

2000-2020



Figure 7 Maps showing Upper Clutha Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

Page 50 | 155

2000-2020

# 6.1.1 Periphyton and Nutrients

Results for the river periphyton trophic state are shown in Figure 5 (periphyton). Most sites in the Upper Clutha are in attribute band 'A' for periphyton as few results exceed 50 chl- $a/m^2$  reflecting negligible nutrient enrichment. Bullock Creek, a spring fed stream that runs through Wanaka township has an interim result of 'D' which places it below the national bottom line, this reflects a higher nutrient enrichment and the possibility of regular nuisance blooms.

Figure 5 shows the MfE (2020) DRP and TN concentrations to manage the NPSFM 2020 periphyton attribute state (periphyton DRP and periphyton TN). All sites have median DRP concentrations that place them in the 'A' or 'B' bands, however 12 Mile Creek and the Matukituki have median DRP concentrations that place them in the 'C' band. For TN median concentrations in the Upper Lakes Rohe are generally in the T50 mg chl-a/m<sup>2</sup> band (or band 'A'), as would be expected in a low nutrient environment. The outliers are the urban stream Horn Creek, which has a median TN concentration of 0.27 mg/l placing it in band 'C' and 12 Mile Creek and Bullock Creek that have median TN concentrations placing them in band 'B'.

Figure 5 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Upper Lakes Rohe show that every site achieves a band 'A', other than Horn Creek which achieves a band 'B' (median DRP). The NPSFM 2020 describes Band 'A' as 'Ecological communities and ecosystem processes are similar to those of natural reference conditions'.

Results for the lakes are also shown in Figure 6. Trophic status is a common method for describing the health of lakes and an indicator of how much growth or productivity occurs in the lake, productivity being directly related to the availability of nutrients (ORC, 2017). Lakes in pristine condition typically have very low nutrient and algal biomass levels. As lakes become more enriched due to changes in land-use and land management practices, lake nutrient levels and algal productivity increases. The NPSFM 2020 describes how phytoplankton affects lake ecological communities. If phytoplankton is in the 'A' band, then 'Lake ecological communities are healthy and resilient, similar to natural reference conditions'. Error! Reference source not found. Error! Reference source not found. shows that this is the case for all the lake sites in the Upper Clutha Rohe. The results for total nitrogen and total phosphorus are also shown in Error! Reference source not found., all results are in the 'A' band reflecting low levels of total nutrients indicating that associated ecological communities are healthy and resilient.

# Toxicants

NOF attribute bands for NH4-N and nitrate (measured as NNN) toxicity (Figure 5) show excellent protection levels against toxicity risk for all Upper Lakes Rohe river and lake SoE monitoring sites, with all sites returning an 'A' band (highest level of protection) for NH4-N; and all sites returning an 'A' band for NNN.

# 6.1.2 Suspended fine sediment (Rivers)

The clarity results for the Upper Lakes Rohe are shown in Figure 5. Sites that have a high degree of glacial flour present in the river, such as the Dart and Rees rivers return some very high turbidity (and suspended sediment) levels despite the rivers being close to natural state. The Buckler Burn (Glenorchy) Invincible Creek (Rees Valley), Ox Burn (Rees Valley) and Timaru Creek (Hawea) are not glacial fed, but have very steep catchments which would be prone to erosion. Horn Creek is the other

Page 51 | 155

#### 2000-2020

stream to have turbidity below the national bottom line, this is an urban stream in the heart of Queenstown and is likely affected by stormwater as well as avian wildlife stirring up the creek bed.

#### 6.1.3 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state of water quality and aquatic ecosystem health at a site. Figure 5 summarises MCI scores for sites monitored for aquatic macroinvertebrates throughout the Upper Lakes Rohe.

The MCI interim scores are somewhat comparable across sites with seven of the nine monitored sites returning comparable MCI scores above 120; reflecting a macroinvertebrate community in 'good' condition. The exceptions are Bullock Creek and Motatapu which return MCI scores between 100 and 110; reflecting moderate nutrient enrichment. In a regional context the Upper Clutha Rohe generally show overall water and habitat quality supports the existence of healthy invertebrate communities. The ASPM interim scores shown in **Error! Reference source not found.** reflect those of the MCI scores for Bullock Creek, but Precipice Creek, despite having a B grade for MCI, obtains a 'C' grade for ASPM, which the NPSFM 2020 describes as 'a community with moderate to severe loss of ecological integrity'.

# 6.1.4 Human health for recreation

Figure 5 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

Compliance for rivers is generally excellent across in the Upper Lakes Rohe, with all sites other than Bullock Creek, Makoroa, Quartz Creek and the Rees returning bacterial water quality above the national bottom line. Bullock Creek has high background bacteria concentrations which indicates an *E. coli* source that is affecting water quality even under low flow conditions, such as water fowl often being present in the stream above the sampling location.

For the lakes, compliance is excellent across in the Upper Lakes Rohe, with all sites at all statistical tests obtaining an 'A' band.

Page 52 | 155

2000-2020

# 6.2 Trend Analysis

Trend analysis results for the Upper Lakes Rohe is shown in Figure 8.



Figure 8 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells

Page 53 | 155

#### 2000-2020

containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the prevision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend.

Results show that over a 10-year period the Dart at the Hillocks returned a 'virtually certain' (improving) trend for TP, but over the same period the Matukituki returned an 'exceptionally unlikely' improving trend for NNN.

Trend analysis for the Upper Lakes Rohe lakes is shown in Figure 8. NNN is 'virtually certain' to be improving from the outlets at Wakatipu, Wanaka and Hawea (over 18 years) and Wakatipu (over 10-years), turbidity is 'exceptionally unlikely' or 'extremely unlikely' to be improving at all three outflow sites (over 18 years).

For discontinuous data sets the two time-periods analysed were three complete years: 1 May 2006 to 30 April 2009, and 1 June 2017 to 31 May 2020. Most of the trends analysed were influenced by censored values (where true values are too low to be measured with precision) and have not been included. All sites had 'virtually certain' improving trends for TN as shown in Figure 8

# 6.3 Water quality summary and discussion: Upper lakes Rohe

The tables in this section summarise:

- 1) River and lake attribute states where the national bottom line is not met (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) All trends using raw data for rivers and continuous data for lakes over the two timeperiods.

Table 4 Summary of river state, where state does not meet the national bottom line cells are coloured red. No lake variables were below the national bottom line.

| sID              | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Timaru Creek     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Rees River       |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Quartz Creek     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Ox Burn          |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Makarora River   |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Invincible Creek |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Horn Creek       |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Dart River       |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Bullock Creek    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Buckler Burn     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Page 54 | 155

#### 2000-2020

Table 5 Summary of river sites (raw data) where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

| npID   | nObs | Freq    | Period     | AnnualSenSlope | DirectionConf          | Descriptor  |  |  |  |  |  |
|--|------|---------|------------|----------------|------------------------|-------------|--|--|--|--|--|
| Dart at The Hillocks   |      |         |            |                |                        |             |  |  |  |  |  |
| Total Nitrogen     57     BiMonth     10     0     Virtually certain |      |         |            |                |                        |             |  |  |  |  |  |
| Total Phosphorus   | 57   | BiMonth | 10         | -0.0037        | Virtually certain      | <u> </u>    |  |  |  |  |  |
|  |      | Matu    | ikituki at | West Wanaka    |                        |             |  |  |  |  |  |
| Ammoniacal Nitrogen  | 101  | Month   | 10         | 0              | Extremely likely       | <b>↑</b> ↑↑ |  |  |  |  |  |
| Nitrite/Nitrate N  | 98   | Month   | 10         | 0.0015         | Exceptionally unlikely | 1111        |  |  |  |  |  |
| Total Phosphorus   | 101  | Month   | 10         | -0.0008        | Virtually certain      | 1111        |  |  |  |  |  |

Table 6 Summary lake sites where trends (continuous data) are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not** found.

|                            |                                       |         |           | Annual           | Trend          | Direction              |            |  |  |  |  |
|----------------------------|---------------------------------------|---------|-----------|------------------|----------------|------------------------|------------|--|--|--|--|
| npID                       | nObs                                  | Freq    | Period    | SenSlope         | Direction      | Conf                   | Descriptor |  |  |  |  |
|                            |                                       | Lal     | ke Wakat  | ipu at Outflow 1 | 0-year dataset | •                      |            |  |  |  |  |
| Ammoniacal N               | 105                                   | BiMonth | 18        | 0                | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Nitrite/Nitrate N          | 58                                    | BiMonth | 10        | 0.0008           | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Nitrite/Nitrate N          | 105                                   | BiMonth | 18        | 0.0003           | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Turbidity                  | 104                                   | BiMonth | 18        | 0.0106           | Increasing     | Extremely unlikely     | 111        |  |  |  |  |
|                            |                                       | L       | ake Wana  | aka at Outlet 10 | -year dataset  |                        |            |  |  |  |  |
| Ammoniacal N               | 60                                    | BiMonth | 10        | 0                | Increasing     | Exceptionally unlikely | 1111       |  |  |  |  |
| Nitrite/Nitrate N          | 105                                   | BiMonth | 18        | 0.0008           | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Turbidity                  | 105                                   | BiMonth | 18        | 0.0065           | Increasing     | Extremely unlikely     | 111        |  |  |  |  |
| Ammoniacal N 60 BiMonth 10 |                                       |         |           | 0                | Increasing     | Exceptionally unlikely | 1111       |  |  |  |  |
|                            |                                       | Lake    | e Hawea ( | Outflow at Dam   | 10-year datase | t                      |            |  |  |  |  |
| Nitrite/Nitrate N          | 106                                   | BiMonth | 18        | 0                | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Turbidity                  | 104                                   | BiMonth | 18        | 0.0120           | Increasing     | Exceptionally unlikely | 1111       |  |  |  |  |
|                            |                                       | Lal     | ke Wakat  | ipu at Outflow 1 | 8-year dataset |                        |            |  |  |  |  |
| Ammoniacal N               | 105                                   | BiMonth | 18        | 0                | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Nitrite/Nitrate N          | 58                                    | BiMonth | 10        | 0.0008           | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Nitrite/Nitrate N          | 105                                   | BiMonth | 18        | 0.0003           | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Turbidity                  | 104                                   | BiMonth | 18        | 0.0106           | Increasing     | Extremely unlikely     | 111        |  |  |  |  |
|                            | Lake Wanaka at Outlet 18-year dataset |         |           |                  |                |                        |            |  |  |  |  |
| Ammoniacal N               | 60                                    | BiMonth | 10        | 0                | Increasing     | Exceptionally unlikely | 1111       |  |  |  |  |
| Nitrite/Nitrate N          | 105                                   | BiMonth | 18        | 0.0008           | Decreasing     | Virtually certain      | 1111       |  |  |  |  |
| Turbidity                  | 105                                   | BiMonth | 18        | 0.0065           | Increasing     | Extremely unlikely     | 111        |  |  |  |  |

Table 7 Summary of trends using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in *Error! Reference source not found.*.

|--|

Page 55 | 155

#### 2000-2020

| Descriptor             | 1111 | 111 | <b>1</b> 1 | Ŷ | ↔ | Ļ | 11 | 111 | 1111 |
|------------------------|------|-----|------------|---|---|---|----|-----|------|
| Rivers - 10-year trend | 3    | 1   | 2          | 1 |   |   |    |     | 1    |
| Rivers - 20-year trend |      |     |            |   |   |   |    |     |      |
| Lakes – 10-year trend  | 1    |     | 1          | 1 |   | 3 | 1  |     | 1    |
| Lakes – 18 year trend  | 4    | 1   |            | 3 |   | 2 |    | 2   | 1    |

Water quality is best at river and stream reaches located at high or mountainous elevations under predominantly native cover. These sites tend to be associated with the Upper Lakes Rohe.

In the Upper Lakes Rohe water quality is generally very good, as it should be. Of the NOF attribute state, the Rees and Dart are below the national bottom line for turbidity, but this is a natural state due to glacial melt-water. Despite very good compliance with the *E.coli* attribute, four sites are below the national bottom line; Bullock Creek is an urban stream running through the Wanaka township with a large population of wildfowl, these and stormwater are likely to be the reason behind the poor grade. The Makarora and Rees are large rivers, with catchment areas largely in national parks. It is likely that the poor water quality is very localised, and possibly reflects the quality of tributaries running over the flats. The same can be said for Quartz Creek that runs over the flats before entering Lake Wanaka. The only other 'D' band was for periphyton in Bullock Creek. The lakes achieved attribute 'A' bands for all parameters assessed.

For trends, only the Dart and Matukituki have been monitored for a sufficiently long time-period for trend analysis to be undertaken. Perhaps the most worrying trend is NNN has shown an increase over the last 10-years in the Matukituki. Trend analysis in the lakes shows that TN has 'virtually certain' improving trends in the outlets of both Wakatipu and Wanaka, however turbidity is almost certainly degrading in Hawea and Wanaka.

Table 6 and Table 7 only show sites with 99%, 95%, 1% and 5% confidence levels. These equate to the 'virtually certain', 'extremely likely', 'exceptionally unlikely' and extremely unlikely' categories. It is important to note when sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction, at these sites the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Upper Lakes Rohe, there are a lot of sites with a zero sen slope, highlighting how low the concentrations of most analytes are in this region.

As stated previously, having accurate information on changes in land use and land management practice would help in identifying drivers of change evident with some water quality variables. The reasons for this are unclear as ORC do not collect any information on changes in land use or land management practices that would allow for confident assessment of drivers of turbidity or NNN in our waterways.

In summary, for the majority of sites across the Upper Clutha reporting region, water quality is excellent and the best in Otago:

- All sites return an 'A' band for the toxicity attribute states of ammonia and nitrate.
- Nutrient concentrations stimulating algae growth low, other than in Bullock Creek where periphyton (chlorophyll *a*) is below the national bottom line.
- Bacterial water quality is excellent across all sites, with the exception of Bullock Creek, the Rees, the Makaroa and Quartz Creek; which are below the national bottom line.
- Clarity (suspended sediment) is only elevated in rivers fed by glaciers.

Page 56 | 155

#### 2000-2020

- Macroinvertebrate scores reflect the low nutrient and scarce food source (periphyton) content of rivers with scores in the 'B' or 'C' band.
- Trend analysis for rivers showed an 'exceptionally unlikely' improving for NNN in the Matukituki River and 'virtually certain' improving trend for TP in the Dart and Matukituki.
- Trend analysis for lakes showed the outlets of Lakes Hawea, Wanaka and Wakatipu had 'exceptionally unlikely' or 'extremely unlikely' improving trends for turbidity, but 'virtually certa certain' improving trends for NNN.

Page 57 | 155

2000-2020



# 7 Dunstan Rohe (Clutha Matu/Au FMU)

Figure 9 Location of water quality monitoring sites in the Dunstan Rohe

Page 58 | 155

2000-2020

# 7.1 State Analysis Results

The results of grading the SoE sites in the Dunstan Rohe according to the NPSFM 2020 NOF criteria are summarised in Figure 10 and Figure 11 and mapped in Figure 12. Many sites in the Dusntan Rohe did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Figure 10 Grading of the river sites of the Dunstan Rohe based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 59 | 155



# Figure 11 Grading of the lake sites of the Dunstan Rohe based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.

## Otago Water Quality: Rivers and Lakes

Page 60 | 155

2000-2020

2000-2020



Figure 12 Maps showing Dunstan Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

Page 61 | 155

2000-2020

#### 7.1.1 Periphyton and Nutrients

Results for the river periphyton trophic state results are shown in Figure 12 (periphyton). The Dunstan Rohe returns a mixed bag of attribute bands for periphyton, the Arrow and Cardrona have an interim 'A' band as few results exceed 50 chl-*a*/m<sup>2</sup> meaning that blooms would be rare, reflecting negligible nutrient enrichment. The Lindis at Lindis Peak and Luggate Creek record interim 'B' bands and the Lindis at Ardgour Road, towards the bottom of the Lindis catchment has an interim result of 'D', below the national bottom line, this reflects a higher nutrient enrichment and the possibility of regular nuisance blooms

Figure 12 shows the MfE (2020) DRP and TN concentrations to manage the NPSFM 2020 periphyton attribute state (periphyton DRP and periphyton TN). Using the 20% exceedance criteria (mid-range), the TN median concentrations in the Dunstan Rohe are generally in the T120 mg chl-a/m2 band (or band 'B'), as would be expected in a low nutrient environment. The outliers are Hayes Creek, which is influenced by high nutrient concentrations in Lake Hayes, the Kawarau River and Mill Creek all of which are placed in band 'C'. The Hawea River, Nevis and Luggate Creek have low median TN concentrations and are graded as band 'B'. Median cocentrations of DRP place all sites in the 'B' or 'C' band.

Figure 12 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Dunstan Rohe show that every site achieves a band 'A', other than Luggate Creek and the Clutha at Luggate Bride which achieve a band 'B' and the noticeable outlier, Hayes Creek which achieves band 'C', due to receiving nutrient rich water from Lake Hayes. The NPSFM 2020 describes Band 'C' as 'Ecological ommunities are impacted by moderate DRP elevation above natural reference conditions. If other conditions also favour eutrophication, DRP enrichment may cause increased algal and plant growth, loss of sensitive macro-invertebrate and fish taxa, and high rates of respiration and decay'.

Results for the lakes are also shown in Figure 12. Trophic status is a common method for describing the health of lakes and an indicator of how much growth or productivity occurs in the lake, productivity being directly related to the availability of nutrients (ORC, 2017). Lakes in pristine condition typically have very low nutrient and algal biomass levels. As lakes become more enriched due to changes in land-use and land management practices, lake nutrient levels and algal productivity increases. The NPSFM 2020 describes how phytoplankton affects lake ecological communities. If the chlorophyll a concentration are in the 'A' band, then 'Lake ecological communities are healthy and resilient, similar to natural reference conditions'. Error! Reference source not found. Figure 12 shows that this is the case forLake Dunstan, but Lake Hayes (10m) and Lake Johnson achieve a 'D' band and Lake Hayes (Bendemeer Bay) achieves a 'C' band. The results for total nitrogen and total phosphorus are also shown in Figure 12, Lake Dunstan achieves 'A' bands reflecting low levels of total nutrients indicating that associated ecological communities are healthy and resilient, but Lake Hayes have higher concentrations of TN and TP. Lake Johnson falls below the national bottom line for TN achieving a 'D' band.

Page 62 | 155

2000-2020

# 7.1.2 Toxicants

NOF attribute bands for NH4-N and nitrate (measured as NNN) toxicity are shown in Figure 12, the results for rivers show excellent protection levels against toxicity risk for nearly all Dunstan Rohe SoE monitoring sites, with all sites bar one (Nevis) returning an 'A' band for NH4-N and all sites returning an 'A' band (highest level of protection) for NNN. For lakes, Lake Johnson and Lake Hayes return 'C' bands and Lake Dunstan an 'A' band.

## 7.1.3 Suspended fine sediment (Rivers)

The clarity results for the Dunstan Rohe are shown in Figure 12. Six sites return an 'A' band which the NPSFM 2020 describes as having 'minimal impact of suspended sediment on instream biota. Ecological communities are similar to those observed in natural reference conditions' (NPSFM, 2020). Five sites return a 'D' band; the Upper Cardrona, Quartz Creek, Mill Creek, Lindis at Lindis Peak and the Clutha at Luggate.

## 7.1.4 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state of water quality and aquatic ecosystem health at a site. Figure 12 summarises MCI scores for sites monitored for aquatic macroinvertebrates throughout the Dunstan Rohe.

The MCI interim scores are somewhat comparable across sites with three of the five monitored sites returning MCI scores between 100 and 110; reflecting a macroinvertebrate community indicative of moderate organic pollution or nutrient enrichment. The exceptions are the Arrow River which returns a MCI score of 119 reflecting 'good' water quality and Mill Creek which returns an MCI score of 85, which is below the national bottom line. The poor score at Mill Creek would be indicative of severe organic pollution or nutrient enrichment as well as poor habitat quality.

The ASPM interim scores shown in Figure 12 reflect those of the MCI scores for the Arrow River and Mill Creek, but the other sites despite having a 'C' grade for MCI, obtain a 'B' grade for ASPM, which the NPSFM 2020 describes as 'a community with mild to moderate loss of ecological integrity'.

# 7.1.5 Human health for recreation (Rivers)

Figure 12 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

Compliance is generally excellent across in the Dunstan Rohe, with all sites other than the Kawarau and the Upper Cardrona having bacterial water quality above the national bottom line.

Figure 10 shows that many of the sites have fewer than the required 60 samples over a maximum of five years, so the grades are interim. For example, the Upper Cardrona returns 'A' grades for all statistical tests bar the 95<sup>th</sup> percentile, as it only has 21 samples over 3 years. It is unknown what this upper catchment site would return as a 95<sup>th</sup> percentile over required the time-period. The Kawarau also does not meet minimum sample requirements, but does returns of the four statistical tests, two return a 'C' grade, one a 'B' grade and one a 'D' grade.

For lakes, Figure 11 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. Compliance is excellent across all lakes in the Dunstan Rohe, only Lake Hayes at Bendermeer Bay achieves a 'B' band, all other sites at all statistical tests obtain an 'A' grade.

Page 63 | 155

2000-2020

# 7.2 Trend Analysis

Trend analysis results for the Dunstan Rohe is shown in Figure 13



Page 64 | 155

#### 2000-2020

Figure 13 Summary of Dunstan Rohe sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend.

Trend analysis for both rivers and lakes show that most of the trends analysed were influenced by censored values, where true values are too low to be measured with precision, this is shown by the black dot in Figure 13. Over a 10-year time-period the Cardrona has the four variables (E.coli, NNN, TN and SQMCI) that shows 'exceptionally unlikely' or 'extremely unlikely' improving trends. Over the same time-period, Luggate Creek has five 'very unlikey' or 'extremely unlikely' improving trends and the Lindis at Ardgour Road has three 'unlikely' or 'very unlikey' improving trends. Over a 20-year time-period, the Cardrona shows the same 'exceptionally unlikely' or 'extremely unlikely' improving trends for TN and NNN, and Mill Creek and the Kawarau have an 'exceptionally unlikely' improving trend for turbidity.

Trends for the lake data ranged across the board from 'virtually certain' to 'exceptionally unlikely' improvements in most variables. The most notable trends are that over 10-years the lake data shows one 'exceptionally unlikely' improving trend for NH4-N at Lake Dunstan and that Lake Hayes has 'virtually certain' improving trends for DRP and TP. The step dataset shows that secchi depth was 'exceptionally unlikely' to be improving in Lake Hayes.

# 7.3 Water quality summary and discussion: Dunstan Rohe

The tables in this section summarise:

- 1) River and lake sites where attributes where the national bottom line is not met (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) All trends using raw data for rivers and continuous data for lakes over the two time-periods.

Table 8 Summary of river and lake state, red cells show where state does not meet the national bottom line in one or more variable

| sID               | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|-------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Upper Cardrona    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Quartz Reef Creek |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Mill Creek        |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Lindis at Peak    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Kawarau at Chards |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Clutha at Luggate |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Page 65 | 155

# 2000-2020

|                                  | NH4-N<br>max | NH4-N<br>median | Chlorophyll a<br>-max | Chlorophyll a<br>median | E.coli<br>Overall | ΤN | ΤP |
|----------------------------------|--------------|-----------------|-----------------------|-------------------------|-------------------|----|----|
| Lake Johnson at South Beach huts |              |                 |                       |                         |                   |    |    |
| Lake Hayes at Mid Lake 10m       |              |                 |                       |                         |                   |    |    |

Page 66 | 155

# 2000-2020

# Table 9 Summary of river sites where trends are greater than 'likely' or 'unlikely' (raw data). Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**

|                      |       |         |         | Annual        |                |                        |                                       |
|----------------------|-------|---------|---------|---------------|----------------|------------------------|---------------------------------------|
| nnID                 | nObs  | Freq    | Period  | SonSlono      | TrendDirection | DirectionConf          | Descriptor                            |
|                      | 11003 | ineq    | renou   | Sensiope      | Trendbirection | Directioncom           | Descriptor                            |
|                      |       |         | Card    | lrona at Mt l | Barker         |                        |                                       |
| Ammoniacal Nitrogen  | 119   | BiMonth | 20      | 0             | Decreasing     | Extremely likely       | 111                                   |
| Dissolved Reactive P | 99    | Month   | 10      | 0             | Decreasing     | Virtually certain      | <u> </u>                              |
| Dissolved Reactive P | 115   | BiMonth | 20      | 0             | Decreasing     | Virtually certain      | <u> </u>                              |
| E. coli              | 100   | Month   | 10      | 3.1637        | Increasing     | Exceptionally unlikely | 1111                                  |
| Nitrite/Nitrate N    | 100   | Month   | 10      | 0.0050        | Increasing     | Exceptionally unlikely | $\uparrow\uparrow\uparrow\uparrow$    |
| Nitrite/Nitrate N    | 118   | BiMonth | 20      | 0.0016        | Increasing     | Exceptionally unlikely | $\uparrow \uparrow \uparrow \uparrow$ |
| SQMCI Score          | 9     | Year    | 10      | -0.4844       | Decreasing     | Exceptionally unlikely | 1111                                  |
| Total Nitrogen       | 99    | Month   | 10      | 0.0019        | Increasing     | Extremely unlikely     | 111                                   |
| Total Nitrogen       | 118   | BiMonth | 20      | 0             | Increasing     | Exceptionally unlikely | 1111                                  |
|                      |       |         | Hawe    | a at Camphi   | ll Bridge      |                        |                                       |
| Nitrite/Nitrate N    | 100   | Month   | 10      | 0.0002        | Increasing     | Extremely unlikely     | 111                                   |
| Total Phosphorus     | 99    | Month   | 10      | 0             | Decreasing     | Extremely likely       | 111                                   |
| Turbidity            | 100   | Month   | 10      | -0.0122       | Decreasing     | Extremely likely       | 111                                   |
|                      |       |         | Kaw     | arau @ Cha    | rds Rd         |                        |                                       |
| Nitrite/Nitrate N    | 98    | Month   | 10      | -0.0009       | Decreasing     | Virtually certain      | 1111                                  |
| Nitrite/Nitrate N    | 218   | Month   | 20      | -0.0004       | Decreasing     | Virtually certain      | 1111                                  |
| Turbidity            | 218   | Month   | 20      | 0.04331       | Increasing     | Exceptionally unlikely | 1111                                  |
|                      |       |         | Lind    | is at Ardgou  | r Road         |                        |                                       |
| Nitrite/Nitrate N    | 105   | Month   | 10      | -0.0055       | Decreasing     | Virtually certain      | 1111                                  |
| Total Nitrogen       | 104   | Month   | 10      | -0.0108       | Decreasing     | Virtually certain      | 1111                                  |
| Total Phosphorus     | 103   | Month   | 10      | -0.0005       | Decreasing     | Extremely likely       | 111                                   |
|                      |       |         | Lin     | dis at Lindis | Peak           |                        |                                       |
| Dissolved Reactive P | 98    | BiMonth | 20      | 0             | Decreasing     | Virtually certain      | 1111                                  |
| E. coli              | 102   | BiMonth | 20      | 0.5339        | Increasing     | Extremely unlikely     | ↓↓↓                                   |
| Total Phosphorus     | 58    | BiMonth | 10      | -0.0006       | Decreasing     | Extremely likely       | 111                                   |
|                      | 1     | 1       | Luggate | e Creek at SH | 16 Bridge      | 1                      | 1                                     |
| Dissolved Reactive P | 58    | BiMonth | 10      | -0.0002       | Decreasing     | Extremely likely       | 111                                   |
| E. coli              | 100   | Month   | 10      | 1.5020        | Increasing     | Extremely unlikely     | 111                                   |
| Nitrite/Nitrate N    | 41    | Qtr     | 10      | 0             | Increasing     | Extremely unlikely     | 111                                   |
| Turbidity            | 100   | Month   | 10      | 0.0429        | Increasing     | Exceptionally unlikely | 1111                                  |
|                      | 1     |         | Mill    | Creek at Fis  | h Trap         | Γ                      | T                                     |
| Ammoniacal Nitrogen  | 79    | Qtr     | 20      | 0             | Decreasing     | Virtually certain      | 1111                                  |
| Dissolved Reactive P | 98    | Month   | 10      | -0.0004       | Decreasing     | Virtually certain      | 1111                                  |
| E. coli              | 98    | Month   | 10      | -3.7270       | Decreasing     | Extremely likely       | 111                                   |
| Nitrite/Nitrate N    | 116   | BiMonth | 20      | -0.0028       | Decreasing     | Extremely likely       | 111                                   |
| Total Nitrogen       | 98    | Month   | 10      | -0.0034       | Decreasing     | Very likely            | <b>†</b> †                            |
| Total Nitrogen       | 116   | BiMonth | 20      | -0.0066       | Decreasing     | Virtually certain      | 1111                                  |
| Total Phosphorus     | 99    | Month   | 10      | -0.0006       | Decreasing     | Extremely likely       | 111                                   |
| Turbidity            | 99    | Month   | 10      | 0.1171        | Increasing     | Extremely unlikely     | 111                                   |
| Turbidity            | 115   | BiMonth | 20      | 0.0671        | Increasing     | Exceptionally unlikely | 1111                                  |
|                      |       |         | Nevis a | t Wentwort    | h Station      |                        |                                       |
| Dissolved Reactive P | 35    | Qtr     | 10      | -0.0005       | Decreasing     | Virtually certain      | 1111                                  |
| Total Phosphorus     | 35    | Qtr     | 10      | -0.0004       | Decreasing     | Extremely likely       | 111                                   |

Page 67 | 155

#### 2000-2020

# Table 10 Summary lake sites where trends (continuous data) are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**

|                      |      |         |           | Annual        | Trend      | Direction          |            |
|----------------------|------|---------|-----------|---------------|------------|--------------------|------------|
| npID                 | nObs | Freq    | Period    | SenSlope      | Direction  | Conf               | Descriptor |
|                      |      | Lak     | ke Dunsta | in at Dead Ma | ns Point   |                    |            |
| Ammoniacal N         | 60   | BiMonth | 10        | 0             | Increasing | Extremely unlikely | 111        |
| Nitrite/Nitrate N    | 107  | BiMonth | 18        | -0.0006       | Decreasing | Virtually certain  | 1111       |
|                      |      | L       | ake Haye  | s at Bendeme  | er Bay     |                    |            |
| Dissolved Reactive P | 48   | BiMonth | 10        | -0.0010       | Decreasing | Virtually certain  | 1111       |
| Total Phosphorus     | 48   | BiMonth | 10        | -0.0018       | Decreasing | Virtually certain  | 1111       |

 Table 11 Overall summary of trends for the Dunstan Rohe using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in Error!

 Reference source not found..

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descsriptor            | 1111                 | <b>↑</b> ↑↑         | <b>↑</b> ↑  | Ŷ      | ÷                   | Ļ        | ΥĻ               | 111                   | 1111                      |
| Rivers - 10-year trend | 6                    | 8                   | 3           | 12     | 7                   | 7        | 4                | 5                     | 4                         |
| Rivers - 20-year trend | 5                    | 2                   |             | 4      | 6                   | 3        |                  | 1                     | 4                         |
| Lakes – 10-year trend  | 1                    |                     | 3           | 2      | 3                   | 2        | 1                | 3                     |                           |
| Lakes – 20-year trend  | 1                    |                     | 1           |        | 5                   |          |                  |                       |                           |

In the Dunstan Rohe compliance with NPSFM 2020 NOF attribute states is generally very good. Table 8 shows that few sites have variables with NOF bands below the national bottom line.

In Table 9 and Table 10 only sites with 99%, 95% ,1% and 5% confidence levels are shown. These equate to the 'virtually certain', 'extremely likely', 'exceptionally unlikely' and extremely unlikely' categories. It is important to note when sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction, at these sites the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Dunstan Rohe, these sites include NH4-N at Lake Dunstan, Cardrona and Mill Creek; DRP at Cardrona and Lindis at Peak; TN at Cardrona; TP at Hawea and NNN at Luggate Creek.

The Cardrona River had the most disturbing amount of 'exceptionally unlikely' or 'extremely unlikely' improving trends, including *E. coli*, TN, NNN and SQMCI. The NNN may be linked to slightly enriched groundwater contributing to surface flows in the lower reaches, possibly due to increasingly intensive land-use asociated with irrigation in the lower Cardrona. The SQMCI trend is based on fewer (i.e., only annual) observations, however confidence the Sen slope is quite large (-0.4 /yr). The other

Page 68 | 155

#### 2000-2020

'exceptionally unlikely' or 'extremely unlikely' improving trends include turbidity (Mill Creek, Luggate Creek and Kawarau) and NNN (Luggate Creek).

Mill Creek has improving trends in DRP, *E.coli*, NNN, TN and TP. This is good news for a catchment with increasing development pressure, however the degrading trend in turbidity suggests that land disturbance, either naturally from the Crown Range or from localised development is impacting on the Creek. The good news is that despite increasing turbidity in Mill Creek, TP (which is often bound to sediment) is 'virtually certain' to be improving in Lake Hayes.

It is difficult to assess why these trends have occurred, particularly as there is no accurate information on changes in land use and land management practice would help in identifying drivers of change evident with some water quality variables.

In summary; for the majority of sites across the Dunstan Rohe, water quality is excellent;

- All sites, other than the Nevis return an 'A' band for the toxicity attribute state of ammonia. All sites return an 'A' band for the toxicity attribute state of nitrate
- Periphyton (chlorophyll *a*) at the lower Lindis site achieves a 'D' band, despite achieving 'A' bands for NNN and DRP nutrient concentrations.
- Bacterial water quality is excellent across most sites, other than the Upper Cardrona and Kawarau which both achieve a 'D' band .
- ASPM scores were generally 'B' band, with Mill Creek a 'D' band. MCI scores were generally 'C' bands with Mill Creek a 'D' band.
- Trend analysis showed 'exceptionally unlikely' improving trends for the Cardrona River (*E.coli*, NNN, SQMCI) and Luggate Creek (E.coli, NNN, turbidity)
- 'Virtually certain' improving trends were seen in Mill Creek (DRP, TP)

Page 69 | 155
2000-2020



# 8 Manuherekia Rohe (Clutha Matu/Au FMU)

Figure 14 Location of water quality monitoring sites in the Manuherekia Rohe

Page 70 | 155

2000-2020

# 8.1 State Analysis Results

The results of grading the SoE sites in the Manuherekia Rohe according to the NPSFM 2020 NOF criteria are summarised in Figure 15 and mapped in Figure 16. Many sites in the Manuherekia Rohe did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Figure 15 Grading of the river sites of the Manuherekia Rohe based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 71 | 155

2000-2020



Figure 16 Maps showing Manuherekia Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

Page 72 | 155

2000-2020

#### **Periphyton and Nutrients**

Results for the river periphyton trophic state results are shown in



(periphyton). Dunstan Creek and Manuherekia at Blackstone Hill are likely to be in attribute band 'B' as few results exceed 120 chl- $a/m^2$ . Results from the Manuherika at Ophir and Galloway show that no results are >200 mg chl- $a/m^2$ , but many individual results have Chla > 120 chl- $a/m^2$ . It is likely that these sites will fall into attribute band 'C'. There is an increase in algae in the lower Manuherekia, compared to the two sites in the upper Manuherekia (Dunstan Creek and Blackstone Hill).

Page 73 | 155

2000-2020



shows the MfE (2020) DRP and TN concentrations to manage the NPSFM 2020 periphyton attribute state (periphyton DRP and periphyton TN). The upper catchment has lower median concentrations of TN placing Manuherekia d/s Forks in the 'A' band and Dunstan Creek, Hills Creek and Blackstone in the 'B' band. Ophir and Galloway have higher TN concentrations and fall into the 'C' band. The tributaries (Poolburn and Thomsons Creek) show a much higher median concentration, accordingly their TN band is 'D'.

The median concentration of DRP in Thomsons Creek and Poolburn is high, they have been allocated a 'D' band status. The Manuherekia d/s Fork and Hills Creek have the lowest DRP median concentration and fall in the 'B' band at 20% exceedance, all the other sites are allocated a 'C' band status for DRP at 20% exceedance.

Figure 16 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Dunstan Rohe show that Thomsons Creek and the Poolburn achieve a 'D' band which is below the national bottom line, the NPSFM 2020 describes this as 'ecological communities are impacted by substantial DRP elevation above natural reference conditions. In combination with other conditions favouring eutrophication, DRP enrichment drives excessive primary production and significant changes in macroinvertebrate and fish communities, as taxa sensitive to hypoxia are lost'.

All other sites other than the lower Manuherekia main-stem (Ophir and Galloway) achieve a band 'A'. The NPSFM 2020 describes Band 'A' as 'Ecological communities and ecosystem processes are similar to those of natural reference conditions. No adverse effects attributable to dissolved reactive phosphorus (DRP) enrichment are expected'

#### 8.1.1 Toxicants

NOF attribute bands for NH4-N and nitrate (measured as NNN) toxicity are shown in Figure 16, the results show excellent protection levels against toxicity risk. All sites other than Hills Creek and Thomsons Creek return an 'A' band for NH4-N and all sites returning an 'A' band (highest level of protection) for NNN.

Page 74 | 155

#### 2000-2020

#### 8.1.2 Suspended fine sediment (Rivers)

The clarity results for the Manuherekia Rohe are shown in Figure 16. Five sites return a NOF band of 'D' which the NPSFM 2020 describes as 'High impact of suspended sediment on instream biota. Ecological communities are significantly altered and sensitive fish and macroinvertebrate species are lost or at high risk of being lost'. Only Dunstan Creek and Manuherekia downstream of Fork return a NOF band of 'A' for sediment. Historical gold mining tailings in the area below Falls Dam are likely to elevate suspended solid concentrations in the main-stem Manuherekia during higher flows.

#### 8.1.3 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state of water quality and aquatic ecosystem health at a site. Figure 16 summarises MCI scores for the Manuherekia. Dunstan Creek and Manuherekia at Ophir achieve a band 'B' for MCI whilst the Manuherekia at Blackstone and the Manuherekia at Galloway achieve a 'C' band, returning MCI scores between 100 and 110; this reflects a macroinvertebrate community indicative of moderate organic pollution or nutrient enrichment.

#### The ASPM interim NOF bands shown in



reflect those of the MCI scores, Dunstan Creek and the Manuherekia at Blackstone obtain band 'B' for ASPM, which the NPSFM 2020 describes as 'a community with mild to moderate loss of ecological integrity' and the two other mainstem Manuherekia sites achieve a 'C' band.

#### 8.1.4 Human health for recreation (Rivers)

Page 75 | 155

2000-2020



summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

The *E. coli* grades are calculated using all data regardless of flow, it is acknowledged that the actual risk will generally be less if a person does not swim during high flows (NPSFM, 2020). Thomsons Creek, Hills Creek, the Poolburn and Ophir and Galloway in the lower Manuherekia fall below the national bottom line with the attribute band either a 'D' or 'E'. Only the upper catchment site, the Manuherekia d/s of Fork (above Falls Dam) achieves 'A' bands for all four statistical tests.

Page 76 | 155

2000-2020

# 8.2 Trend Analysis



Trend analysis results for the Manuherekia Rohe is shown in Figure 17

Figure 17 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined inTable 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend.

Dunstan Creek has the largest number of trends showing 'exceptionally unlikely' or 'extremely unlikely' improvement. These include E.coli (Figure 18), NNN and turbidity over a 10-year period and *E.coli*, NNN and TN over a 20-year period. Other trends include an 'exceptionally unlikely' improving trend for *E. coli* at Ophir (over 10-years).

There are three sites with 'virtually certain' or 'extremely likely' improving trends, NH4-N at Dunstan Creek and Galloway and DRP at Ophir.

Page 77 | 155

2000-2020



Figure 18 Dunstan Creek, E.coli is 'exceptionally unlikely' to be improving (over 10-years)

# 8.3 Water quality summary and discussion: Manuherekia Rohe

The tables in this section summarise:

- 1) River sites where the national bottom line is not met (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) All trends using raw data for rivers.
- Table 12 Summary of river, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve a band 'D'.

| sID                     | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|-------------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Thomsons Creek at SH85  |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Poolburn at Cob Cottage |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Manuherekia d/s Fork    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Manuherekia at Ophir    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Manuherekia at Galloway |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Manuherekia at          |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Hills Creek             |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Page 78 | 155

#### 2000-2020

| Duncton Crook   |  |  |  |  |  |  |  |
|-----------------|--|--|--|--|--|--|--|
| Dulistali Creek |  |  |  |  |  |  |  |
|                 |  |  |  |  |  |  |  |

# Table 13 Summary of river sites where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**.

| npID                 | nObs | Freq    | Period     | AnnualSenSlope  | DirectionConf          | Descriptor |
|----------------------|------|---------|------------|-----------------|------------------------|------------|
|                      |      | Dunsta  | an Creek a | at Beattie Road |                        | 1          |
| Ammoniacal N         | 67   | Qtr     | 20         | 0               | Virtually certain      | 1111       |
| E. coli              | 57   | BiMonth | 10         | 3.875332        | Exceptionally unlikely | 1111       |
| Nitrite/Nitrate N    | 67   | Qtr     | 20         | 0.002833        | Exceptionally unlikely | 1111       |
| Total Nitrogen       | 67   | Qtr     | 20         | 0.003444        | Exceptionally unlikely | 1111       |
| Turbidity            | 56   | BiMonth | 10         | 0.036977        | Extremely unlikely     | 111        |
|                      |      | Mai     | nuherekia  | at Galloway     |                        |            |
| Ammoniacal N         | 118  | BiMonth | 20         | 0               | Virtually certain      | 1111       |
| Turbidity            | 118  | BiMonth | 20         | 0.059861        | Exceptionally unlikely | 1111       |
|                      |      | М       | anuherek   | kia at Ophir    |                        |            |
| Dissolved Reactive P | 101  | Month   | 10         | -0.00072        | Virtually certain      | 1111       |
| E. coli              | 101  | Month   | 10         | 8.899505        | Exceptionally unlikely | 1111       |

Table 14 Overall summary of trends for the Manuherekia Rohe using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in *Error! Reference source not found.*.

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descriptor             | 1111                 | 111                 | <b>1</b> 1  | Ŷ      | ↔                   | Ļ        | ↓↓               | 111                   | 1111                      |
| Rivers - 10-year trend | 1                    |                     |             | 10     | 4                   | 4        | 1                | 1                     | 2                         |
| Rivers - 20-year trend | 2                    |                     |             | 1      | 4                   | 1        |                  | 3                     |                           |

In the Manuherekia Rohe state analysis identified that upstream of Falls Dam water quality was generally very good and achieved the NPSFM 2020 attribute band 'A' for all attributes measured. For periphyton the mainstem upper Manuherekia (Blackstone Hill) and Dunstan Creek achieved attribute band 'B', but in the lower Manuherekia (Galloway and Ophir) this dropped to attribute band 'C'. For *E. coli* the upper Manuherekia achieved attribute band 'B'/'C' but the lower Manuherekia main-stem and all tributaries achieved an attribute band 'D', placing them below the national bottom line.

In the Manuherekia catchment soils with poorer drainage characteristics are found on the true right of the Manuherekia River, particularly around the Thomsons Creek and Lauder Creek catchments. The P a g e 79 | 155

#### 2000-2020

implication of poor soil drainage is run-off (from rainfall/irrigation) can transport soil and associated bacteria and nutrients to the nearest watercourse, contributing to poor water quality. The poor water quality in Thomsons Creek is likely replicated across all Creeks originating in the Dunstan Mountains, water quality becomes poorer as these tributaries flow over productive farmland towards the Manuherekia.

Across the Manuherekia Rohe there are a number of sites with degrading water quality trends, as shown in Table 13. Tributary sites which have 'state' below the national bottom line are likely contributing to the degrading trends in the main-stem. At Ophir an 'exceptionally unlikely' improving trend for E.coli was identified, but trends in Dunstan Creek were degrading for E.coli, NNN and turbidity (10-years) and *E.coli*, NNN and TN (over 20-years).

In Table 13 only sites with 99%, 95% ,1% and 5% confidence levels are shown. These equate to the 'virtually certain', 'extremely likely', 'exceptionally unlikely' and extremely unlikely' categories. It is important to note when sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction, at these sites the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Manuherekia Rohe, these sites include NH4-N at Dunstan Creek and Manuherekia at Galloway.

#### In summary:

- Bacterial water quality is excellent in the Manuherekia above Falls Dam, concentrations
  increase downstream with both Ophir and Galloway achieving band 'D', below the national
  bottom line.
- Bacterial water quality is below the national bottom line at all tributary sites (Hills Creek, Thomsons Creek and the Poolburn)
- Nutrients increase in the main-stem between Blackstone and Ophir, then DRP improves downstream to achieve band 'B' at Galloway.
- The tributaries, Poolburn and Thomsons Creek, have poor water quality across all attribute states other than toxicity mainly achieving band 'D', below the NPSFM 2020 bottom line.
- Dunstan Creek has degrading trends for *E. coli*, NNN and turbidity (over 10-years)
- Ophir has an 'exceptionally unlikely' improving trend for *E. coli* (over 10-years).

Page 80 | 155

2000-2020

# 9 Roxburgh Rohe (Clutha Matu/Au FMU)



Figure 19 Location of water quality monitoring sites in the Roxburgh Rohe

Page 81 | 155

2000-2020

# 9.1 State Analysis Results

The results of grading the SoE sites in the Roxburgh Rohe according to the NPSFM 2020 NOF criteria are summarised in Figure 20 and mapped Figure 21. Many sites in the Roxburgh Rohe did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Figure 20 Grading of the river and lake sites in the Roxburgh based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 82 | 155

2000-2020



Figure 21 Maps showing Roxburgh Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

Page 83 | 155

2000-2020

#### 9.1.1 Periphyton and Nutrients

Results for the river periphyton trophic state results are shown in Figure 20 (periphyton). The Roxburgh Rohe does not have any sites that are monitored for chlorophyll a, but MfE (2020) TN concentrations to manage the NPSFM 2020 periphyton attribute states show that the Fraser River achieves a band 'A' as few results exceed 50 chl- $a/m^2$  meaning that blooms would be rare, reflecting negligible nutrient enrichment, the Clutha at Millers Flat a band 'B', the Teviot River a band 'C' and the Benger Burn achieves a band 'D'.

The MfE (2020) DRP concentrations to manage the NPSFM 2020 periphyton attribute states show the same pattern as the TN bands, other than the Fraser at Old Man Range which achieves a 'B' for DRP, when it achieved an 'A' band for TN.

Figure 21 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Roxburgh Rohe show that every site achieves a band 'A', other than the Benger burn which achieves band 'C'. The NPSFM 2020 describes band 'C' as 'Ecological communities are impacted by moderate DRP elevation above natural reference conditions. If other conditions also favour eutrophication, DRP enrichment may cause increased algal and plant growth, loss of sensitive macro-invertebrate and fish taxa, and high rates of respiration and decay'

The NPSFM 2020 describes how phytoplankton affects lake ecological communities. If the chlorophyll a concentration is in the 'A' band, then 'Lake ecological communities are healthy and resilient, similar to natural reference conditions'. **Error! Reference source not found.** Results for Lake Onslow are shown in Figure 21, the lake achieves an 'A' band for maximum chlorophyll a, but drops to a 'B' band for median chlorophyll a. Lake Onslow achieves a 'B' band for TN and a 'C' band for TP, which indicates that ecological communities are slightly-moderately impacted by additional algal and plant growth arising from nutrient levels above natural reference conditions.

#### 9.1.2 Toxicants (Rivers)

In the Roxburgh Rohe the NOF attribute bands for NH4-N and nitrate (measured as NNN) toxicity) show excellent protection levels against toxicity risk as all monitoring sites return an 'A' band for NH4-N and NNN.

#### 9.1.3 Suspended fine sediment

The clarity results for the Roxburgh Rohe are shown in Figure 20**Error! Reference source not found.**. The Fraser River returns a NOF band of 'A' which denotes '*minimal impact of suspended sediment on instream biota. Ecological communities are similar to those observed in natural reference conditions'* (NPSFM, 2020). The Clutha at Millers Flat returns a NOF band of 'B' and the Benger burn and Teviot return a NOF band of 'D' for suspended fine sediment, which is below the national bottom line.

#### 9.1.4 Human health for recreation

Figure 20 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

Lake Onslow, the Fraser River and the Clutha at Millers Flat return 'A' bands across all four statistical tests, but the Teviot and Benger Burn achieve a 'D' band because their interim 95<sup>th</sup> percentile is >1200 *E.coli*/100ml. Lake Onslow returns a

Page 84 | 155

2000-2020

# 9.2 Trend Analysis

Results from trend analysis for the Roxburgh Rohe is shown in Figure 22.



Figure 22 Summary of Roxburgh Rohe sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined inTable 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend.

Trend analysis for both rivers and lakes are given in Figure 22. In the 10-year time frame, Lake Onslow shows 'exceptionally unlikely' improving trends for NH4-N and chlorophyll a, where most of the other variables show 'likely' to 'virtually certain' improving trends. For the Clutha River at Millers Flat, trend analysis shows a 20-year 'exceptionally unlikely' improvement in turbidity and an 'unlikely' improvement in NH4-N, however nutrient concentrations have improving trends, NNN is 'virtually certain' to have improved over 20-years and TP is 'extremely likely' to have improved.

Page 85 | 155

#### 2000-2020

# 9.3 Water quality summary Roxburgh Rohe

The tables in this section summarise:

- 1) River and lake sites where attributes where the national bottom line is not met (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) All trends using raw data for rivers and continuous data for lakes over the two time-periods.

Table 15 Summary of river state, red cells show where state does not meet the national bottom line in one or more variable.

| sID                    | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|------------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Teviot at Bridge Huts  |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Fraser at Old Mans     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Clutha at Millers Flat |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Benger burn at SH8     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Table 16 Summary of river sites where trends are greater than 'likely' or 'unlikely' (raw data). Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

| npID              | nObs | Freq  | Period    | AnnualSenSlope | DirectionConf          | Descriptor  |
|-------------------|------|-------|-----------|----------------|------------------------|-------------|
|                   |      |       | Clutha at | Millers Flat   |                        |             |
| Nitrite/Nitrate N | 237  | Month | 20        | -0.00034       | Virtually certain      | 1111        |
| Total Phosphorus  | 235  | Month | 20        | -5.52417       | Extremely likely       | <b>↑</b> ↑↑ |
| Turbidity         | 237  | Month | 20        | 0.03662        | Exceptionally unlikely | 1111        |

Table 17 Summary of Lake Onslow trends when they are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

| npID              | nObs | Freq    | Period   | DirectionConf          | AnnualSenSlope | Descriptor  |
|-------------------|------|---------|----------|------------------------|----------------|-------------|
|                   |      |         | Lake Ons | low at Boat Ramp       |                |             |
| Ammoniacal N      | 49   | BiMonth | 10       | Exceptionally unlikely | 0              | 1111        |
| Chlorophyll a     | 49   | BiMonth | 10       | Exceptionally unlikely | 0              | 1111        |
| Nitrite/Nitrate N | 49   | BiMonth | 10       | Extremely likely       | 0              | <u></u> ΥΥΥ |
| Total Phosphorus  | 49   | BiMonth | 10       | Extremely likely       | -0.00096       | <u></u> ΥΥΥ |
| Turbidity         | 36   | Qtr     | 10       | Virtually certain      | -0.36326       | <b>1111</b> |

Table 18 Overall summary of trends for Lake Onslow (continuous data). Confidence is expressed categorically based on the levels defined in *Error! Reference source not found*.

Page 86 | 155

#### 2000-2020

|                       | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | схсериопану<br>unlikely |
|-----------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|-------------------------|
| Descriptor            | 1111                 | 111                 | <b>1</b> 1  | Ŷ      | ↔                   | Ļ        | 11               | 111                   | 1111                    |
| River – 10-year trend |                      |                     |             | 4      | 1                   |          |                  |                       |                         |
| River – 20-year trend | 2                    | 2                   | 1           |        |                     |          |                  |                       | 2                       |
| Lakes – 10-year trend |                      |                     |             |        |                     |          |                  |                       |                         |

The State analysis identified water quality in the Roxburgh Rohe rivers is generally good and the NPSFM 2020 band 'A' was achieved for most attributes. The only exceptions were for suspended fine sediment which was below the national bottom line in the Teviot and the Benger Burn. The suspended fine sediment in the Teviot is likely due to Lake Onslow, the main input to the river, as the lake is shallow and susceptible to sediment resuspension from wind-driven waves. E.coli was also below the national bottom line at these two sites.

In Table 16 and Table 17 only sites with 99%, 95%, 1% and 5% confidence levels are shown. These equate to the 'virtually certain', 'extremely likely', 'exceptionally unlikely' and extremely unlikely' categories. When sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Roxburgh Rohe, Lake Onslow had three parametes with a zero Sen slope; NH4-N, chlorophyll a and NNN. Lake Onslow, over 20-years had a 'virtually certain' improving trend for NNN and an 'extremely likely' improving trend for TP. In the same timeframe, turbidity showed an 'exceptionally unlikely' improving trend.

In summary:

- Water quality in the Clutha at Millers Flat and Fraser River generally achieve 'A' bands.
- The Teviot River does not meet the national bottom line for *E. coli* or suspended fine sediment.
- The Benger Burn does not meet the national bottom line for suspended fine sediment or *E. coli* and 'D' bands are achieved for periphyton DRP and periphyton TN.
- The Clutha at Millers Flat has an 'exceptionally unlikely' improving trend for turbidity (over 20-years).

Page 87 | 155

2000-2020



# 10 Lower Clutha Rohe (Clutha Matu/Au FMU)

Figure 23 Location of water quality monitoring sites in the Lower Clutha Rohe

Page 88 | 155

2000-2020

## 10.1 State Analysis Results

The results of grading the SoE sites in the Lower Clutha Rohe according to the NPSFM 2020 NOF criteria are summarised in Figure 24 and mapped in Figure 25. Many sites in the Lower Clutha Rohe did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicates the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Page 89 | 155

#### 2000-2020

Figure 24 Grading of River and Lake sites in the Lower Clutha Rohe, based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline



Page 90 | 155

#### 2000-2020

Figure 25 Maps showing Lower Clutha Rohe sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

#### 10.1.1 Periphyton and Nutrients

Periphyton trophic state results to date are given in Figure 24 and show that the Lower Clutha Rohe returns either band 'A' or band 'B' for periphyton. The Blackcleugh Burn, Upper Pomahaka and Lower Waipahi have an interim 'A' band as few results exceed 50 chl- $a/m^2$ , reflecting negligible nutrient enrichment. The Waitahuna records an interim band 'B'.

Figure 25 shows the MfE (2020) DRP and TN concentrations to manage the NPSFM 2020 periphyton attribute state (periphyton DRP and periphyton TN). Using the 20% exceedance criteria (mid-range), the DRP and TN median concentrations in the Lower Clutha Rohe generally exceed the T200 mg chla/m<sup>2</sup> band and most sites achieve a band 'D', as would be expected in a high nutrient environment. The outliers are the Clutha at Balclutha, Upper Pomahaka and Blackcleugh Burn which achieve a band 'B'.

Figure 24 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Lower Clutha Rohe are varied, the sites with low nutrients achieve band 'A', the NPSFM 2020 describes this attribute state as 'ecological communities and ecosystem processes are similar to those of natural reference conditions. No adverse effects attributable to dissolved reactive phosphorus (DRP) enrichment are expected'.

The Pomahaka catchment has eight sites, the upper two sites (Upper Pomahaka and Pomahaka at Glenken) achieve 'A' bands. The tributaries entering the Pomahaka tend to have very high DRP, for example the Crookston Burn, Heriot Burn and Wairuna achieve band 'D'. The effect of the high DRP inputs is that the lower mainstem Pomahaka (Burkes Ford) achieves a 'B' band.

The NPSFM 2020 describes how phytoplankton (measured as chlorophyll a) affects lake ecological communities. If phytoplankton is in the 'A' band, then 'Lake ecological communities are healthy and resilient, similar to natural reference conditions'. Figure 24 shows that Lake Tuakitoto is in the 'D' band, which is described as 'ecological communities have undergone or are at high risk of a regime shift to a persistent, degraded state (without native macrophyte/seagrass cover), due to impacts of elevated nutrients'. Lake Tuakitoto achieves 'D' bands for both total nitrogen and total phosphorus, a 'D' band reflects high nutrient enrichment, which is consistent for a shallow (normal lake levels of about one metre) freshwater wetland (ORC, 2004).

#### 10.1.2 Toxicants

NOF attribute bands for NH4-N are given in **Error! Reference source not found.** The national bottom line for NH4-N is below band 'B'. In the Lower Clutha all sites achieve band 'A' band other than four

Page 91 | 155

#### 2000-2020

sites in the Pomahaka catchment (Crookston Burn, Heriot Burn and Wairuna and Waipahi) which achieve a band 'B', which affords a 95% species protection level.

NOF attribute bands for nitrate (measured as NNN) toxicity are given in Figure 24, again the national bottom line is below band 'B'. In the Lower Clutha Rohe most sites achieve either an 'A' or 'B' band, other than Wairuna which achieves a 'C' band (annual 95<sup>th</sup> percentile). The NPSFM 2020 describes the 'C' band as NNN having 'growth effects on up to 20% of species (mainly sensitive species such as fish). No acute effects.'

Lake Tuakitoto returns a 'B' band (95% species protection level) for NH4-N toxicity, showing good protection levels against toxicity risk.

#### 10.1.3 Suspended fine sediment

The clarity results for Lower Clutha Rohe are shown in Figure 24. Most of the sites return a NOF band of 'D', which the NPSFM 2020 describes as 'high impact of suspended sediment on instream biota. Ecological communities are significantly altered and sensitive fish and macroinvertebrate species are lost or at high risk of being lost'. Four sites; Waiwera, Waipahi at Cairns Peak, Upper Pomahaka and Blackcleugh Burn, return an 'A' band.

#### 10.1.4 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state if water quality and aquatic ecosystem health at a site. Figure 24 summarises MCI scores for sites monitored for aquatic macroinvertebrates throughout the Lower Clutha Rohe.

Three of the monitored sites; Waipahi at Waipahi, Wairuna and Waiwera achieve an interim MCI score below the national bottom line (MCI 90). The NPSFM 2020 describes this state as 'reflecting a macroinvertebrate community indicative of sever organic pollution or nutrient enrichment. Communities are largely composed of taxa insensitive to inorganic pollution/nutrient enrichment'. The Upper Pomahaka and Blackcleugh Burn achieve the highest MCI scores, achieving a 'B' band.

The ASPM interim scores shown in Figure 25 generally reflect those of the MCI scores, only the Waiwera falls below the national bottom line with a median score of 0.3 to achieve a 'D' grade. Four sites achieve a 'B' grade.

#### 10.1.5 Human health for recreation (Rivers)

Figure 25 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

Compliance is generally poor across the Lower Rohe, with 12 of 15 sites returning bacterial water quality below the national bottom line. The NPSFM 2020 describes band 'D' as '30% of the time the estimated risk is  $\geq$ 50 in 1,000 (>5% risk). The predicted average infection >3%'. Only the Blackcleugh Burn achieved an 'A' band.

In the Pomahaka catchment, of the eight sites monitored three sites; the Crookston Burn, Heriot Burn and Wairuna achieved an 'E' band, three sites; Waipahi at Cairns Peak, Upper Pomahaka, Pomahaka

Page 92 | 155

#### 2000-2020

at Burkes Ford and Pomahaka at Glenken achieved a 'D' band and one site; Waipahi at Waipahi achieved a 'B' band.

# 10.2 Trend Analysis



Trend analysis results for the Lower Clutha Rohe are shown in Figure 26 and Figure 27

Figure 26 Summary of Lower Clutha Rohe sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend P a g e 93 | 155

#### 2000-2020

indicates improvement is expressed using the categorical levels of confidence defined in Table 2Table 1. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend.



Figure 27 Summary of Lake Tuakitoto trends, categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring).

Trend analysis for the Lower Clutha Rohe rivers is shown in Figure 26. The Waitahuna returns 'exceptionally unlikely' improving trends over 20-years for DRP, *E. coli*, TN, TP and turbidity.

Over the 20-year period DRP and NNN were most likely to show 'exceptionally unlikely' improvement, and NH4-N most likely to show 'virtually certain' improvement. Over the 10-year period most sites and most analytes are showing 'likely' to 'virtually certain' improvement, for example E.coli in the Heriot Burn (Figure 28).

Trend analysis for Lake Tuakitoto is shown in Figure 27, DRP over both 10-year and 18-year periods is 'exceptionally unlikely' to be improving.

Page 94 | 155

2000-2020



Figure 28 Heriot Burn. E.coli is 'virtually certain' to be improving (over 10-years)

# 10.3 Water quality summary and discussion: Lower Clutha Rohe

The tables in this section summarise:

- 1) River and lake sites where attributes where the national bottom line is not met (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) All trends using raw data for rivers and continuous data for lakes over the two time-periods.

Table 19 Summary of river and lake state, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve band 'D'.

| sID                    | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|------------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Waiwera at Maws Farm   |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Waiwera at Confluence  |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Waitahuna at Tweeds    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Wairuna at Millar Road |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
|                        |             |                |      |              |           |        |     |              |           | Р          | age              | 95              | 155                        |

#### 2000-2020

| Waipahi at Waipahi       |  |  |  |  |  |  |  |
|--------------------------|--|--|--|--|--|--|--|
| Waipahi at Cairns Peak   |  |  |  |  |  |  |  |
| Upper Pomahaka at        |  |  |  |  |  |  |  |
| Tuapeka at 700m u/s      |  |  |  |  |  |  |  |
| Pomahaka at Glenken      |  |  |  |  |  |  |  |
| Pomahaka at Burkes       |  |  |  |  |  |  |  |
| Lovells Creek at Station |  |  |  |  |  |  |  |
| Heriot Burn at Park Hill |  |  |  |  |  |  |  |
| Crookston Burn at Kelso  |  |  |  |  |  |  |  |
| Clutha at Balclutha      |  |  |  |  |  |  |  |
| Blackcleugh Burn at      |  |  |  |  |  |  |  |

## Table 20 Summary of Lake Tuakitoto State where attributes are graded D or below

|                          | NH4-N<br>max | NH4-N<br>median | Chlorophyll a<br>-max | Chlorophyll a<br>median | E.coli<br>Overall | TN | ΤP |
|--------------------------|--------------|-----------------|-----------------------|-------------------------|-------------------|----|----|
| Lake Tuakitoto at Outlet |              |                 |                       |                         |                   |    |    |

Page 96 | 155

#### 2000-2020

| nnID                 | nObc  | Frog      | Doriod     | Annual              | Direction              | Descriptor  |
|----------------------|-------|-----------|------------|---------------------|------------------------|-------------|
|                      |       | rieq      | Periou     | Annual<br>Palalutha | Direction              | Descriptor  |
| Discolved Reactive R | 110   | Month     | 10         |                     | Virtually cortain      | ****        |
| Dissolved Reactive P | 220   | Month     | 20         | 1 95 05             | Extromoly likoly       |             |
| Total Nitrogan       | 239   | Month     | 20         | -1.0E-05            | Extremely likely       |             |
| Total Nitrogen       | 110   | Month     | 20         | -0.00101            | Extremely likely       | TTT         |
| Total Phosphorus     | 220   | Month     | 10         | -0.00019            | Futromoly likely       | TT          |
| Turkidite            | 239   | Nonth     | 20         | -9.4E-05            |                        | TTT         |
| Turbidity            | 239   | Crock     | 20         | 0.050084            |                        | 1 1111      |
| Ammoniacal N         | 00    | DiMonth   |            |                     | \/irtually/cortain     |             |
| Ammoniacai N         | 99    | Bilvionth | 20         | -0.00117            |                        | 11111       |
| Dissolved Reactive P | 99    | Bilvionth | 20         | 0.000914            | Exceptionally unlikely | ++++        |
| Total Phosphorus     | 99    | Bilvionth | 20         | 0.000942            | Exceptionally unlikely | 1 1111      |
|                      | 110   | Herio     | Burn at    |                     | Fotoson alto librato   |             |
| Ammoniacai N         | 110   | Blivionth | 20         | -0.00031            | Extremely likely       |             |
| Dissolved Reactive P | 103   | Nonth     | 10         | -0.00099            |                        |             |
| Dissolved Reactive P | 116   | BilMonth  | 20         | 0.000996            | Exceptionally unlikely | 1111        |
| E. coli              | 103   | Month     | 10         | -39.9303            | Virtually certain      | 1111        |
| Nitrite/Nitrate N    | 116   | BiMonth   | 20         | 0.018769            | Exceptionally unlikely | 1111        |
| Total Nitrogen       | 104   | Month     | 10         | -0.03997            | Virtually certain      | 1111        |
| Total Nitrogen       | 116   | BiMonth   | 20         | 0.015408            | Extremely unlikely     | 111         |
| Turbidity            | 104   | Month     | 10         | -0.25686            | Extremely likely       | <u>^</u>    |
|                      |       | Pom       | ahaka at   | Burkes Ford         | 1                      | 1           |
| Ammoniacal N         | 115   | BiMonth   | 20         | -0.00025            | Virtually certain      | 1111        |
| Dissolved Reactive P | 102   | Month     | 10         | -0.0004             | Virtually certain      | 1111        |
| Dissolved Reactive P | 115   | BiMonth   | 20         | 0.000198            | Extremely unlikely     | 111         |
|                      |       | Po        | mahaka a   | t Glenken           | 1                      | 1           |
| Ammoniacal N         | 116   | BiMonth   | 20         | 0                   | Virtually certain      | 1111        |
| Turbidity            | 57    | BiMonth   | 10         | -0.10541            | Extremely likely       | <u>^</u>    |
|                      |       | Wa        | ipahi at C | airns Peak          |                        |             |
| Nitrite/Nitrate N    | 68    | Qtr       | 20         | 0.015656            | Exceptionally unlikely | 1111        |
| Total Nitrogen       | 57    | BiMonth   | 10         | -0.02694            | Extremely likely       | 111         |
| Total Nitrogen       | 68    | Qtr       | 20         | 0.023244            | Exceptionally unlikely | 1111        |
| Turbidity            | 57    | BiMonth   | 10         | -0.39658            | Virtually certain      | 1111        |
| Turbidity            | 66    | Qtr       | 20         | -0.1245             | Extremely likely       | 111         |
|                      |       | V         | /aipahi at | : Waipahi           |                        |             |
| Ammoniacal N         | 104   | Month     | 10         | 0                   | Extremely likely       | <u>^</u>    |
| Ammoniacal N         | 116   | BiMonth   | 20         | -0.00042            | Virtually certain      | 1111        |
| Dissolved Reactive P | 104   | Month     | 10         | -0.00051            | Virtually certain      | 1111        |
| Dissolved Reactive P | 116   | BiMonth   | 20         | 0.000248            | Extremely unlikely     | 111         |
| MCI                  | 10    | Year      | 10         | -0.78438            | Extremely unlikely     | 111         |
| Turbidity            | 104   | Month     | 10         | -0.1321             | Extremely likely       | 111         |
| Turbidity            | 115   | BiMonth   | 20         | -0.05558            | Extremely likely       | <u>↑</u> ↑↑ |
|                      |       | Wa        | iruna at N | Villar Road         | · · · · · · · · ·      |             |
| Ammoniacal N         | 100   | Month     | 10         | -0.00233            | Virtually certain      | 1111        |
| Ammoniacal N         | 68    | Otr       | 20         | -0.002              | Virtually certain      | 1111        |
| Dissolved Reactive P | 100   | Month     | 10         | -0.001              | Virtually certain      | <u></u>     |
| Dissolved Reactive P | 68    | Otr       | 20         | 0.000541            | Extremely unlikely     | 111         |
| Nitrite/Nitrate N    | 68    | Otr       | 20         | 0.031128            | Exceptionally unlikely | 1111        |
| Turbidity            | 100   | Month     | 10         | -0 37005            | Extremely likely       | 111         |
|                      | 100   | Waita     | huna at T  | weeds Bridge        | Extremely interv       |             |
| Dissolved Reactive P | 105   | Month     | 10         | -0.00027            | Extremely likely       | 111         |
| Dissolved Reactive P | 117   | RiMonth   | 20         | 0.00027             | Excentionally unlikely |             |
| E coli               | 117   | BiMonth   | 20         | 8 810/25            |                        |             |
|                      | 10    | Voor      | 10         | 1 01612             |                        | <u> </u>    |
| Nitrito /Nitrato N   | 117   | BiMonth   | 10         | 0.002619            | Extromoly unlikely     | 4444        |
|                      | 116   | BiMonth   | 20         | 0.002010            |                        | 444         |
|                      | 1 110 |           | 20         | 0.000/33            |                        | 1 1111      |

# Table 21 Summary of river sites where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in *Error! Reference source not found.*.

Page 97 | 155

#### 2000-2020

| Total Phosphorus     | 116 | BiMonth | 20 | 0.000817 | Exceptionally unlikely | 1111 |  |  |
|----------------------|-----|---------|----|----------|------------------------|------|--|--|
| Turbidity            | 117 | BiMonth | 20 | 0.072094 | Exceptionally unlikely | 1111 |  |  |
| Waiwera at Maws Farm |     |         |    |          |                        |      |  |  |
| Turbidity            | 106 | Month   | 10 | -0.12939 | Extremely likely       | 111  |  |  |

Table 22 Summary lake sites where trends (continuous data) are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not** found.

| nplD                     | nObs | Freq    | Period | TrendDirection | AnnualSenSlope | Descriptor |  |  |  |
|--------------------------|------|---------|--------|----------------|----------------|------------|--|--|--|
| Lake Tuakitoto at Outlet |      |         |        |                |                |            |  |  |  |
| Dissolved Reactive P     | 59   | BiMonth | 10     | Increasing     | 0.001987755    | 1111       |  |  |  |
| Dissolved Reactive P     | 104  | BiMonth | 18     | Increasing     | 0.001845163    | 1111       |  |  |  |
| E. coli                  | 59   | BiMonth | 10     | Decreasing     | -5.947082768   | 111        |  |  |  |
| Total Phosphorus         | 104  | BiMonth | 18     | Increasing     | 0.002423052    | 1111       |  |  |  |

Table 23 Overall summary of trends for the Lower Clutha Rohe using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in *Error! Reference source not found.*.

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descriptor             | 1111                 | 111                 | <b>↑</b> ↑  | 1      | ↔                   | Ļ        | ΥĻ               | 111                   | 1111                      |
| Rivers - 10-year trend | 8                    | 9                   | 12          | 21     | 16                  | 8        |                  | 1                     | 1                         |
| Rivers - 20-year trend | 5                    | 6                   | 3           | 7      | 9                   | 8        | 2                | 5                     | 13                        |
| Lakes – 10-year trend  |                      | 2                   |             | 1      | 3                   | 1        |                  |                       | 1                         |
| Lakes – 18 year trend  |                      | 1                   | 1           | 1      |                     |          | 2                |                       | 2                         |

In the Lower Clutha Rohe water quality generally has poor water clarity and high bacteria and nutrient concentrations. Of the NOF attribute states, *E. coli* was below the bottom line in 12 of the 15 sites monitored, suspended solids below the national bottom line in seven of the 15 sites and DRP in four of the monitored sites.

Lake Tuakitoto is a large freshwater wetland situated in the lower Clutha River catchment, Lovells Creek is the main inflow into the Lake. Lovells Creek scores poorly across all attribute states and is a reflection of the catchment, which largely consists of intensively grazed pasture with some scrub, and plantation forestry. Lake Tuakitoto scores 'D' bands for TP, TN and chlorophyll a (phytoplankton), this situation is unlikely to change, due to the shallow nature of the lake and poor flushing flows.

Alongside the poor state, trend analysis shows that water quality continues to degrade at some sites. The Waitahuna has degrading trends for DRP, *E. coli*, TN, TP and turbidity. The reason for this is unknown, as stated previously, having accurate information on changes in land use and land management practice will help in identifying drivers of change evident with some water quality variables.

In the case of the Pomahaka catchment, of the six monitoring sites which achieve a band 'D' or E' for E.coli most score poorly across all statistical tests with four of the six sites returning 'D' grades for the median E.coli statistic. It is thought that insufficient effluent storage and a prevalence of mole and tile

Page 98 | 155

#### 2000-2020

drains through areas of the lower Pomahaka catchment result in these high *E. coli* concentrations. This is being addressed through Plan change 6AA, one of the aims of which is to strengthen provisions on farm effluent management. ORC is working throughout the Pomahaka catchment with groups such as the Pomahaka Watercare Group, the Landcare Trust and the Clutha Development Trust to address water quality issues. A large part of this effort is focused on improving bacterial water quality.

The Pomahaka catchment shows some positives, thre are far fewer degrading trends over the last 10years than in the 20-year time-period. The Heriot Burn shows a 'virtually certain' improving trend for *E. coli* and TN, equally the Wairuna shows an 'virtually certain' improvement in NH4-N and DRP. The lower Pomahaka site at Burkes Ford also shows encouraging results, with DRP showing 'virtually certain' improvement

In Table 21 and Table 22 only sites with 99%, 95%, 1% and 5% confidence levels are shown. These equate to the 'virtually certain', 'extremely likely', 'exceptionally unlikely' and extremely unlikely' categories. When sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction, at these sites the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Lower Clutha Rohe only NH4-N at Waipahi was in this category.

#### In summary:

- Every site achieves 'A' or 'B' band for ammonia and nitrate toxicity.
- Nutrient concentrations are generally high, other than in the main-stem Clutha, Blackcleugh Burn and Upper Pomahaka.
- No sites achieve 'A' bands in every NOF attribute state monitored. All sites have varying degrees of degraded water quality;
- In the Pomahaka catchment, bacterial water quality is severely degraded at all monitoring sites other than the lower Waipahi.
- The Heriot Burn, Crookston Burn, Waiwera River and Waipahi at Cairns Peak are the worst
  performing sites of the Lower Clutha/Pomahaka reporting region failing to meet the national
  bottom line for many attributes. The mainstem Pomahaka becomes degraded with distance
  downstream due to poor water quality inputs from these tributaries.
- The Waitahuna has degrading trends for DRP, E. coli, TN, TP and turbidity.

Page 99 | 155

2000-2020

# 11 Taieri FMU



Figure 29 Location of water quality monitoring sites in the Taieri FMU

Page 100 | 155

2000-2020

# 11.1 State Analysis Results

The results of grading the SoE sites in the Taieri FMU according to the NPSFM 2020 NOF criteria are summarised in Figure 30 and Figure 31 and mapped in Figure 32. Many sites in the Taieri FMU did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Figure 30 Grading of the river sites of the Taieri FMU based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with P a g e 101 | 155

#### 2000-2020



coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Figure 31 Grading of the lake sites of the Taieri FMU based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 102 | 155

2000-2020



Figure 32 Maps showing Taieri FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines

Page 103 | 155

2000-2020

#### 11.1.1 Periphyton and Nutrients

Results for the river periphyton trophic state results are shown in Figure 32 (periphyton). Periphyton trophic state results to date show that the Taieri FMU achieves either 'A' or 'B' attribute bands for periphyton. The Kye Burn, Taieri at Outram and Taieri at Sutton achieve an interim 'A' band as few results exceed 50 chl-*a*/m<sup>2</sup>, reflecting negligible nutrient enrichment. The Silverstream and Taieri at Waipiata record interim 'B' bands this reflects a low nutrient enrichment but the possibility of occassional blooms (NPSFM, 2020)

Figure 31 shows the MfE (2020) DRP and TN attribute bands to manage the NPSFM 2020 periphyton attribute state (periphyton DRP and periphyton TN). Using the 20% exceedance criteria (mid-range), the TN median concentrations in the Taieri FMU are generally in the T120 mg chl-a/m<sup>2</sup> or T200 mg chl-a/m<sup>2</sup> band ('B' or 'C'). The only outliers are the Owhiro Stream and the Silverstream which achieve band 'D', both are located on the Taieri Plain.

Using the 20% exceedance criteria (mid-range), the DRP median concentrations in the Taieri FMU are generally in the T200 mg chl-a/m<sup>2</sup> band (or band 'C'), reflecting moderate nutrient enrichment. The Silverstream at Three Mile Road has a lower DRP median concentration and achieves band 'B'. Five sites achieve a band 'D', three mainstem sites (Waipiata, Patearoa Maniototo Road Bridge, Creamery Road Bridge), Owhiro Stream and the Contour Channel.

Figure 32 shows median DRP for an attribute state around wider ecological health, rather than just chlorophyll *a*. The results in the Taieri FMU show that most sites achieve either a band 'A' or band 'B', the NPSFM 2020 describes this as 'having ecological communities and ecosystem processes similar or slightly impacted by minor DRP elevation above natural reference condition'. Three sites achieved a 'C' band, including two mainstem Taieri sites (Patearoa Road Bridge and Waipiata). Two sites achieved a band 'D' for the DRP median statistic, the Owhiro Stream and the Taieri at Creamery Road Bridge.

The NPSFM 2020 describes how phytoplankton affects lake ecological communities. If phytoplankton is in the 'A' band, then 'Lake ecological communities are healthy and resilient, similar to natural reference conditions'. Figure 31 shows that Lake Waihola is in the 'C' band, which the NPSFM 2020 describes as 'ecological communities have undergone or are at high risk of a regime shift to a persistent, degraded state, due to impacts of elevated nutrients'. Lake Waihola achieves 'C' bands for both TN and TP, a 'C' band reflecting nutrient enrichment well above natural reference conditions, which is consistent for a shallow freshwater wetland (ORC, 2004).

#### 11.1.2 Toxicants (Rivers)

The NOF attribute bands for NH4-N are shown in Figure 32 and show excellent protection levels against toxicity risk. All sites return an 'A' band other than the Contour Channel, Silver Stream, Taieri at Allanton and Taieri at Waipiata which achieve a 'B' band. Lake Waihola returns an 'A' band for NH4-N toxicity.

The NOF attribute band for nitrate toxicity (measured as NNN) are shown in Figure 32. All sites return an 'A' band. The NPSFM (2020) describes this state as '*high conservation value system*. Unlikely to be effects even on sensitive species'.

Page 104 | 155

2000-2020

#### 11.1.3 Suspended fine sediment

The suspended fine sediment results for the Taieri FMU are shown in Figure 32. Ten sites return a NOF band of 'D' which the NPSFM 2020 describes as 'high *impact of suspended sediment on instream biota'*. five of which are mainstem Taieri sites; Taieri at Waipiata, Taieri at Tiroiti, Taieri at Sutton, Taieri at Outram and Taieri at Allanton. *At the other end of the scale, nine sites returned 'A' band,* they are all tributary sites and include Whare Creek, Waipori, Sutton Stream, Silverstream, Nenthorn, Kyeburn, Deepstream and 3 O'Clock Stream. It should be noted that Sutton Stream, Taieri at Outram and the Taieri at Tiroiti were monitored by NIWA as part of the NRQWN, the results are not modelled.

#### 11.1.4 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state of water quality and aquatic ecosystem health at a site. Figure 32 summarises MCI scores for the Taieri FMU. All sites achieve a 'C' band, returning MCI scores between 90 and 110; this reflects a macroinvertebrate community indicative of moderate organic pollution or nutrient enrichment. The ASPM interim scores shown in Figure 32 show that all sites achieve a 'B' band other than the Silverstream that obtains a band 'D' grade for ASPM, which the NPSFM 2020 describes as 'macroinvertebrate communities have severe loss loss of ecological integrity'. It is likely that the substrate in the Silverstream is largely to blame for poorer macroinvertebrate scores.

#### 11.1.5 Human health for recreation (Rivers)

Figure 31 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band. Compliance is generally good across the Taieri FMU, of the 22 sites, 12 achieve an 'A' band, three a 'B' band (Taieri main-stem sites at Linnburn, Sutton and Waipiata), the other sites returned bacterial water quality below the national bottom line (four 'D' bands and one an 'E' band). Lake Waihola an 'A' band mid lake, and a 'B' band at the edge of lake site (jetty).

Page 105 | 155
2000-2020

### 11.2 Trend Analysis

Trend analysis results for the Taieri FMU is shown in Figure 33 and Figure 34



Figure 33 Summary of Taieri FMU river sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the

Page 106 | 155

#### 2000-2020



monitoring). White cells indicate site/variables where there were insufficient data to assess the trend

Figure 34 Summary of Taieri FMU lake sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined inTable 2... Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend

Trend analysis for the Taieri FMU rivers is shown in Figure 33 and for Lake Waihola in Figure 34. Most sites have variables with 'exceptionally unlikely' or 'extremely unlikely' improving trends, these include:

- The Waipori River, TN (10-years)
- The Taieri at Waipiata, DRP (20-years)
- The Taieri at Tiroiti, DRP, E. coli, TN (10-years), DRP, TN (20-years)
- The Taieri at Stonehenge, DRP (20-years)
- The Taieri at Outram, DRP, E. coli, NNN, TN, TP, turbidity (10-years), DRP, TN, TP, turbidity (20-years)
- The Sutton Stream, DRP, E. coli, TN (10-years), DRP, TN TP (20-years)
- The Taieri at Linnburn, E. coli (20-years)
- The Taieri at Allanton, DRP, NNN, TN and turbidity (20-years)
- Sutton Stream, DRP, E. coli, TN (10-years), DRP, TN, TP (20-years)
- Silverstream at Taieri Depot, NNN, TN (20-years).
- Lake Waihola, DRP, E. coli, TP, turbidity (18 years)

Sites with 'virtually certain' or 'extremely likely' improving trends include:

- The Taieri at Waipiata, DRP, NNN, TP, turbidity (10-years), NH4-N (20-years)
- Taieri at Sutton, DRP, NNN (10-years)
- Taieri at Stonehenge, DRP, turbidity (10-years), NH4-N (20-years)
- Taieri at Linnburn and Allanton, NH4-N (20-years)
- Sutton Stream, NH4-N, NNN (10-years), NNN (20-years)

Page 107 | 155

#### 2000-2020

- Silverstream at Taieri and Deep Stream, turbidity (10-years)
- Kye Burn, DRP (10-years)
- Lake Waihola, NH4-N (20-years)

Page 108 | 155

#### 2000-2020

# 11.3 Water quality summary Taieri FMU

The tables in this section summarise:

- 1) River and lake sites where attributes where the national bottom line is not met (NPSFM, 2020)
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'
- 3) All trends using raw data for rivers and continuous data for lakes over the two time-periods

Figure 35 Summary of river state, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve band 'D'.

| sID                       | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|---------------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Taieri at Waipiata        |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Tiroiti         |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Sutton          |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Puketoi         |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Patearoa        |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Outram          |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Creamery        |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Taieri at Allanton Bridge |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Silverstream at Taieri    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Owhiro Stream             |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Meggat Burn               |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Contour Channel           |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Page 109 | 155

#### 2000-2020

| npID                 | nObs | Freq      | Period            | AnnualSenSlope | DirectionConf          | Descriptor |
|----------------------|------|-----------|-------------------|----------------|------------------------|------------|
|                      |      | D         | eep Strea         | m at SH87      |                        |            |
| Turbidity            | 35   | Qtr       | 10                | -0.05435       | Virtually certain      | 1111       |
|                      |      | Кує       | Burn at S         | SH85 Bridge    |                        | 1          |
| Dissolved Reactive P | 107  | Month     | 10                | 0              | Virtually certain      | 1111       |
|                      | 1    | Silver    | rstream a         | t Taieri Depot |                        | 1          |
| Nitrite/Nitrate N    | 101  | Month     | 10                | 0.023347       | Exceptionally unlikely | 1111       |
| Nitrite/Nitrate N    | 106  | BiMonth   | 20                | 0.016066       | Exceptionally unlikely |            |
| Total Nitrogen       | 101  | Month     | 10                | 0.025188       | Exceptionally unlikely | 1111       |
| Total Nitrogen       | 106  | BilVionth | 20                | 0.013899       | Exceptionally unlikely | 1111       |
| lurbidity            | 101  | Month     | 10                | -0.08362       | Virtually certain      | 11111      |
| A                    | 110  | Su        | tton Strea        | am at SH87     | Affair all social to   | 1          |
| Ammoniacal N         | 116  | Nonth     | 10                | -0.00103       | Virtually certain      |            |
| Dissolved Reactive P | 116  | Nonth     | 10                | 0              | Exceptionally unlikely | ++++       |
|                      | 236  | Nonth     | 20                | 0              | Exceptionally unlikely | ++++       |
| E. COII              | 116  | Nonth     | 10                | 5.941039       | Exceptionally unlikely | ++++       |
| Nitrite/Nitrate N    | 226  | Month     | 10                | -0.00103       | Virtually certain      |            |
| Total Nitragan       | 230  | Month     | 20                | -0.00023       |                        |            |
| Total Nitrogen       | 224  | Month     | 20                | 0.007475       | Exceptionally unlikely |            |
| Total Rhosphorus     | 234  | Month     | 20                | 0.00167        | Exceptionally unlikely |            |
|                      | 235  | Tai       | 20<br>ori at Alla | 0.000161       | Extremely unlikely     | 1 +++      |
| Ammoniacal N         | 112  | RiMonth   | 20                |                | Virtually cortain      | ****       |
| Dissolved Reactive R | 112  | BiMonth   | 20                | 0 00024        | Excontionally unlikely |            |
| Nitrito Nitrato N    | 112  | BiMonth   | 20                | 0.00024        | Exceptionally unlikely |            |
| Total Nitrogon       | 112  | BiMonth   | 20                | 0.002008       | Exceptionally unlikely |            |
| Turbidity            | 112  | BiMonth   | 20                | 0.00455        | Exceptionally unlikely |            |
|                      | 115  | Taiori    | 20                | Urn Runs Road  |                        | ++++       |
| Ammoniacal N         | 102  | BiMonth   | 20                |                | Virtually certain      | <u></u>    |
| E coli               | 102  | BiMonth   | 20                | 0 980764       | Extremely unlikely     | 4.4.4      |
| 2. 0011              | 102  | Bittionen | Taieri at         | Outram         | Extremely unintery     |            |
| Dissolved Reactive P | 119  | Month     | 10                | 0.000155       | Exceptionally unlikely | 1111       |
| Dissolved Reactive P | 239  | Month     | 20                | 0              | Exceptionally unlikely | 1111       |
| E. coli              | 119  | Month     | 10                | 2.165185       | Exceptionally unlikely | 1111       |
| Nitrite/Nitrate N    | 119  | Month     | 10                | 0.001067       | Exceptionally unlikely | 1111       |
| Total Nitrogen       | 119  | Month     | 10                | 0.019004       | Exceptionally unlikely | 1111       |
| Total Nitrogen       | 239  | Month     | 20                | 0.002825       | Exceptionally unlikely | 1111       |
| Total Phosphorus     | 119  | Month     | 10                | 0.001663       | Exceptionally unlikely | 1111       |
| Total Phosphorus     | 238  | Month     | 20                | 0.000205       | Extremely unlikely     | 111        |
| Turbidity            | 119  | Month     | 10                | 0.150039       | Exceptionally unlikely | 1111       |
| Turbidity            | 238  | Month     | 20                | 0.051847       | Exceptionally unlikely | 1111       |
|                      |      | Ta        | aieri at St       | onehenge       |                        |            |
| Ammoniacal N         | 118  | BiMonth   | 20                | 0              | Virtually certain      | 1111       |
| Dissolved Reactive P | 114  | Month     | 10                | -5.5E-05       | Extremely likely       | 111        |
| Dissolved Reactive P | 118  | BiMonth   | 20                | 0              | Extremely unlikely     | ttt        |
| Turbidity            | 102  | Month     | 10                | -0.05996       | Virtually certain      | 1111       |
|                      |      |           | Taieri at         | Sutton         |                        |            |
| Dissolved Reactive P | 113  | Month     | 10                | -0.00042       | Virtually certain      | 1111       |
| Nitrite/Nitrate N    | 113  | Month     | 10                | -0.00225       | Virtually certain      | 1111       |
|                      |      |           | Taieri a          | t Tiroiti      | -                      |            |
| Dissolved Reactive P | 119  | Month     | 10                | 0.000185       | Exceptionally unlikely | 1111       |
| Dissolved Reactive P | 239  | Month     | 20                | 0              | Exceptionally unlikely | 1111       |
| E. coli              | 119  | Month     | 10                | 3.858275       | Exceptionally unlikely | 1111       |
| Total Nitrogen       | 119  | Month     | 10                | 0.013466       | Exceptionally unlikely | 1111       |
| Total Nitrogen       | 239  | Month     | 20                | 0.006387       | Exceptionally unlikely | 1111       |
|                      |      |           | Taieri at         | Waipiata       |                        |            |

# Figure 36 Summary of river sites where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**.

Page 110 | 155

#### 2000-2020

| Ammoniacal N         | 113 | BiMonth | 20        | 0                 | Virtually certain      | <u> </u> |
|----------------------|-----|---------|-----------|-------------------|------------------------|----------|
| Dissolved Reactive P | 114 | Month   | 10        | -0.0006           | Virtually certain      | 1111     |
| Dissolved Reactive P | 113 | BiMonth | 20        | 0.000249          | Extremely unlikely     | ↓↓↓      |
| Nitrite/Nitrate N    | 114 | Month   | 10        | -0.00133          | Extremely likely       | <u>^</u> |
| Total Phosphorus     | 114 | Month   | 10        | -0.001            | Extremely likely       | <u>^</u> |
| Turbidity            | 102 | Month   | 10        | -0.20041          | Virtually certain      | <u> </u> |
|                      |     | Waipor  | i at Waip | ori Falls Reserve |                        |          |
| Total Nitrogen       | 101 | Month   | 10        | 0.005418          | Exceptionally unlikely | 1111     |

Figure 37 Summary lake sites where trends (continuous data) are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

|                      |      |         |                       | Trend      |                | Descriptor  |
|----------------------|------|---------|-----------------------|------------|----------------|-------------|
| npID                 | nObs | Period  | DirectionConf         | Direction  | AnnualSenSlope |             |
|                      |      | Lak     | e Waihola at End of j | etty       |                |             |
| Ammoniacal N         | 105  | BiMonth | 18                    | Decreasing | 0              | <b>111</b>  |
| Dissolved Reactive P | 105  | BiMonth | 18                    | Increasing | 0              | 1111        |
| E. coli              | 106  | BiMonth | 18                    | Increasing | 1.000684932    | 111         |
| Total Phosphorus     | 104  | BiMonth | 18                    | Increasing | 0.001271858    | 1111        |
| Turbidity            | 104  | BiMonth | 18                    | Increasing | 0.14570082     | <b>↓</b> ↓↓ |

Figure 38 Overall summary of trends for the Taieri FMU using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in **Error! Reference** source not found.

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descriptor             | 1111                 | 111                 | <b>1</b> 1  | Ŷ      | <b>↔</b>            | Ļ        | 11               | 111                   | 1111                      |
| Rivers - 10-year trend | 10                   | 3                   | 9           | 17     | 12                  | 13       | 2                |                       | 15                        |
| Rivers - 20-year trend | 4                    |                     | 1           | 4      | 4                   | 18       | 2                | 6                     | 12                        |
| Lakes – 10-year trend  |                      |                     | 2           | 3      | 1                   | 3        |                  |                       |                           |
| Lakes – 18 year trend  |                      |                     |             |        |                     | 3        |                  | 2                     | 2                         |

In the Taieri FMU water quality is generally good with the majority of sites and attributes achieving 'A' and 'B' bands, as seen in Figure 31. The exception is for the DRP and TN (periphyton) attributes which achieved mainly 'C' bands.

Of the tributaries, the lower Taieri plain has some of the poorest water quality in the region. Three streams are monitored in the plain, the Contour Channel, the Silverstream and the Owhiro Stream. Although the Upper Silverstream has good water quality and generally meets NOF attribute 'A' bands, the lower Silverstream has a poorer outcome. At this stage it has been straightened and has little riparian vegetation (flood banks) it is also influenced to some extent by storm-water from Mosgiel township. The Silverstream returned 'D' bands for ASPM, *E. coli* and TN (periphyton). Although the Silverstream achieves an 'A' band for NNN and a 'C' band for DRP, the levels are not low enough to prevent cyanobacteria (phormidium) growth. Factors facilitating this are likely to be the lack of shade

Page 111 | 155

#### 2000-2020

and few flushing flows which have proven to be ideal conditions for phormidium as this species of cyanobacteria tends to bloom most years.

The Contour Channel achieves a 'D' band for E.coli and DRP (periphyton), a 'C' band for DRP, TN (periphyton) and suspended fine sediment. The other stream that flows across the Taieri plain is the Owhiro Stream, this is a very small stream with areas of intensive agriculture in its catchment, it returns attribute bands below the national bottom line for DRP, E.coli, Periphyton (TN and DRP) and suspended fine sediment. Despite relatively good bacterial water quality throughout the reporting region, *E. coli* is the worst performing variable with five of 22 sites failing to meet the national bottom line, all of these sites are in the lower Taieri.

Lake Waihola shows nutrient and phytoplankton concentrations generally in the NOF 'C' bands, this is typical of a productive lake with increased levels of nutrients and algae over those that would be expected under natural or near natural conditions. Lake Waihola has episodic algal blooms typical of an eutrophic lake,

Trend analysis shows many degrading trends for the Taieri FMU. The Silverstream although returning band 'A's for NNN and DRP shows degrading trends for NNN. Of the main-stem Taieri sites, over the last 10-years, from the upper catchment to the lower catchment, Stonehenge a 'virtually certain' improving trend for turbidity, Waipiata has improving trends for DRP, NNN, TP and turbidity, Tiroiti has degrading trends for DRP, E.coli and TN, and Outram has degrading trends across most attribute states (*E. coli*, NN, TN, TP and turbidity).

#### In summary:

- Risk to ammonia and nitrate toxicity is negligible across the Taieri FMU.
- The lower Taieri has high concentrations of *E. coli*. Five sites in the lower Taieri fail to meet the national bottom line for *E. coli* the attribute. The rest of the sites monitored had excellent bacterial compliance, achieving an 'A' or 'B' band.
- The Owhiro Stream in the lower Taieri catchment has the worst level of compliance against NOF attribute states of any site across the Taieri River FMU.
- TN is of moderate concentrations across the catchment, no sites returned an 'A' band, but 10 sites achieved a 'B' band, only the Silverstream and Owhiro achieved 'D' bands.
- DRP generally complies with the NOF 'A' or 'B' band.
- Where macroinvertebrate monitoring takes place, MCI is generally 'fair'.
- Lake Waihola returns attribute bands consistent with those expected of an eutrophic lake.
- Trend analysis shows many degrading trends across the FMU, particularly in the lower Taieri at Outram.

Page 112 | 155

2000-2020

# 12 Dunedin Coast FMU



Figure 39 Location of water quality monitoring sites in the Dunedin Coast FMU

Page 113 | 155

2000-2020

#### 12.1 State Analysis Results

The results of grading the SoE sites in the Dunedin Coast FMU according to the NPSFM 2020 NOF criteria are summarised in Figure 40 and mapped in Figure 41. Many sites in the Dunedin Coast FMU did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements



Figure 40 Grading of the river sites of the Dunedin Coast FMU based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 114 | 155

#### 2000-2020



Figure 41 Maps showing Dunedin Coast FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

Page 115 | 155

2000-2020

#### 12.1.1 Periphyton and Nutrients

Results for the river periphyton trophic state results are shown in Figure 40 (periphyton). Periphyton trophic state results to date show that Akatore Creek is likely to be in attribute band 'C' for periphyton as results tend to be between >120 and  $\leq$ 200 chl- $a/m^2$  meaning moderate nutrient enrichment. The Kaikorai Stream is likely be in attribute band 'D' for periphyton as results tend to be >200 chl- $a/m^2$  reflecting high nutrient enrichment and the possibility of regular nuisance blooms.

To manage the NPSFM 2020 periphyton attribute state (MfE, 2020) median concentrations of DRP and TN align to attribute bands (i.e., periphyton DRP and periphyton TN in Figure 41). Using the 20% exceedance criteria (mid-range), the TN median concentrations in the Dunedin Coast FMU are generally in or greater than the T200 mg chl-a/m<sup>2</sup> band ('C' or 'D').

The DRP median concentrations in the Dunedin Coast FMU are generally in the T200 mg chl- $a/m^2$  band or greater (band 'C' or 'D').

Figure 41 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Dunedin Coast Rohe show that two sites achieve an 'A' band (Waitati and Akatore), one site a 'B' band (Kaikorai) and three sites a 'C' band (Leith at Dundas Street and Tokomairiro at Lisnatunny and Blackbridge). The NPSFM 2020 describes band 'C' as 'Ecological communities impacted by moderate DRP elevation above natural reference conditions. If other coditions also favour eutrophication, DRP enrichment may cause increased algal and plant growth, loss of sensitive macro-invertebrate and fish taxa, and high rates of respiration and decay'

The Leith achieves a 'D' band

#### 12.1.2 Toxicants (Rivers)

NOF attribute bands for NH4-N are shown in Figure 41, it should be noted that the national bottom line is below band 'B'. In the Dunedin Coast Rohe three sites have excellent protection levels against ammonia toxicity (Akatore Creek, Tokomairio at Lisnatunny and Tokomairio at West Branch Bridge) with all sites returning an 'A' band (highest level of protection) for NH4-N. Of the remaining sites, the Kaikorai Stream returned a 'C' band for the annual maximum which is below the national bottom line. The NPSFM 2020 describes the 'C' band as 'ammonia starts impacting regularly on the 20% most sensitive species (reduced survival of most sensitive species)'. The other three sites returned 'B' bands.

NOF attribute bands for nitrate (measured as NNN) toxicity are shown in Figure 41, again the national bottom line is below band 'B'. In the Dunedin Coast Rohe all sites achieve an 'A' band

#### 12.1.3 Suspended fine sediment (Rivers)

The clarity results for the Dunedin Coast Rohe are shown in Figure 41. Most of the sites return a NOF band of 'A' which denotes 'minimal impact of suspended sediment on instream biota. Ecological communities are similar to those observed in natural reference conditions' (NPSFM, 2020). The Tokomairio at West Branch Bridge and Lindsay's Creek return a NOF band of 'B' and the Tokomairio

Page 116 | 155

#### 2000-2020

at Blackbridge achieves a 'D' band, which the NPSFM 2020 describes as 'moderate to high impact of suspended sediment on instream biota. Sensitive fish species may be lost'

#### 12.1.4 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state if water quality and aquatic ecosystem health at a site. Figure 40 summarises MCI scores for sites monitored for aquatic macroinvertebrates throughout the Dunedin Coast Rohe.

The MCI interim scores are somewhat comparable across sites with two of the four monitored sites returning MCI scores at or below the national bottom line (MCI 90). The NPSFM 2020 describes this state as *'reflecting a macroinvertebrate community indicative of sever organic pollution or nutrient enrichment. Communities are largely composed of taxa insensitive to inorganic pollution/nutrient enrichment'*. Akatore Creek and the Tokomairiro at West Branch Bridge returned an MCI score above the national bottom line, achieving a 'C' grade.

The ASPM interim scores shown in Figure 40 generally reflect those of the MCI scores however Lindsay's Creek, despite having a 'D' grade for MCI, obtains an 'A' grade for ASPM, which the NPSFM 2020 describes as 'a community with mild to moderate loss of ecological integrity'.

#### 12.1.5 Human health for recreation (Rivers)

Figure 40 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band. Compliance is generally poor across the Dunedin Coast Rohe, with all sites other than the Waitati River returning bacterial water quality below the national bottom line.

All sites have high background bacteria concentrations which may indicate an *E. coli* source that is affecting water quality even under low flow conditions, such as the presence of water fowl or stock, the urban streams may receive E.coli from urban infrastructure (drains).

Page 117 | 155

2000-2020

# 12.2 Trend Analysis: Rivers



Trend analysis results for the Dunedin Coast FMU is shown in Figure 42

Figure 42 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend

Trend analysis for the Dunedin Coast Rohe rivers is shown in Figure 42, over a 10-year period the Kaikorai has an 'exceptionally unlikely' improving trend for TP and over a 20-year period an 'extremely unlikely' improving trend for NNN and TN.

In the Leith, over a 20-year period, there is an 'exceptionally unlikely' improving trend for DRP and NNN and in the Tokomairiro at West Branch Bridge, over a 10-year period there is a 'virtually certain' (improving) trend for DRP, but over the 20-year period there is an 'exceptionally unlikely' improving trend for NNN and TN.

Page 118 | 155

2000-2020

## 12.3 Water quality summary Dunedin Coast FMU

The tables in this section summarise:

- 1) River and lake sites where attributes do not meet the national bottom line (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) Summary of all trends using raw data for rivers and continuous data for lakes over the two time-period .

Table 24 Summary of river and lake state, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve band 'D'.

| sID                       | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|---------------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Tokomairiro at W Br Br    |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Tokomairiro at Lisnatunny |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Tokomairio at Blackbridge |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Lindsay's Creek           |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Leith at Dundas St        |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Kaikorai at Brighton Rd   |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Akatore Creek             |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Table 25 Summary of river sites (not flow adjusted) where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

| npID                 | nObs | Freq    | Period     | AnnualSenSlope    | DirectionConf          | Descriptor                         |
|----------------------|------|---------|------------|-------------------|------------------------|------------------------------------|
|                      |      | Kaikora | i Stream a | at Brighton Road  |                        |                                    |
| Ammoniacal N         | 116  | BiMonth | 20         | 0                 | Extremely likely       | <b>↑</b> ↑↑                        |
| Nitrite/Nitrate N    | 116  | BiMonth | 20         | 0.008079          | Extremely unlikely     | $\uparrow \uparrow \uparrow$       |
| Total Nitrogen       | 116  | BiMonth | 20         | 0.007256          | Extremely unlikely     | 111                                |
| Total Phosphorus     | 56   | BiMonth | 10         | 0.000904          | Exceptionally unlikely | 1111                               |
|                      |      | Leith   | at Dunda   | s Street Bridge   |                        |                                    |
| Ammoniacal N         | 115  | BiMonth | 20         | 0                 | Virtually certain      | <u> </u>                           |
| Dissolved Reactive P | 114  | BiMonth | 20         | 0.000377          | Exceptionally unlikely | $\uparrow\uparrow\uparrow\uparrow$ |
| Nitrite/Nitrate N    | 115  | BiMonth | 20         | 0.012718          | Exceptionally unlikely | 1111                               |
| Total Nitrogen       | 115  | BiMonth | 20         | 0.006574          | Extremely unlikely     | 111                                |
| Turbidity            | 115  | BiMonth | 20         | -0.04083          | Extremely likely       | <u>↑</u> ↑↑                        |
|                      |      | Tokomai | riro at W  | est Branch Bridge |                        |                                    |
| Ammoniacal N         | 68   | Qtr     | 20         | -0.00038          | Virtually certain      | <u> </u>                           |
| Dissolved Reactive P | 102  | Month   | 10         | -0.00038          | Virtually certain      | <u> </u>                           |
| Nitrite/Nitrate N    | 68   | Qtr     | 20         | 0.00677           | Exceptionally unlikely | 1111                               |
| Total Nitrogen       | 68   | Qtr     | 20         | 0.011153          | Exceptionally unlikely | 1111                               |
| Turbidity            | 68   | Qtr     | 20         | 0.046185          | Extremely unlikely     | 111                                |

Page 119 | 155

#### 2000-2020

# Table 26 Overall summary of trends for the Dunedin Coast FMU using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in *Error! Reference source not found.*.

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descriptor             | 1111                 | <b>11</b>           | <b>^</b>    | Ŷ      | <b>↔</b>            | Ļ        | 11               | 111                   | 1111                      |
| Rivers - 10-year trend | 1                    |                     | 4           | 7      | 9                   | 8        | 3                |                       | 1                         |
| Rivers - 20-year trend | 2                    | 2                   | 1           | 2      | 2                   | 2        | 2                | 4                     | 4                         |

In the Dunedin Coast FMU water quality generally has high bacteria and nutrient concentrations. The Kaikorai has an ammonia toxicity band of 'C' placing it below the national bottom line, the only site in Otago with this grade. *E. coli* was below the bottom line in seven of the eight sites monitored and TN returned a band 'D' in four of the eight sites monitored. The Kaikorai and Leith fail to meet the national bottom line for MCI.

These Dunedin City sites having a high degree of urbanisation with associated hardstanding requiring a stormwater network with point source discharges into the watercourses, these are likely to be conduits for contaminants during both wet and dry weather. Alongside the poor state, trend analysis shows that water quality in these urban streams continues to degrade. Over 20-years, the Kaikorai, Leith and Tokomairiro have degrading trends for TN and NNN, in addition the Kaikorai has a degrading trend for TP (10-years), the Leith for DRP (20-years) and the Tokomairio for turbidity (20-years).

There were three 'virtually certain' improving trends, two in the Tokomairiro at West Brach Bridge, and one in the Leith. When sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Dunedin Coast FMU, these sites include NH4-N in the Tokomairo at West Branch Bridge (20-years) and NH4-N in the Leith (10-years).

In summary:

- The Kaikorai has ammonia toxicity below the national bottom line.
- Nutrient concentrations are generally high, four of eight sites achieve a 'D' band for total nitrogen, the other sites return a 'C' band.
- Bacterial water quality is severely degraded at all monitoring sites other than the Waitati River.
- Trend analysis shows that in the Dunedin FMU there were more improving trends iln the last 10-years than degrading trends.

Page 120 | 155

2000-2020



# 13 North Otago FMU

Figure 43 Location of water quality monitoring sites in the North Otago FMU

Page 121 | 155

2000-2020

#### 13.1 State Analysis Results

The results of grading the SoE sites in the North Dunedin FMU according to the NPSFM 2020 NOF criteria are summarised in Figure 44 and mapped in Figure 45. Many sites in the North Otago FMU did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements.



Figure 44 Grading of the river sites of the North Otago FMU based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 122 | 155

#### 2000-2020



Figure 45 Maps showing North Otago FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines

Page 123 | 155

2000-2020

#### 13.1.1 Periphyton and Nutrients

Results for the river periphyton trophic state results are shown in Figure 44 (periphyton). Periphyton trophic state results to date show that the North Otago FMU returns mainly interim 'C' and 'D' bands. Kakanui at McCones, Oamaru Creek, Shag at Craig Road and the Waianakarua South Branch have interim results of 'D', below the national bottom line, this reflects an elevated nutrient enrichment and the possibility of regular nuisance blooms.

To manage the NPSFM 2020 periphyton attribute state (MfE, 2020) median concentrations of DRP and TN align to attribute bands (i.e., periphyton DRP and periphyton TN in Figure 44). Using the 20% exceedance criteria (mid-range), the TN median concentrations in the Dunedin Coast FMU are generally in or greater than the T200 mg chl-a/m<sup>2</sup> band (band 'C' or 'D'). The DRP median concentrations in the Dunedin Coast FMU are also generally in the T200 mg chl-a/m<sup>2</sup> band (equivalent to band 'C' or 'D'). The sites in band 'D' are the Awamoko, Kakaho Creek, Oamaru Creek and Waiareka Creek.

Figure 44 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the North Otago FMU show that most sites achieves band 'A'. Four sites, Awamoko, Kakaho Creek, Oamaru Creek, Waiareka Creek and Welcome Creek achieve a band 'D', which the NPSFM (2020) describes as 'ecological communities impacted by substantial DRP elevation above natural reference conditions'.

#### 13.1.2 Toxicants (Rivers)

NOF attribute bands for NH4-N are shown for the North Otago sites in Figure 44, it should be noted that the national bottom line is below band 'B'. In the North Otago FMU 10 sites have excellent protection levels against ammonia toxicity with all sites returning an 'A' band (highest level of protection) for NH4-N. The remaining sites, returned 'B' bands for the annual maximum. The NPSFM 2020 describes the 'B' band as 'ammonia starts impacting occasionally on the 5% most sensitive species'.

NOF attribute bands for nitrate (measured as NNN) toxicity are given for North Otago sites in Figure 44, again the national bottom line is below band 'B'. In the North Otago FMU all sites achieve an 'A' band, other than Welcome Creek which achieves a 'B' band, the NPSFM 2020 describes 'B' band as NNN having 'some growth effect on up to 5% of species'

#### 13.1.3 Suspended fine sediment (Rivers)

The clarity results for the North Otago FMU are shown in Figure 44. All sites return a NOF band of 'A' which denotes 'minimal impact of suspended sediment on instream biota. Ecological communities are similar to those observed in natural reference conditions' (NPSFM, 2020).

#### 13.1.4 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state of water quality and aquatic ecosystem health at a site. Figure 45 summarises MCI scores for sites monitored for aquatic macroinvertebrates throughout the North Otago FMU.

The MCI interim scores are low across all sites, five sites achieve a band 'C' and five are below the national bottom line, achieving a band 'D' which with NPSFM (2020) describes as a 'macroinvertebrate community indicative of severe organic pollution or nutrient enrichment'.

Page 124 | 155

#### 2000-2020

The ASPM interim scores shown in Figure 44; four of the sites obtains a 'B' band however another four sites have MCI scores below the national bottom line. The Waikouaiti despite obtaining a 'C' grade for MCI, obtains a 'D' grade for ASPM.

#### 13.1.5 Human health for recreation

Figure 44 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

Compliance is in the North Otago FMU is mixed, with eight of 16 sites returning bacterial water quality below the national bottom line. The NPSFM 2020 describes band 'D' as '30% of the time the estimated risk is  $\geq$ 50 in 1,000 (>5% risk). The predicted average infection >3%'. Only the three Shag River sites achieved an 'A' band.

### 13.2 Trend Analysis: Rivers

Trend analysis results for the North Otago FMU is shown in Figure 46



Page 125 | 155

#### 2000-2020

Figure 46 Summary of North Otago FMU sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend

Trend analysis for the North Otago FMU rivers is shown in Figure 46. 'Virtually certain' or 'exceptionally unlikely' trends are indicated at most sites. Sites with 'exceptionally unlikely' or 'extremely unlikely' improving trends include:

- Waikouaiti at Confluence, turbidity (10-years).
- Waiareka Creek, E. coli, NNN, TN (10-years), E. coli, NNN, TN, turbidity (20-years).
- Waianakarua at Browns, NNN and over 20-years for *E.coli*, NNN (Figure 47), TN and turbidity Kakanui at Clifton over 10-years for *E.coli* and over 20-years for NNN.
- Trotters Creek, turbidity (10-years).
- Shag at Goodwood, DRP, turbidity (20-years), Shag at Craig Road, NNN, TN, turbidity (20-years).
- Kakanui at McCones, E. coli, NNN, TN (20-years).
- Kakanui at Clifton, E. coli (10-years), E. coli, NNN (20-years).

Sites with 'virtually certain' or 'extremely likely 'improving trends include:

- Waikouaiti at Conflucence, DRP (10-years).
- Kakanui at McCones, turbidity (10-years).
- NH4-N (20-years) for Waiareka Creek, Waianakarua at Browns, Shag at Craig Road and Kakanui at Clifton Falls.

Page 126 | 155

2000-2020



Figure 47 Waianakarua River, 10-year trend in Nitrite/Nitrate Nitrogen

Page 127 | 155

2000-2020

## 13.3 Water quality summary North Otago FMU

The tables in this section summarise:

- 1) River and lake sites where attributes do not meet the national bottom line (NPSFM, 2020).
- 2) Trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'.
- 3) Summary of all trends using raw data for rivers and continuous data for lakes over the two time-periods.

Table 27 Summary of river and lake state, red cells show where state does not meet the national bottom line in one or more variable. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve a band 'D'.

| sID                | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|--------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Welcome Creek      |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Waikouaiti         |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Waiareka Creek     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Waianakarua        |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Trotters Creek     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Shag at Goodwood   |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Shag at Craig Road |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Pleasant River     |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Oamaru Creek       |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Kakanui at McCones |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Kakanui at Clifton |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Kakaho Creek       |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Awamoko            |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Page 128 | 155

#### 2000-2020

| npID                 | nObs | Freq    | Period      | AnnualSenSlope   | DirectionConf          | Descriptor                            |
|----------------------|------|---------|-------------|------------------|------------------------|---------------------------------------|
|                      |      | Kakar   | ui at Clift | ton Falls Bridge |                        |                                       |
| Ammoniacal N         | 111  | BiMonth | 20          | 0                | Virtually certain      | <u>^</u>                              |
| Dissolved Reactive P | 104  | Month   | 10          | 0                | Extremely likely       | <b>^^</b>                             |
| E. coli              | 109  | Month   | 10          | 5.728637         | Exceptionally unlikely | 1111                                  |
| E. coli              | 110  | BiMonth | 20          | 1.839            | Extremely unlikely     | 111                                   |
| Nitrite/Nitrate N    | 111  | BiMonth | 20          | 0.00064          | Exceptionally unlikely | 1111                                  |
|                      |      | ķ       | Kakanui at  | t McCones        | · · · · ·              |                                       |
| E. coli              | 114  | BiMonth | 20          | 2.005766         | Exceptionally unlikely | 1111                                  |
| Nitrite/Nitrate N    | 114  | BiMonth | 20          | 0.013947         | Exceptionally unlikely | 1111                                  |
| SQMCI Score          | 10   | Year    | 10          | -0.1021          | Very unlikely          | 11                                    |
| Total Nitrogen       | 113  | BiMonth | 20          | 0.016878         | Exceptionally unlikely | 1111                                  |
| Turbidity            | 102  | Month   | 10          | -0.03946         | Virtually certain      | 1111                                  |
|                      |      |         | Kauru a     | t Ewings         |                        |                                       |
| Dissolved Reactive P | 57   | BiMonth | 10          | 0                | Extremely likely       | 111                                   |
| E. coli              | 102  | Month   | 10          | 2.461253         | Extremely unlikely     | 111                                   |
|                      |      |         | Shag at C   | raig Road        | •                      |                                       |
| Ammoniacal N         | 110  | BiMonth | 20          | 0                | Virtually certain      | 1111                                  |
| Nitrite/Nitrate N    | 110  | BiMonth | 20          | 0.005144         | Exceptionally unlikely | 1111                                  |
| Total Nitrogen       | 110  | BiMonth | 20          | 0.008768         | Exceptionally unlikely | 1111                                  |
| Turbidity            | 110  | BiMonth | 20          | 0.020714         | Exceptionally unlikely | 1111                                  |
|                      |      | Sha     | ig at Good  | dwood Pump       |                        |                                       |
| Dissolved Reactive P | 103  | Month   | 10          | -0.00024         | Extremely likely       | 111                                   |
| Dissolved Reactive P | 111  | BiMonth | 20          | 0                | Extremely unlikely     | 111                                   |
| Turbidity            | 114  | BiMonth | 20          | 0.012219         | Extremely unlikely     | 111                                   |
|                      |      | Trott   | ers Creek   | at Mathesons     | •                      |                                       |
| Turbidity            | 58   | BiMonth | 10          | 0.066027         | Extremely unlikely     | 111                                   |
|                      |      | Wa      | aianakaru   | ia at Browns     |                        |                                       |
| Ammoniacal N         | 114  | BiMonth | 20          | 0                | Virtually certain      | 1111                                  |
| E. coli              | 100  | Month   | 10          | 8.860577         | Exceptionally unlikely | $\uparrow \uparrow \uparrow \uparrow$ |
| E. coli              | 113  | BiMonth | 20          | 1.501027         | Exceptionally unlikely | 1111                                  |
| Nitrite/Nitrate N    | 102  | Month   | 10          | 0.013059         | Exceptionally unlikely | $\uparrow \uparrow \uparrow \uparrow$ |
| Nitrite/Nitrate N    | 111  | BiMonth | 20          | 0.007787         | Exceptionally unlikely | $\uparrow \uparrow \uparrow \uparrow$ |
| Total Nitrogen       | 102  | Month   | 10          | 0.013379         | Extremely unlikely     | 111                                   |
| Total Nitrogen       | 111  | BiMonth | 20          | 0.007521         | Exceptionally unlikely | $\uparrow \uparrow \uparrow \uparrow$ |
| Turbidity            | 114  | BiMonth | 20          | 0.012135         | Exceptionally unlikely | 1111                                  |
|                      | -    | Waiar   | reka Cree   | k at Taipo Road  |                        |                                       |
| Ammoniacal N         | 114  | BiMonth | 20          | -0.00079         | Virtually certain      | <u> </u>                              |
| Dissolved Reactive P | 105  | Month   | 10          | 0.008087         | Exceptionally unlikely | 1111                                  |
| E. coli              | 114  | BiMonth | 20          | 6.725544         | Exceptionally unlikely | 1111                                  |
| Nitrite/Nitrate N    | 114  | BiMonth | 20          | 0.022186         | Exceptionally unlikely | 1111                                  |
| Total Nitrogen       | 114  | BiMonth | 20          | 0.03203          | Exceptionally unlikely | 1111                                  |
| Total Phosphorus     | 106  | Month   | 10          | 0.00644          | Extremely unlikely     | 111                                   |
| Turbidity            | 114  | BiMonth | 20          | 0.037376         | Exceptionally unlikely | 1111                                  |
|                      |      | Waik    | ouaiti at ( | Confluence d/s   |                        |                                       |
| Dissolved Reactive P | 32   | Qtr     | 10          | 0                | Virtually certain      | <u> </u>                              |
| Turbidity            | 48   | BiMonth | 10          | 0.04634          | Extremely unlikely     | ttt                                   |

# Table 28 Summary of river sites where trends are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**

Page 129 | 155

2000-2020

# Table 29 Overall summary of trends for the North Otago FMU using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in *Error! Reference source not found.*.

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descriptor             | 1111                 | <b>11</b>           | <b>1</b> 1  | Ŷ      | <b>↔</b>            | Ļ        | 11               | 111                   | 1111                      |
| Rivers - 10-year trend | 2                    | 3                   | 9           | 13     | 17                  | 15       | 4                | 5                     | 4                         |
| Rivers - 20-year trend | 4                    |                     | 1           | 6      | 4                   | 3        | 1                | 3                     | 15                        |

In the North Otago FMU all sites other than the Kauru and Upper Shag return at least one attribute band below the national bottom line. All sites return 'A' or 'B' bands for ammonia and nitrate toxicity, and all sites return 'A' bands for suspended fine sediment. 'D' bands are returned for DRP in four of the 16 sites, for *E. coli* in eight of the 16 sites and for periphyton in four of the eight sites monitored. Macroinvertebrate metrics were generally in the 'C' or 'D' bands.

The urban stream, Oamaru Creek returns the most 'D' bands, likely due to the influence or urban runoff on the Creek, Waiareka Creek and the Awamoko also return mostly 'D' bands but these sites are in a rural setting.

Trend analysis identifies many 'exceptionally unlikely' improving trends over both the 10 and 20-year periods. In the last 10-years, four sites continue to show 'exceptionally unlikely' improving trends, Clifton Falls (*E. coli*), Waianakarua (*E. coli*, NNN, TN) and Waiareka Creek (DRP, TP). The source of *E. coli* at Kakanui at Clifton has been identified as red billed gulls roosting in the gorge upstream of the monitoring site. When sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the North Otago FMU, these sites include NH4-N and DRP at the Kakanui at Clifton site, DRP at Ewings, NH4-N at the Shag at Craig Road and the Waianakarua, and DRP at the Waikouaiti and the Shag at Goodwood.

Previous reports have identified land-use intensification as a driver of poor water quality however ORC do not collect detailed information on land-use, land management practices or changes in either of the two that allow for inference as to the drivers of degrading or improving trends in water quality.

In summary:

- All sites are in the 'A' or 'B' band for ammonia and nitrate toxicity
- TN concentrations are generally low enough to meet the 20% spatial exceedance criteria for periphyton growth. DRP does not meet this criteria at four sites (Awamoko, Kakaho, Oamaru Creek and Waiareka Creek).
- Bacterial water quality is severely degraded at eight of the 15 monitoring sites. The three Shag River sites attain an 'A' grade for E. coli.

Page 130 | 155

#### 2000-2020

- Trend analysis shows that in the North Otago there were fewer degrading trends over the last 10-years compared to the last 20-years. In the last 10-years there are also more improving trends than degrading trends.
- Over the last 10-years, the Waianakarua has three 'extremely likely' or 'exceptionally unlikely' improving trends, more than any of the other North Otago sites.

Page 131 | 155

2000-2020

# 14 Catlins FMU



Figure 48 Location of water quality monitoring sites in the Catlins FMU

Page 132 | 155

2000-2020

#### 14.1 State Analysis Results

The results of grading the SoE sites in the Catlins FMU based on the NPSFM 2020 NOF criteria are summarised in Figure 49 and mapped in Figure 50. Many sites in the Catlins FMU did not meet the sample number requirements (shown in Table 1) and accordingly are shown as white cells with coloured circles. Most sites for some variables have white cells, this indicates that the variable was not monitored.

A small square in the upper left quadrant of the cells indicate the site grade for the baseline period (2012-2017) where the sample numbers for that period met the minimum sample number requirements



Figure 49 Grading of the river sites of the Catlins FMU based on the NOF criteria. Grades for sites that did not meet the sample number requirements in Table 1 are shown as white cells with coloured circles. The white cells indicate sites for which the variable was not monitored. Small square in the upper left quadrant of the cells indicate the site grade for the baseline

Page 133 | 155

2000-2020



Figure 50 Maps showing Catlins FMU sites coloured according to their state grading as indicated by NOF attribute bands. Bands for sites that did not meet the sample number requirements specified in Table 1 are shown without black outlines.

Page 134 | 155

2000-2020

#### 14.1.1 Periphyton and Nutrients

Periphyton trophic state results to date are given in Figure 49 and show that of the two sites monitored in the Catlins FMU, the Tahakopa returns an interim 'A' band as few results exceed 50 chl- $a/m^2$  reflecting low nutrient enrichment and the Owaka returned a 'C' band reflecting a more nutrient rich environment.

To manage the NPSFM 2020 periphyton attribute state (MfE, 2020) median concentrations of DRP and TN align to attribute bands (i.e., periphyton DRP and periphyton TN in Figure 49). Using the 20% exceedance criteria (mid-range), the TN median concentrations in the Catlins FMU are generally in or greater than the T200 mg chl-a/m<sup>2</sup> band ('C' or 'D'). The DRP median concentrations in the Dunedin Coast FMU are generally in the T200 mg chl-a/m<sup>2</sup> band (band 'C').

Figure 49 also shows DRP attribute states for ecosystem health (DRP median and Q95). The results in the Catlins Rohe show that one site achieves an 'A' band (Tahakopa), one site a 'B' band (Maclennan) and two sites a 'C' band (Catlins and Owaka). The NPSFM 2020 describes band 'C' as 'Ecological communities impacted by moderate DRP elevation above natural reference conditions. If other coditions also favour eutrophication, DRP enrichment may cause increased algal and plant growth, loss of sensitive macro-invertebrate and fish taxa, and high rates of respiration and decay'

#### 14.1.2 Toxicants (Rivers)

NOF attribute bands for NH4-N are given in Figure 49, the national bottom line for toxicants is below band 'B'. The Catlins FMU has three sites returning an 'A' band (highest level of protection) for NH4-N. The remaining site (the Catlins) returned a 'B' band for the annual maximum. The NPSFM 2020 describes the 'B' band as 'ammonia starts impacting occasionally on the 5% most sensitive species'.

NOF attribute bands for nitrate (measured as NNN) toxicity are given in Figure 49Error! Reference source not found.. In the Catlins FMU all sites achieve an 'A' band, other than the Owaka which achieves a 'B' band across both statistical metrics, the NPSFM 2020 describes 'B' band as NNN having 'some growth effect on up to 5% of species'

#### 14.1.3 Suspended fine sediment (Rivers)

The suspended fine sediment results for the Catlins FMU are shown in Figure 49. Most sites return a NOF band of 'D' which denotes 'high limpact of suspended sediment on instream biota. Ecological communities are significantly altered and sensitive fish and macroinvertebrate species are lost or at high risk of being lost' (NPSFM, 2020). The Maclennan returns a band of 'B'.

#### 14.1.4 Aquatic Life (Rivers)

Macroinvertebrate Community Index (MCI) scores provide an integrated indicator of the general state of water quality and aquatic ecosystem health at a site. Figure 49Error! Reference source not found.

Page 135 | 155

#### 2000-2020

summarises MCI scores for sites monitored for aquatic macroinvertebrates throughout the Catlins FMU.

All sites return an MCI score between 90 and 110, or a band 'C' reflecting a macroinvertebrate community indicative of moderate organic pollution or nutrient enrichment. Figure 49 also summarises ASPM scores for sites monitored for aquatic macroinvertebrates throughout the Catlins FMU. All sites return an ASPM score between 0.4 and 0.6, or a band 'B' reflecting a macroinvertebrate community indicative of mild to moderate loss of ecological integrity.

#### 14.1.5 Health for recreation (Rivers)

Figure 49 summarises compliance for *E. coli* against the four statistical tests of the NOF *E. coli* attribute. The overall attribute state is based on the worst grading with the national bottom line being a 'D' band.

Compliance is quite poor across the Catlins FMU, with two sites; Owaka and Tahakopa, returning bacterial water quality below the national bottom line on all four statistical metrics. The other two sites return an overall 'C' band despite returning 'A' band median scores.

Page 136 | 155

#### 2000-2020

#### 14.2 Trend Analysis Results – Rivers

Trend analysis results for the Catlins River is shown in Figure 51. Over a 20-year period the Catlins has 'exceptionally unlikely' improving trends for *E. coli*, NNN and TN, with an 'extremely unlikely' improving trend for TP. In the shorter timefrme there are three 'extremely likely' or 'virtually certain' improving trends for NH4-N, DRP and turbidity. There are no degrading trends in the last 10-years.



Figure 51 Summary of Upper Clutha sites categorised according to the level of confidence that their 10 and 20-year raw water quality trends indicate improvement. Confidence that the trend indicates improvement is expressed using the categorical levels of confidence defined in Table 2. Cells containing a black dot indicate site/variable combinations where the Sen Slope was evaluated as zero (i.e., a trend rate that cannot be quantified given the precision of the monitoring). White cells indicate site/variables where there were insufficient data to assess the trend

#### 14.3 Water quality summary Catlins FMU

The tables in this section summarise:

- 1) river and lake sites where attributes where the national bottom line is not met (NPSFM, 2020)
- 2) trends in river and lake sites when the trends are greater than 'likely' or 'unlikely'
- 3) all trends using raw data for rivers and continuous data for lakes over the two time-periods

Table 30 Summary of river and lake state, red cells show where state does not meet the national bottom line in one or more variable

Page 137 | 155

#### 2000-2020

| sID                  | NH4-N - max | NH4-N - median | ASPM | DRP - median | DRP – Q95 | E.coli | MCI | NNN - median | NNN – Q95 | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |
|----------------------|-------------|----------------|------|--------------|-----------|--------|-----|--------------|-----------|------------|------------------|-----------------|----------------------------|
| Tahakopa at Tahakopa |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Owaka at Katea Road  |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |
| Catlins at Houipapa  |             |                |      |              |           |        |     |              |           |            |                  |                 |                            |

Table 31 Summary of river sites where trends (raw data) are greater than 'likely' or 'unlikely'. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

| npID                 | nObs | Freq    | Period | AnnualSenSlope | DirectionConf          | Descriptor |  |
|----------------------|------|---------|--------|----------------|------------------------|------------|--|
| Catlins at Houipapa  |      |         |        |                |                        |            |  |
| Ammoniacal N         | 102  | Month   | 10     | 0              | Extremely likely       | 111        |  |
| Dissolved Reactive P | 101  | Month   | 10     | -0.00032       | Extremely likely       | 111        |  |
| E. coli              | 114  | BiMonth | 20     | 3.066082       | Exceptionally unlikely | 1111       |  |
| Nitrite/Nitrate N    | 114  | BiMonth | 20     | 0.007367       | Exceptionally unlikely | 1111       |  |
| Total Nitrogen       | 114  | BiMonth | 20     | 0.007508       | Exceptionally unlikely | 1111       |  |
| Total Phosphorus     | 113  | BiMonth | 20     | 0.000292       | Extremely unlikely     | 111        |  |
| Turbidity            | 101  | Month   | 10     | -0.14904       | Extremely likely       | <b>^^</b>  |  |

Table 32 Overall summary of trends for the Catlins FMU using raw data for rivers and continuous data for lakes. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** 

|                        | Virtually<br>certain | Extremely<br>likely | Very likely | Likely | As likely as<br>not | Unlikely | very<br>unlikely | Extremely<br>unlikely | Exceptionally<br>unlikely |
|------------------------|----------------------|---------------------|-------------|--------|---------------------|----------|------------------|-----------------------|---------------------------|
| Descriptor             | 1111                 | <b>111</b>          | <b>^</b>    | 1      | ↔                   | Ļ        | 11               | 111                   | 1111                      |
| Rivers - 10-year trend |                      | 3                   | 2           |        | 2                   |          |                  |                       |                           |
| Rivers - 20-year trend |                      |                     |             |        | 1                   | 1        | 1                |                       | 3                         |

The Catlins FMU is expected to have good water quality, due to the intact nature of the headwaters and native vegetation, however cleared valleys allow intensive farming activities. When comparing to the NOF attribute states, water quality is variable. All sites return 'A' or 'B' bands for ammonia and nitrate toxicity. The Owaka and Tahakopa return 'D' bands for *E. coli* and 'C' bands are returned for MCI at all sites. Suspended fine sediment returns 'D' bands at all sites other than the Maclennan, which achieves an 'A' band.

There were three 'extremely likely' or 'very likely' improving trends in the Catlins in the last 10-years (NH4-N, DRP and turbidity). When sites have a zero sen slope alongside a reasonably high-level of confidence in trend direction the rate of the trend (i.e., the Sen slope) is at a level that is below the detection precision of the monitoring programme. In the Catlins River, this was true for NH4-N, (Table 31).

In summary:

Page 138 | 155

#### 2000-2020

- All sites are in the 'A' or 'B' band for ammonia and nitrate toxicity.
- TN and DRP concentrations are low enough to meet the 20% spatial exceedance criteria for periphyton growth, other than TN at Owaka.
- Bacterial water quality is degraded in the Owaka and Tahakopa.
- Trend analysis for the Catlins shows four 'extremely, or exceptionally unlikely' improving trends over 20-years for *E. coli*, NNN, TP and TN.
- There were no degrading trends over the last 10-years.

Page 139 | 155

2000-2020

# 15 Otago Summary

This section gives an overview of water quality state and water quality trends across Otago.

#### 15.1 Otago Overview: Water Quality State

Table 33 shows a summary of lake sites where attributes fall below the national bottom line. All lakes do not meet the NPSFM 2020 national bottom line for chlorophyll-a and Lake Tuakitoto does not meet the NPSFM 2020 national bottom line for TN and TP.

| FMU/Rohe          | siD                         | Chlorophyll a max | Chlorophyll a median | N | ΤP |
|-------------------|-----------------------------|-------------------|----------------------|---|----|
|                   | Lake Johnson at South Beach |                   |                      |   |    |
| Dunstan Rohe      | Lake Hayes at Mid Lake 10m  |                   |                      |   |    |
| Dunedin Coast FMU | Lake Tuakitoto at Outlet    |                   |                      |   |    |

Table 33 Summary of lake sites where attributes fall below the NPSFM 2020 national bottom line

Table 34 details all river sites where attributes do not meet the NPSFM 2020 bottom line. The table shows that 46 sites do not meet the NPSFM 2020 bottom line for E.coli and 40 sites did not meet the NPSFM 2020 bottom line for suspended fine sediment and for DRP 14 sites were in band 'D' (ecosystem health) attribute.

The Ministry for the Environment produced guidance (MfE, 2020) for defining nutrient concentrations to manage the NPSFM 2020 periphyton attribute states in rivers. The guidance is centered around spatial exceedances for TN and DRP. TN and DRP median concentrations are compared to the spatial exceedance criteria of 20%. At this level there is some risk (ie, 20%) that the chlorophyll *a* response at some sites will exceed the desired chlorophyll *a* threshold, even if the DRP or TN concentration targets are achieved. Results in Table 34 show that 25 sites for TN and 23 sites for DRP were elevated above the the 20% exceedance criteria.

Page 140 | 155

#### 2000-2020

Table 34 Summary of river sites where attributes are below the NPSFM 2020 national bottom line. There is no national bottom line for DRP, but DRP (median and Q95) have been included in the table when sites achieve a band 'D'.

Page 141 | 155
### 2000-2020

| FMU/Rohe             | siD   | WdSA | DRP - median | 560 – dyd | E.coli | MCI | Periphyton | Periphyton (DRP) | Periphyton (TN) | Suspended fine<br>sediment |     |
|----------------------|---|------|--------------|-----------|--------|-----|------------|------------------|-----------------|----------------------------|-----|
|                      | Timaru Creek  |      |              |           |        |     |            |                  |                 |                            | l l |
| he                   | Rees River  |      |              |           |        |     |            |                  |                 |                            |     |
| Ro                   | Ox Burn   |      |              |           |        |     |            |                  |                 |                            |     |
| kes                  | Makarora River  |      |              |           |        |     |            |                  |                 |                            | l l |
| r La                 | Invincible Creek  |      |              |           |        |     |            |                  |                 |                            |     |
| ppe                  | Horn Creek<br>Dart River  |      |              |           |        |     |            |                  |                 |                            |     |
| ∍                    | Bullock Creek   |      |              |           |        |     |            |                  |                 |                            | Ì   |
|                      | Buckler Burn  |      |              |           |        |     |            |                  |                 |                            |     |
| he                   | Upper Cardrona  |      |              |           |        |     |            |                  |                 |                            |     |
| n Rc                 | Mill Creek  |      |              |           |        |     |            |                  |                 |                            |     |
| sta                  | Lindis at Peak  |      |              |           |        |     |            |                  |                 |                            | l l |
| unc                  | Kawarau at Chards   |      |              |           |        |     |            |                  |                 |                            |     |
| -                    | Thomsons Creek at SH85  |      |              |           |        |     |            |                  |                 |                            |     |
| ohe                  | Poolburn at Cob Cottage   |      |              |           |        |     |            |                  |                 |                            |     |
| ia F                 | Manuherikia d/s Fork  |      |              |           |        |     |            |                  |                 |                            |     |
| erek                 | Manuherikia at Ophir<br>Manuherikia at Galloway                 |      |              |           |        |     |            |                  |                 |                            |     |
| nhe                  | Manuherikia at Blackstone                                       |      |              |           |        |     |            |                  |                 |                            |     |
| lan                  | Hills Creek   |      |              |           |        |     |            |                  |                 |                            | l l |
| 4                    | Dunstan Creek   |      |              |           |        |     |            |                  |                 |                            |     |
| βini                 | Fraser at Old Mans Range  |      |              |           |        |     |            |                  |                 |                            |     |
| dxo                  | Clutha at Millers Flat  |      |              |           |        |     |            |                  |                 |                            | Ì   |
| R                    | Benger burn at SH8  |      |              |           |        |     |            |                  |                 |                            | l l |
|                      | Waiwera at Maws Farm  |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Waitahuna at Tweeds Bridge                                      |      |              |           |        |     |            |                  |                 |                            |     |
| _                    | Wairuna at Millar Road  |      |              |           |        |     |            |                  |                 |                            |     |
| ohe                  | Waipahi at Waipahi  |      |              |           |        |     |            |                  |                 |                            |     |
| aR                   | Waipani at Cairns Peak<br>Unner Pomahaka at Aitchison Runs Road |      |              |           |        |     |            |                  |                 |                            |     |
| uth                  | Tuapeka at 700m u/s bridge                                      |      |              |           |        |     |            |                  |                 |                            | Ì   |
| LC CI                | Pomahaka at Glenken   |      |              |           |        |     |            |                  |                 |                            |     |
| 9MG                  | Pomahaka at Burkes Ford   |      |              |           |        |     |            |                  |                 |                            |     |
| P                    | Heriot Burn at Park Hill Road                                   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Crookston Burn at Kelso Road                                    |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Clutha at Balclutha   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Taieri at Waipiata  |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Taieri at Tiroiti   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Taieri at Sutton  |      |              |           |        |     |            |                  |                 |                            |     |
| ⊇                    | Taieri at Puketoi<br>Taieri at Patearoa                         |      |              |           |        |     |            |                  |                 |                            |     |
| FΜ                   | Taieri at Outram  |      |              |           |        |     |            |                  |                 |                            |     |
| ieri                 | Taieri at Creamery  |      |              |           |        |     |            |                  |                 |                            |     |
| Ta                   | Taieri at Allanton Bridge<br>Silverstream at Taieri Dopot       |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Owhiro Stream   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Meggat Burn   |      |              |           |        |     |            |                  |                 |                            |     |
| <b>Dunedin Coast</b> | Contour Channel   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Tokomairiro at Lisnatunny                                       |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Tokomairio at Blackbridge                                       |      |              |           |        |     |            |                  |                 |                            | l l |
|                      | Lindsay's Creek   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Leith at Dundas St<br>Kaikorai at Brighton Bd                   |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Akatore Creek   |      |              |           |        |     |            |                  |                 |                            | Ì   |
|                      | Welcome Creek   |      |              |           |        |     |            |                  |                 |                            | l l |
| _                    | Waixouaiti<br>Waiareka Creek                                    |      |              |           |        |     |            |                  |                 |                            | ł   |
| MU                   | Waianakarua   |      |              |           |        |     |            |                  |                 | -                          | ł   |
| 0 F                  | Trotters Creek  |      |              |           |        |     |            |                  |                 |                            | i i |
| Itag                 | Shag at Goodwood  |      |              |           |        |     |            |                  |                 |                            |     |
| North Ot             | Shag at Craig Road<br>Pleasant River                            |      |              |           |        |     |            |                  |                 |                            | ł   |
|                      | Oamaru Creek  |      |              |           |        |     |            |                  |                 |                            | Ì   |
|                      | Kakanui at McCones  |      |              |           |        |     |            |                  |                 |                            | l l |
|                      | Kakanui at Clifton  |      |              |           |        |     |            |                  |                 |                            |     |
|                      | Awamoko   |      |              |           |        |     |            |                  |                 |                            |     |
| ٦S                   | Tahakopa at Tahakopa  |      |              |           |        |     |            |                  |                 |                            | l l |
| atli                 | Owaka at Katea Road   |      |              |           |        |     | ļ          | P                | uge             |                            | 155 |
| J                    | Catility at Houlpapa  | I    |              | l         | I      | l   | l          |                  | 0 -             |                            | 1   |

#### 2000-2020

## 15.2 Otago Overview: Water Quality Trends

Figure 52 and Figure 53 show colour coded bar charts representing the proportions of sites with trends indicating improving and degrading water quality based on the categories defined in Table 2 for the 10- and 20-year trend periods, respectively. These plots show that 20-year trends were predominantly degrading for all variables apart from ammoniacal nitrogen and for the 10-year trend period the predominant trend direction was variable by water quality variable. Interpretation of these plots should also take into account that there were variable nmbers of sites included in the different time-periods (Table 3)



Figure 52 Summary plot representing the proportion of river sites with improving 10-year time-period trends at each categorical level of confidence. The plot shows the proportion of sites for which the trend indicated improving water quality at levels of confidence defined in Table 2. Green colours indicate sites with improving trends, and red-orange colours indicate sites with degrading trends. Trends used in this graph are not flow-adjusted, and have a hi-censor filter applied.

Page 143 | 155



Figure 53 Summary plot representing the proportion of river sites with improving 20-year time-period trends at each categorical level of confidence. The plot shows the proportion of sites for which the trend indicated improving water quality at levels of confidence defined inTable 2.. Green colours indicate sites with improving trends, and red-orange colours indicate sites with degrading trends. Trends used in this graph are not flow-adjusted, and have a hi-censor filter applied.

Page 144 | 155

2000-2020

#### 2000-2020

LWP (2020b) provides maps showing trend direction for rivers across Otago, the 10-year raw water quality trend direction is shown in Figure 54 and the 20-year raw water quality trend direction in Figure 55



Figure 54 Map of river sites classified by confidence that their 10-year raw water quality trend direction indicated improving water quality. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**. LWP (2020b)

Page 145 | 155

2000-2020



Figure 55 Map of river sites classified by confidence that their 20-year raw water quality trend direction indicated improving water quality. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** LWP. (2020b)

Page 146 | 155

### 2000-2020

LWP (2020b) also provide maps for lake trend direction, these are shown in Figure 56 and Figure 57Figure 56 Map of lake sites classified by confidence that their 10-year water quality trend direction indicated improving water quality. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.**. LWP (2020b)

Page 147 | 155

2000-2020



Figure 56 Map of lake sites classified by confidence that their 10-year water quality trend direction indicated improving water quality. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** LWP (2020b)

Page 148 | 155

2000-2020



Figure 57 Map of lake sites classified by confidence that their 18-year water quality trend direction indicated improving water quality. Confidence is expressed categorically based on the levels defined in **Error! Reference source not found.** LWP (2020b)

Page 149 | 155

2000-2020

Page 150 | 155

2000-2020

# Acknowledgements

Thanks to LWP Ltd for calculation of state and trend metrics, provision of the summary maps and matrices and for assistance and support. Thanks also to ORC's Environmental Monitoring team for their collection of surface water samples, there would be no report without them. Finally, thanks to Jason Augspurger and Hugo Borges for overall helpfulness.

Page 151 | 155

2000-2020

# References

ANZECC, A., 2000. Australian and New Zealand Guidelines for Fresh and Marine Water Quality. Australian and New Zealand Environment and Conservation Council and Agriculture and Resource Management Council of Australia and New Zealand, Canberra:1–103.

Burdon, F.J.; McIntosh, A.R.; Harding, J.S. Habitat loss drives threshold response of benthic invertebrate communities to deposited sediment in agricultural streams. Ecol. Appl. 2013, 23, 1036–1047

Clapcott J, Young R, Sinner J, Wilcox M, Storey R, Quinn J, Daughney C, Canning A, 2018. Freshwater biophysical ecosystem health framework. Prepared for Ministry for the Environment. Cawthron Report No. 3194. 89 p. plus appendices.

Choquette, A.F., R.M. Hirsch, J.C. Murphy, L.T. Johnson, and R.B. Confesor Jr, 2019. Tracking Changes in Nutrient Delivery to Western Lake Erie: Approaches to Compensate for Variability and Trends in Streamflow. Journal of Great Lakes Research 45:21–39.

Fraser. C; Snelder, T. 2020a. State of Lake and River Wate Quality in the Otago Region. For records up to 30 June 2020

Fraser. C; Snelder, T. 2020b. ORC River, Groundwater and Lake water quality Trend analysis. For 10 and 20-year periods up to September 2020.

Franklin, P., D. Booker, and R. Stoffels, 2020. Contract 23184: Task 2 - Turbidity and Visual Clarity Threshold Conversion. NIWA.

Greenland, S., S.J. Senn, K.J. Rothman, J.B. Carlin, C. Poole, S.N. Goodman, and D.G. Altman, 2016. Statistical Tests, P Values, Confidence Intervals, and Power: A Guide to Misinterpretations. European Journal of Epidemiology 31:337–350.

Helsel, D.R., 2012. Reporting Limits. Statistics for Censored Environmental Data Using Minitab and R. John Wiley & Sons, pp. 22–36.

Helsel, D.R., R.M. Hirsch, K.R. Ryberg, S.A. Archfield, and E.J. Gilroy, 2020. Statistical Methods in Water Resources. Report, Reston, VA.

Hirsch, R.M., J.R. Slack, and R.A. Smith, 1982. Techniques of Trend Analysis for Monthly Water Quality Data. Water Resources Research 18:107–121.

Hickey, C., 2014. Derivation of Indicative Ammoniacal Nitrogen Guidelines for the National Objectives Framework. Memo prepared for Ms Vera Power, Ministry for the Environment, by NIWA.

Kienzle (S.W) and Schmidt (J). Hydrological impacts of irrigated agriculture in the Manuherekia catchment, Otago, New Zealand. Journal of Hydrology (NZ) 47 (2): 67-84 2008

Larned, S., A. Whitehead, C.E. Fraser, T. Snelder, and J. Yang, 2018. Water Quality State and Trends in New Zealand Rivers. Analyses of National-Scale Data Ending in 2017. prepared for Ministry for the Environment, NIWA.

Page 152 | 155

#### 2000-2020

Larned, S., T. Snelder, M. Unwin, G. McBride, P. Verburg, and H. McMillan, 2015. Analysis of Water Quality in New Zealand Lakes and Rivers. Prepared for the Ministry for the Environment. Wellington: Ministry for the Environment.

Larned, S., A. Whitehead, C.E. Fraser, T. Snelder, and J. Yang, 2018. Water Quality State and Trends in New Zealand Rivers. Analyses of National-Scale Data Ending in 2017. prepared for Ministry for the Environment, NIWA.

LWP Ltd., 2020a. State of Lake and River Water Quality in the Otago Region. For records up to 30 June 2020, prepared for Otago Regional Council.

LWP Ltd., 2020b. ORC River, Groundwater and Lake water quality Trend analysis. For 10 and 20-year periods up to September 2020, prepared for Otago Regional Council.

McBride, G.B., 2005. Using Statistical Methods for Water Quality Management: Issues, Problems and Solutions. John Wiley & Sons.

McBride, G.B., 2019. Has Water Quality Improved or Been Maintained? A Quantitative Assessment Procedure. Journal of Environmental Quality.

McBride, G., R.G. Cole, I. Westbrooke, and I. Jowett, 2014. Assessing Environmentally Significant Effects: A Better Strength-of-Evidence than a Single P Value? Environmental Monitoring and Assessment 186:2729–2740.

Ministry for the Environment. 2018. A Guide to Attributes in Appendix 2 of the National Policy Statement for Freshwater Management (as amended 2017). Wellington: Ministry for the Environment

Ministry for the Environment. 2020. Action for healthy waterways: Guidance on look-up tables for setting nutrient targets for periphyton. Wellington: Ministry for the Environment

Ministry for Environment and Ministry of Health, 2003. Microbiological Water Quality Guidelines for Marine and Freshwater Recreational Areas. Ministry for the Environment. https://www.mfe.govt.nz/sites/default/files/microbiological-quality-jun03.pdf.

NZ Government, 2020. National Policy Statement for Freshwater Management 2020.

Otago Regional Council, 2004. Regional Plan: Water for Otago

Otago Regional Council. (2004). Regional Plan: Water for Otago. Dunedin: Otago Regional Council.

Otago Regional Council, 2017. State of the environment: Surface water quality in Otago. 2006 to 2011. Otago Regional Council.

Smith, D.G., G.B. McBride, G.G. Bryers, J. Wisse, and D.F. Mink, 1996. Trends in New Zealand's National River Water Quality Network. New Zealand Journal of Marine and Freshwater Research 30:485–500.

Snelder, T.H. and B.J.F. Biggs, 2002. Multi-Scale River Environment Classification for Water Resources Management. Journal of the American Water Resources Association 38:1225–1240.

Snelder T., Biggs, B. and Weatherhead, M. (2004), New Zealand River Environment Classification User Guide. Prepared by NIWA for the Ministry for the Environment

Page 153 | 155

#### 2000-2020

Stark JD, Maxted JR, 2007. A user guide for the Macroinvertebrate Community Index. Prepared for the Ministry for the Environment. Cawthron Report No.1166. 58 p.

Stocker et al., 2014, IPCC Climate Change 2013: The Physical Science Basis - Findings and Lessons Learned

Whitehead, A.L., Booker, D.J. (2020). NZ River Maps: An interactive online tool for mapping predicted freshwater variables across New Zealand. NIWA, Christchurch. <u>https://shiny.niwa.co.nz/nzrivermaps/</u> REC classes are CWM for M d/s Fork, CDM for Dunstan Creek and CDH for all other sites.

Wasserstein, R.L. and N.A. Lazar, 2016. The ASA Statement on P-Values: Context, Process, and Purpose. Taylor & Francis.

Page 154 | 155

2000-2020

Page 155 | 155

The general subject of each matter to be considered while the public is excluded, the reason for passing this resolution in relation to each matter, and the specific grounds under section 48(1) of the Local Government Official Information and Meetings Act 1987 for the passing of this resolution are as follows:

| General subject<br>of each matter to   | Reason for passing this resolution in relation to each matter  | Ground(s) under section<br>48(1) for the passing of this  |  |  |
|--|--|---|--|--|
| be considered  |  | resolution  |  |  |
| 1.1 Minutes of<br>the public-<br>excluded Council<br>Meeting of 24<br>February 2021. | To protect the privacy of natural<br>persons, including that of deceased<br>natural persons – Section 7(2)(a);<br>To enable any local authority holding<br>the information to carry on, without<br>prejudice or disadvantage, negotiations<br>(including commercial and industrial<br>negotiations) – Section 7(2)(i);<br>To prevent the disclosure or use of<br>official information for improper gain or<br>improper advantage – Section 7(2)(j)   | Section 48(1)(a);<br>Subject to subsection (3), a<br>local authority may by<br>resolution exclude the public<br>from the whole or any part<br>of the proceedings of any<br>meeting only on 1 or more of<br>the following grounds:<br>(a) that the public conduct<br>of the whole or the relevant<br>part of the proceedings of<br>the meeting would be likely<br>to result in the disclosure of<br>information for which good<br>reason for withholding<br>would exist. |  |  |
| 3.1 LiDAR Data<br>Capture Contract<br>Approval                                       | To protect information which is subject<br>to an obligation of confidence where<br>the making available of the information<br>would be likely to:<br>i. prejudice the supply of similar<br>information, or information from the<br>same source, where it is in the public<br>interest that such information should<br>continue to be supplied – Section<br>7(2)(c)(i);<br>To carry on, without prejudice or<br>disadvantage, negotiations (including<br>commercial and industrial negotiations)<br>– Section 7(2)(i) | Section 48(1)(a);<br>Subject to subsection (3), a<br>local authority may by<br>resolution exclude the public<br>from the whole or any part<br>of the proceedings of any<br>meeting only on 1 or more of<br>the following grounds:<br>(a) that the public conduct<br>of the whole or the relevant<br>part of the proceedings of<br>the meeting would be likely<br>to result in the disclosure of<br>information for which good<br>reason for withholding<br>would exist. |  |  |
| 3.2 Port Otago<br>Director<br>Appointment  | To protect the privacy of natural<br>persons, including that of deceased<br>natural persons – Section 7(2)(a);<br>To enable the council to carry out,<br>without prejudice or disadvantage,<br>commercial activities – Section 7(2)(h)   | Section 48(1)(a);<br>Subject to subsection (3), a<br>local authority may by<br>resolution exclude the public<br>from the whole or any part<br>of the proceedings of any<br>meeting only on 1 or more of<br>the following grounds:<br>(a) that the public conduct<br>of the whole or the relevant  |  |  |

|  |   | part of the proceedings of<br>the meeting would be likely<br>to result in the disclosure of<br>information for which good<br>reason for withholding<br>would exist.   |
|--|---|---|
| 3.3 Head Office<br>Accommodation<br>Update | <ul> <li>To protect information which if public would;</li> <li>ii. unreasonably prejudice the commercial position of the person who supplied or who is the subject of the information – Section 7(2)(b)(ii);</li> <li>To protect information which is subject to an obligation of confidence where the making available of the information would be likely to: <ul> <li>i. prejudice the supply of similar information, or information from the same source, where it is in the public interest that such information should continue to be supplied – Section 7(2)(c)(i);</li> <li>To enable any local authority holding the information to carry out, without prejudice or disadvantage, commercial activities – Section 7(2)(h)</li> <li>To enable any local authority holding the information to carry on, without prejudice or disadvantage, negotiations (including commercial and industrial negotiations) – Section 7(2)(i)</li> </ul> </li> </ul> | Section 48(1)(a);<br>Subject to subsection (3), a<br>local authority may by<br>resolution exclude the public<br>from the whole or any part<br>of the proceedings of any<br>meeting only on 1 or more of<br>the following grounds:<br>(a) that the public conduct<br>of the whole or the relevant<br>part of the proceedings of<br>the meeting would be likely<br>to result in the disclosure of<br>information for which good<br>reason for withholding<br>would exist. |

This resolution is made in reliance on section 48(1)(a) of the Local Government Official Information and Meetings Act 1987 and the particular interest or interests protected by section 6 or section 7 of that Act or section 6 or section 7 or section 9 of the Official Information Act 1982, as the case may require, which would be prejudiced by the holding of the whole or the relevant part of the proceedings of the meeting in public.